UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM 10-K

T ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the fiscal year ended December 31, 2015

OR

£ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from to

Commission File Number 1-8957

ALASKA AIR GROUP, INC.

Delaware

(State of Incorporation)

91-1292054

(I.R.S. Employer Identification No.)

19300 International Boulevard, Seattle, Washington 98188

Telephone: (206) 392-5040

Securities registered pursuant to Section 12(b) of the Act:

Common Stock, \$0.01 Par Value

New York Stock Exchange

Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes T No £

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes £ No T

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes £ No T

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes T No £

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. T

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See definitions of "large accelerated filer", "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act: Large accelerated filer T Accelerated filer £ Non-accelerated filer £ Smaller reporting company £

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act.): Yes £ No T

As of January 31, 2016, shares of common stock outstanding totaled 124,729,056. The aggregate market value of the shares of common stock of Alaska Air Group, Inc. held by nonaffiliates on June 30, 2015, was approximately \$8.2 billion (based on the closing price of \$64.43 per share on the New York Stock Exchange on that date).

DOCUMENTS INCORPORATED BY REFERENCE

Portions of Definitive Proxy Statement relating to 2016 Annual Meeting of Shareholders are incorporated by reference in Part III.

ALASKA AIR GROUP, INC. ANNUAL REPORT ON FORM 10-K FOR THE YEAR ENDED DECEMBER 31, 2015

TABLE OF CONTENTS

<u>PART I</u>		<u>4</u>
<u>ITEM 1.</u>	OUR BUSINESS	<u>4</u>
<u>ITEM 1A.</u>	<u>RISK FACTORS</u>	<u>12</u>
<u>ITEM 1B.</u>	UNRESOLVED STAFF COMMENTS	<u>17</u>
<u>ITEM 2.</u>	PROPERTIES	<u>17</u>
<u>ITEM 3.</u>	LEGAL PROCEEDINGS	<u>17</u>
<u>ITEM 4.</u>	MINE SAFETY DISCLOSURES	<u>17</u>
<u>PART II</u>		<u>17</u>
<u>ITEM 5.</u>	<u>MARKET FOR THE REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS, AND ISSUER</u> <u>PURCHASES OF EQUITY SECURITIES</u>	<u>17</u>
<u>ITEM 6.</u>	SELECTED CONSOLIDATED FINANCIAL AND OPERATING DATA	<u>18</u>
<u>ITEM 7.</u>	MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	<u>18</u>
<u>ITEM 7A.</u>	QUANTITATIVE AND QUALITATIVE DISCLOSURE ABOUT MARKET RISK	<u>38</u>
<u>ITEM 8.</u>	CONSOLIDATED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA	<u>39</u>
<u>ITEM 9.</u>	CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE	<u>69</u>
<u>ITEM 9A.</u>	CONTROLS AND PROCEDURES	<u>69</u>
<u>ITEM 9B.</u>	OTHER INFORMATION	<u>72</u>
<u>PART III</u>		<u>72</u>
<u>ITEM 10.</u>	DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE	<u>72</u>
<u>ITEM 11.</u>	EXECUTIVE COMPENSATION	<u>72</u>
<u>ITEM 12.</u>	<u>SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT, AND RELATED</u> <u>STOCKHOLDER MATTERS</u>	<u>72</u>
<u>ITEM 13.</u>	CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE	<u>72</u>
<u>ITEM 14.</u>	PRINCIPAL ACCOUNTANT FEES AND SERVICES	<u>72</u>
PART IV		<u>72</u>
<u>ITEM 15.</u>	<u>EXHIBITS</u>	<u>72</u>
	<u>SIGNATURES</u>	<u>73</u>

As used in this Form 10-K, the terms "Air Group," the "Company," "our," "we" and "us," refer to Alaska Air Group, Inc. and its subsidiaries, unless the context indicates otherwise. Alaska Airlines, Inc. and Horizon Air Industries, Inc. are referred to as "Alaska" and "Horizon," respectively, and together as our "airlines."

2

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

In addition to historical information, this Form 10-K contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995. Forward-looking statements are those that predict or describe future events or trends and that do not relate solely to historical matters. You can generally identify forward-looking statements as statements containing the words "believe," "expect," "will," "anticipate," "intend," "estimate," "project," "assume" or other similar expressions, although not all forward-looking statements contain these identifying words. Forward-looking statements involve risks and uncertainties that could cause actual results to differ materially from historical experience or the Company's present expectations.

You should not place undue reliance on our forward-looking statements because the matters they describe are subject to known and unknown risks, uncertainties and other unpredictable factors, many of which are beyond our control.

Our forward-looking statements are based on the information currently available to us and speak only as of the date on which this report was filed with the SEC. We expressly disclaim any obligation to issue any updates or revisions to our forward-looking statements, even if subsequent events cause our expectations to change regarding the matters discussed in those statements. Over time, our actual results, performance or achievements will likely differ from the anticipated results, performance or achievements that are expressed or implied by our forward-looking statements, and such differences might be significant and materially adverse to our shareholders. For a discussion of these and other risk factors in this Form 10-K, see "Item 1A: Risk Factors." Please consider our forward-looking statements in light of those risks as you read this report.

PART I

ITEM 1. OUR BUSINESS

Alaska Air Group ("Air Group") operates Alaska Airlines ("Alaska") and Horizon Air ("Horizon"), which together with its partner regional airlines serve more than 100 cities through an expansive network in Alaska, the Lower 48, Hawaii, Canada, Mexico, and Costa Rica. During 2015, we carried an all-time high 32 million passengers and earned record full-year adjusted earnings of \$842 million.

Our objective is to be one of the most respected U.S. airlines by our customers, employees, and shareholders. We believe our success depends on our ability to provide safe air transportation, develop relationships with customers by providing exceptional customer service and low fares, and maintain a competitive cost structure to compete effectively. It is important to us that we achieve our objective as a socially responsible company that values not just our performance, but also our people, our community, and our environment.

While aircraft and technology enable us to provide air transportation, we recognize this is fundamentally a people business. Our employees maintain and strengthen our relationships with customers, and our success depends on our employees working together to successfully execute on our strategy. In 2015, Alaska was named one of America's Top 100 Employers by Forbes Magazine. We launched our "Beyond Service" two-day customer service workshop for all of our customer-facing employees to provide them with the framework and tools they need to improve upon our already award-winning customer service. In that vein, in 2015, Alaska Airlines ranked "Highest in Customer Satisfaction among Traditional Network Carriers" by J.D. Power for the eighth year in a row. Alaska Airlines also held the No. 1 spot in the Wall Street Journal's "Middle Seat" scorecard for U.S. airlines for three years in a row. We have been the leader in the industry for on-time performance among major airlines for the past six years. For achieving safety, customer service, operational and financial goals, we rewarded our employees with a record \$120 million in incentive pay during 2015.

In support of the communities that we serve, we strive to be an industry leader in environmental and community stewardship. In 2015, Air Group improved fuel efficiency by 2.2% from the prior year, as measured by available seat miles flown per gallon. Air Group donated \$12 million to approximately 1,050 charitable organizations and our employees volunteered more than 27,000 hours of community service. Included in this amount are annual contributions of our ongoing multi-year grants to several organizations - we pledged \$1.5 million to support job training for workers at Seattle-Tacoma airport through Port Jobs, \$2.5 million for Seattle's bike-share program, \$1 million for the Alaska Native Science and Engineering Program, and \$2.5 million to Seattle's Museum of Flight to guide students toward a future in science, technology, engineering, and math. In 2015, we granted additional support to Washington Information Science Education Consortium, Washington State University, Kupu of Hawaii, University of Hawaii, and pledged \$40 million to the the University of Washington, in part, to increase our impact on education supporting specific programs and scholarships.

We earned record financial results in 2015, marking our 12th consecutive annual profit on an adjusted basis. We achieved an after-tax return on invested capital of 25.2%, approximately three times our weighted average cost of capital. Strong earnings improved our cash flow and strengthened our balance sheet resulting in a debt-to-capital ratio of 27%, which compares favorably with other high-quality industrial transport companies and the companies in the S&P 500. Due to our strong financial health, we are one of only two U.S. airlines with investment grade credit ratings. With the cash generated by the continued success we have had in the past decade, we are able to continue to invest in our business for profitable growth and to enhance the customer experience. All of our 737-800/900/900ER aircraft feature innovative Recaro seats with power at every seat, our Wi-Fi enabled in-flight entertainment system, and our branded in-flight experience, Alaska BeyondTM.

As we look to the future, we will build on the success of the past few years by executing our strategic plan — the Five Focus Areas:

Safety and Compliance

We have an unwavering commitment to run a safe and compliant operation, and we will not compromise this commitment in the pursuit of other initiatives. Alaska and Horizon are awaiting final FAA certification of our fully-implemented Safety Management System (SMS). SMS helps identify and manage risk and builds a sustainable culture of safety for every Air Group employee. Once again, for 2015, 100% of our Alaska and Horizon aircraft technicians completed the requirements for the FAA's "Diamond Certificate of Excellence" award. This is the 14th consecutive year Alaska Airlines has received the award and the 14th time in the last 16 years for Horizon. In 2014, we launched Ready, Safe, Go - a safety campaign designed to increase safety awareness across the Air Group System. With our Ready, Safe, Go program in its second year, employee

4

awareness of their individual contributions to a strong culture of safety continues to increase. In 2015, Alaska and Horizon employees made a "Personal Commitment to Safety" during facilitated discussions with company leaders.

People Focus

Our success depends on our employees. Higher employee engagement drives higher productivity, superior execution, and better customer service, which is why we listen to our employees for feedback in shaping our strategy. Employee engagement scores from our annual employee survey are at historical highs. In addition to the "Beyond Service" training for our customer-facing employees, all of Air Group's leaders participated in a multi-day leadership training called "Gear Up 2." This is the second phase of our award-winning leadership training where our leaders spent time focusing on several of our core leadership principles in an effort to give them the tools they need to effectively lead a highly-engaged workforce.

We understand that aligning our employees' goals with the Company's goals is important in achieving success. All employees participate in our Performance-Based Pay (PBP) and Operational Performance Rewards (OPR) programs, which encourage employees to work together to achieve metrics related to safety, profitability, on-time performance, low costs and customer satisfaction. Over the last five years, our incentive programs have paid out on average, over 8.6% of annual pay, or more than one month's pay, for most employees. This is consistent with one of our guiding principles that we want to pay our people well with a goal of reaching the industry's best productivity over time. To that end, we are currently in long-term agreements with all of our major work groups, which provide the Company, employees, and investors with long-term stability.

Hassle-Free Customer Experience

We want to be the easiest airline to fly. In each step of the customer's journey, from booking a ticket to check-in, from flying in our aircraft to claiming baggage at the final destination, we want to provide a hassle-free experience for our customers. Our industry-leading on-time performance for the past six years reflects the reliable service we provide our customers, and we were the first airline to guarantee checked baggage delivery to the carousel within 20 minutes. Customers can tag their own bags at airport kiosks, or from their homes, and we have fingerprint scan entry to our airport lounges. We lead the travel industry in mobile innovation with a 5-star rated iPhone app, and Android and Microsoft apps that allow passengers to purchase tickets, check-in, upgrade seats, and reserve food for the flight - all with helpful notifications that inform customers when there are changes to their flights. The Transportation Security Administration (TSA) Pre-Check Program is available in 60 of our locations, which allows eligible customers to opt-in for reduced screening requirements. To hear directly from our customers, we have the Alaska Listens survey with five simple questions designed to get timely feedback - and we guarantee a response within 72 hours if there is an issue that needs to be resolved. As passengers take more control of their travel experience, we are able to reduce the time it takes a customer to move from the airport curb to the aircraft.

Energetic and Compelling Brand

We want to be recognized as the preferred airline to fly. In January 2016, we introduced a bold new brand expression, including an updated identity, livery, and look and feel for our digital and physical experiences. We believe there's an opportunity to deepen the emotional connection with our customers as we continue to expand and grow. Our updated brand expression draws upon our rich heritage while infusing it with additional warmth and energy - to better reflect how customers feel about our brand and the great service that we pride ourselves in delivering.

We continue to invest in a better customer experience. Onboard, customers will continue to enjoy more of what they love with free and premium entertainment direct to their devices, Pacific Northwest-inspired food and beverages and custom leather seats with power outlets for laptops and personal devices and larger overhead bins for carry-on bags. In the fall of 2016, we will introduce a new Premium Class section in the main cabin with increased pitch and other amenities.

We also continue to invest in advertising across markets. We recently renewed our partnership with Seattle Seahawks quarterback Russell Wilson, as our Chief Football Officer. We also announced a ten-year sponsorship with the University of Washington, which includes naming rights to the Alaska Airlines Field at Husky Stadium and the Alaska Airlines Arena at Hec Edmunson Pavillion, among other benefits.

Low Fares, Low Costs and Network Growth

We believe that in order to provide low fares for customers in a growing network of destinations, while returning value to our shareholders, it is imperative for us to maintain a competitive cost structure. In 2015, we lowered our unit costs, excluding fuel, by 0.7% on a consolidated basis, representing the sixth consecutive year of annual reduction. We achieved this through a continued focus on productivity, cost management, and network growth. We increased employee productivity in 2015 and will continue to focus on that metric as we leverage growth. We also manage fuel costs by flying larger, more fuel-efficient aircraft, which have increased our fuel efficiency as measured by available seat miles flown per gallon by 5.6% over the last five years. Additionally, we have added split-scimitar winglets to 94 aircraft, which are expected to increase fuel efficiency by



approximately 1.5% per aircraft. Looking forward, we have committed to purchasing 31 737-900ER and 37 737-MAX aircraft with deliveries from 2016 to 2022, and two Q400 aircraft with deliveries in 2018, although these are subject to change. In addition, we will increase regional capacity by adding 18 E175s with contractual deliveries from 2016 to 2017 and we may order an additional 30 regional jets with delivery beginning in 2017 if we reach acceptable commercial terms. These aircraft deliveries position us for growth and help to ensure that we will continue to operate the most fuel-efficient aircraft available for the foreseeable future.

In 2015, we added 20 new markets to our route structure. We continued to strengthen our Seattle network to offer further utility to our customers by offering non-stop flights to markets like New York (JFK), Charleston, Raleigh-Durham, Nashville, Oklahoma City, and Milwaukee. We also added flying from Los Angeles (LAX) to Liberia and San Jose, Costa Rica and Baltimore. We have grown capacity over 7% annually on average for the past 20 years and we plan to continue to grow between 4% and 8% annually in the next several years. For 2016, we plan to grow capacity approximately 8%.

AIR GROUP

Alaska Air Group is a Delaware corporation incorporated in 1985 and the holding company of Alaska Airlines and Horizon Air. Although Alaska and Horizon both operate as airlines, their business plans, competition, and economic risks differ substantially. Alaska Airlines is an Alaska corporation that was organized in 1932 and incorporated in 1937. Horizon Air Industries is a Washington corporation that first began service and was incorporated in 1981. Horizon was acquired by Air Group in 1986. Alaska operates a fleet of passenger jets (mainline) and contracts with Horizon, SkyWest Airlines, Inc. (SkyWest) and Peninsula Airways, Inc. (PenAir) for regional capacity such that Alaska receives all passenger revenue from those flights. Horizon operates a fleet of turboprop aircraft and sells all of its capacity to Alaska pursuant to a capacity purchase arrangement. The majority of our revenues are generated by transporting passengers. The percentage of revenues by category is as follows:

	2015	2014	2013	2012	2011
Mainline passenger revenue	70%	70%	70%	71%	69%
Regional passenger revenue	15%	15%	16%	16%	17%
Other revenue	13%	13%	12%	11%	12%
Freight and Mail revenue	2%	2%	2%	2%	2%
Total	100%	100%	100%	100%	100%

We attempt to deploy aircraft into the network in ways that best optimize our revenues and profitability, and reduce our seasonality.

The percentage of our capacity by region is as follows:

	2015	2014	2013	2012	2011
West Coast	36%	36%	34%	35%	37%
Transcon/midcon	24%	22%	22%	19%	19%
Hawaii	18%	18%	19%	20%	16%
Alaska	15%	15%	16%	17%	18%
Mexico	6%	6%	7%	7%	9%
Canada	1%	3%	2%	2%	1%
Total	100%	100%	100%	100%	100%

MAINLINE

We offer extensive north/south service within the western U.S., Canada, Mexico, and Costa Rica, as well as passenger and dedicated cargo services to and within the state of Alaska. We also provide long-haul east/west service to Hawaii and cities in the mid-continental and eastern U.S., primarily from Seattle, where we have our largest concentration of departures; although we do offer long-haul departures from other cities as well.

In 2015, we carried 23 million revenue passengers in our mainline operations. At December 31, 2015, Alaska's operating fleet consisted of 147 Boeing 737 jet aircraft, compared to 137 B737 aircraft as of December 31, 2014.

The percentage of mainline passenger capacity by region and average stage length is presented below:

	2015	2014	2013	2012	2011
West Coast	31%	31%	28%	29%	31%
Transcon/midcon	27%	25%	25%	22%	21%
Hawaii	20%	20%	21%	22%	18%
Alaska	16%	16%	18%	18%	20%
Mexico	6%	7%	7%	8%	8%
Canada	%	1%	1%	1%	2%
Total	100%	100%	100%	100%	100%
Average Stage Length	1,195	1,182	1,177	1,161	1,114

REGIONAL

Our regional operations consist of flights operated by Horizon, SkyWest and PenAir. In 2015, our regional operations carried approximately 9 million revenue passengers, primarily in the states of Washington, Oregon, Idaho and California. Horizon is the largest regional airline in the Pacific Northwest and

carries about 85% of Air Group's regional revenue passengers.

Based on 2015 passenger enplanements on regional aircraft, our leading airports are Seattle and Portland. At December 31, 2015, Horizon's operating fleet consisted of 52 Bombardier Q400 turboprop aircraft. Horizon flights are listed under Alaska's designator code in airline reservation systems, and in customerfacing locations. The regional fleet operated by SkyWest consisted of eight CRJ 700 aircraft and five E175 aircraft.

The percentage of regional passenger capacity by region and average stage length is presented below:

	2015	2014	2013	2012	2011
West Coast	62%	66%	66%	68%	68%
Pacific Northwest	19%	19%	21%	20%	19%
Canada	7%	8%	9%	9%	9%
Alaska	5%	4%	2%	2%	2%
Midcon	6%	2%	1%	%	%
Mexico	1%	1%	1%	1%	2%
Total	100%	100%	100%	100%	100%
Average Stage Length	348	339	329	332	329

MILEAGE PLAN

The Alaska Airlines Mileage Plan[™] program provides a comprehensive suite of frequent flier benefits. Miles can be earned by flying on Alaska or on one of our 16 airline partners, or by using the Alaska Airlines Visa Signature card, or through other non-airline partners. Our extensive list of airline partners includes carriers associated with two of the three major global alliances (Oneworld and SkyTeam), making it easier for our members to earn miles and reach preferred status in our Mileage PlanTM, and have access to a large network of travel destinations. Further, members can receive 25,000 bonus miles (30,000 beginning in the Spring of 2016) upon signing up for the Alaska Airlines Visa Signature card and earn triple miles on purchases made on Alaska Airlines flights or on alaskaair.com. Alaska Airlines Visa Signature cardholders also receive an annual companion ticket that allows members to purchase an additional ticket for \$99 plus taxes, with no restrictions or black-out dates, and a free first checked bag for all parties in the same itinerary. Earned miles can be redeemed for flights on Alaska Airlines or on any of our partner airlines, or for upgrades to First Class on Alaska Airlines for as low as 15,000 miles. All of these benefits give our Mileage PlanTM members more value for their travel on Alaska Airlines, which led to our Mileage PlanTM receiving the highest ranking by frequent fliers in the first-ever J.D. Power Airline Loyalty/Rewards Program Satisfaction Report in 2014 and again in 2015.

Mileage PlanTM revenues represent approximately 11% of Air Group's total revenues. Furthermore, our Mileage PlanTM helps drive more revenue by attaining new customers and building customer loyalty through the benefits that we provide. The Mileage Plan™ provides more value per dollar spent on the Alaska Airlines Visa Signature card, in comparison to other frequent flier programs in the industry. Summary of the benefits provided in comparison to some of our competitors are as follows:

	Alaska Airlines Signature Visa	Platinum Select AAdvantage	Gold Delta SkyMiles	United Mileage Plus Explorer	Southwest Rapid Rewards Premier
Bonus miles awarded	30,000 after spending \$1,000 in 3 months*	30,000 after spending \$1,000 in 3 months	30,000 after spending \$1,000 in 3 months	30,000 after spending \$1,000 in 3 months	50,000 after spending \$2,000 in 3 months
Annual fee	\$75	\$95	\$95	\$95	\$99
Miles for "on" spend	3x	2x	2x	2x	2x
Companion fare	Yes - annual companion fare purchased for \$99 plus tax.	No	No	No	No
First bag free	Yes	Yes	Yes	Yes	No bag fees
*Expected launch of 30,000 bonus mi	les in May 2016, Currently, bon	us miles are 25.000.			

ected launch of 30,000 bonus miles in May 2016. Currently, bonus miles are 25,000.



AGREEMENTS WITH OTHER AIRLINES

Our agreements fall into three different categories: Frequent Flier, Codeshare, and Interline agreements. Frequent Flier agreements offer mileage credits and redemptions for our Mileage Plan[™] members. Alaska offers one of the most comprehensive frequent flier programs for our Mileage Plan[™] members through our frequent flier partnerships with 16 domestic and international carriers.

Codeshare agreements allow one or more marketing carriers to sell seats on a single operating carrier that services passengers under multiple flight numbers. The sale of codeshare seats can vary depending on the sale arrangement. For example, in a free-sale arrangement, the marketing carrier sells the operating carrier's inventory without any restriction; whereas in a block space arrangement, a fixed amount of seats are sold to the marketing carrier by the operating carrier. The interchangeability of the flight code between carriers provides a greater selection of flights for customers, along with increased flexibility for mileage accrual and redemption.

Interline agreements allow airlines to jointly offer a competitive, single-fare itinerary to customers traveling via multiple carriers to a final destination. An interline itinerary offered by one airline may not necessarily be offered by the other, and the fares collected from passengers are prorated and distributed to interline partners according to preexisting agreements between the carriers. Frequent Flier, Codeshare, and Interline agreements help increase our traffic and revenue by providing more route choices to customers.

We have marketing alliances with a number of airlines that provide frequent flier and codesharing opportunities. Alliances are an important part of our strategy and enhance our revenues by:

- offering our customers more travel destinations and better mileage credit/redemption opportunities, including elite qualifying miles on all of our major U.S. and international airline partners;
- giving our Mileage Plan[™] program a competitive advantage because of our partnership with carriers from two of the three major global alliances (Oneworld and SkyTeam);
- giving us access to more connecting traffic from other airlines; and
- providing members of our alliance partners' frequent flier programs an opportunity to travel on Alaska and its regional affiliates while earning
 mileage credit in our partners' programs.

Most of our codeshare relationships are free-sale codeshares, where the marketing carrier sells seats on the operating carrier's flights from the operating carrier's inventory, but takes no inventory risk. Our marketing agreements have various termination dates, and at any time, one or more may be in the process of renegotiation.

The comprehensive summary of alliances with other airlines is as follows:

	Frequent Flier Agreement	Codeshare — Alaska Flight # on Flights Operated by Other Airline	Codeshare — Other Airline Flight # on Flights Operated by Alaska / Horizon / SkyWest
Major U.S. or International Airlines			
Aeromexico	Yes	No	Yes
American Airlines ^(a)	Yes	Yes	Yes
Air France	Yes	No	Yes
British Airways	Yes	No	No
Cathay Pacific Airways	Yes	No	Yes
Delta Air Lines ^(a)	Yes	Yes	Yes
Emirates	Yes	No	Yes
Icelandair	Yes	No	Yes
Hainan Airlines	Yes	No	No
KLM	Yes	No	Yes
Korean Air	Yes	No	Yes
LAN S.A.	Yes	No	Yes
Fiji Airways ^(b)	Yes	No	Yes
Qantas	Yes	No	Yes
Regional Airlines			
Rav'n Alaska	Yes	Yes	No
PenAir ^(b)	Yes	Yes	No
(a) Alaska has codeshare agreements with American a	nd Delta regional affiliate carriers as v	vell.	

b) These airlines do not have their own frequent flier program. However, Alaska's Mileage Plan™ members can earn and redeem miles on these airlines' route systems.

The following is the financial impact of our marketing alliances:

	2015	2014	2013	2012	2011
Air Group Marketed Revenues	90%	91%	90%	90%	89%
Codeshare Agreements:					
American Airlines	4%	3%	2%	3%	3%
Delta Air Lines	2 %	2%	4%	3%	4%
Others	1%	1%	1%	1%	1%
Interline Agreements:					
Domestic Interline	2%	2%	2%	2%	2%
International Interline	1%	1%	1%	1%	1%
Total Operating Revenue	100%	100%	100%	100%	100%

OTHER REVENUE

Other revenue consists of freight and mail revenue, and ancillary revenue such as bag fees, change fees, on-board food and beverage, and Boardroom membership. Total other revenue, excluding Mileage Plan[™] revenue, represents about 7% of our total revenues. In recent years, we have seen growth in our ancillary revenue as we expand services on-board such as Tom Douglas signature meals, in-flight entertainment, and Wi-Fi. Although we do charge bag fees, we offer a 20-minute bag guarantee so that we deliver value to our customers through fast, reliable service. In 2015, we added a free bag as a permanent feature of our affinity credit card. As we focus on ways to better serve our customers, we expect our ancillary revenues will continue to grow.

GENERAL

The airline industry is highly competitive, subject to various uncertainties, and has historically been characterized by low profit margins. Uncertainties include general economic conditions, volatile fuel prices, industry instability, new competition, a largely unionized work force, the need to finance large capital expenditures and the related availability of capital, government regulation, including taxes and fees, and potential aircraft incidents. Airlines have high fixed costs, primarily for wages, aircraft fuel, aircraft ownership, and facilities rents. Because expenses of a flight do not vary significantly based on the number of passengers carried, a relatively small change in the number of passengers or in pricing has a disproportionate effect on an airline's operating and financial results. In other words, a minor shortfall in expected revenue levels could cause a disproportionately negative impact on our operating and financial results. Passenger demand and ticket prices are, to a large measure, influenced by the general state of the economy, current global economic and political events, and total available airline seat capacity.

In 2015, the airline industry reported record revenues and profits as the global economy continued to recover and oil prices declined significantly. As the industry strengthens, airlines are now making significant investments in airports, in new planes, and in new services to differentiate their customer service offering. Thus, the level of competition is expected to increase.

FUEL

Our business and financial results are highly affected by the price and, potentially, the availability of aircraft fuel. The cost of aircraft fuel is volatile and outside of our control, and it can have a significant and immediate impact on our operating results. Over the past five years, aircraft fuel expense ranged from 22% to 35% of operating expenses. Fuel prices are impacted by changes in both the price of crude oil and refining margins, and can vary by region in the U.S.

The average annual price of crude oil in the last five years has ranged from a low of \$49 per barrel in 2015 from a high of \$98 in 2013. For us, a \$1 per barrel change in the price of oil equates to approximately \$12 million of fuel cost annually. Said another way, a one-cent change in our fuel price per gallon will impact our expected annual fuel cost by approximately \$5 million per year.

Refining margins, which represent the price of refining crude oil into aircraft fuel, are a smaller portion of the overall price of jet fuel, but also contributed to the price volatility in recent years. Average annual refining margin prices have fluctuated between \$20 per barrel and \$36 per barrel in the last five years, and averaged \$20 in 2015.

Generally, West Coast jet fuel prices are somewhat higher and more volatile than prices in the Gulf Coast or on the East Coast. Our average raw fuel cost per gallon decreased 39% in 2015, decreased 6% in 2014, and 4% in 2013.

The percentages of our aircraft fuel expense by crude and refining margins, as well as the percentage of our aircraft fuel expense of operating expenses are as follows:

	2015	2014	2013	2012	2011
Crude oil	62%	72%	71%	65%	70%
Refining margins	26%	18%	19%	25%	24%
Other ^(a)	12%	10%	10%	10%	6%
Total	100%	100%	100%	100%	100%
Aircraft fuel expense	22%	32%	34%	35%	34%

(a) Other includes gains and losses on settled fuel hedges, unrealized mark-to-market fuel hedge gains or losses, taxes and other into-plane costs.

We use crude oil call options as hedges to decrease our exposure to the volatility of jet fuel prices. Historically, we have had jet fuel refining margin swap contracts, but we discontinued the use of the refining margin swaps in the third quarter of 2014. Call options effectively cap our pricing for crude oil, limiting our exposure to increasing fuel prices for about half of our planned fuel consumption. With call options, we are hedged against spikes in crude oil prices, and during a period of decline in crude oil prices, we only forfeit cash previously paid for hedge premiums. Currently, we start hedging approximately 18 months in advance of crude oil consumption.

We believe that operating fuel-efficient aircraft is the best hedge against high fuel prices. Alaska operates an all-Boeing 737 fleet and Horizon operates an all-Bombardier Q400 turboprop fleet. Air Group's fuel-efficiency rate expressed in available seat miles flown per gallon (ASMs/g) improved from 74.4 ASMs/g in 2011 to 78.6 ASMs/g in 2015. These improvements have not only reduced our fuel consumption rate, but also the amount of greenhouse gases and other pollutants that our operations emit.

COMPETITION

Competition in the airline industry is intense and unpredictable. Our competitors consist primarily of other airlines and, to a lesser extent, other forms of transportation. Competition can be direct, in the form of another carrier flying the exact non-stop route, or indirect, where a carrier serves the same two cities non-stop from an alternative airport in that city or via an itinerary requiring a connection at another airport. We compete with other domestic airlines, and a limited number of international airlines, on nearly all of our scheduled routes. Our largest competitor is Delta Air Lines, who has significantly increased their capacity in Seattle over the past two years. Approximately 60% of our capacity to and from Seattle competes with Delta. Based on schedules filed with the U.S. Department of Transportation, we expect the amount of competitive capacity overlap with all carriers to increase more than 13% in the first half of 2016, weighted based on our network.

We believe that the following principal competitive factors are important to our customers:

- Safety record
- Customer service and reputation

We compete with other airlines in areas of customer service such as on-time performance, passenger amenities - including first class seating, quality of on-board products, aircraft type, and comfort. In 2015, Alaska Airlines ranked "Highest in Customer Satisfaction among Traditional Network Carriers" by J.D. Power and Associates for the eighth year in a row. All of our 2015 mainline aircraft deliveries included the Boeing Sky Interior, our Alaska Beyond™ in-flight experience, which features our streaming in-flight entertainment, gourmet food designed by Tom Douglas, and comfortable seats with additional space and power as part of our exceptional, above and beyond flight experience. In 2015, we introduced Preferred Plus seating and Boeing's new Space Bins™. Preferred Plus gives customers the option to upgrade to bulkhead and exit row seats at check-in for a nominal fee and includes early boarding and a free cocktail. Space Bins™ provide space for up to 50% more carry-on bags on board our aircraft, providing a more hassle-free boarding experience for our passengers. In 2016, we will be launching a Premium Class of service on our airplanes that will provide extra legroom, early boarding, premium snacks and complimentary alcoholic beverages.

• Fares and ancillary services

The pricing of fares is a significant competitive factor in the airline industry, and the increased availability of fare information on the Internet allows travelers to easily compare fares and identify competitor promotions and discounts. Pricing is driven by a variety of factors including, but not limited to, market-specific capacity, market share per route/geographic area, cost structure, fare vs. ancillary revenue strategies, and demand.

For example, airlines often discount fares to drive traffic in new markets or to stimulate traffic when necessary to improve load factors. In addition, traditional network carriers have been able to reduce their operating costs through bankruptcies and mergers, while low-cost carriers have continued to grow their fleets and expand their networks, potentially enabling them to better control costs per available seat mile (the average cost to fly an aircraft seat one mile), which in turn may enable them to lower their fares. These factors can reduce our pricing power and that of the airline industry as a whole.

Domestic airline capacity is dominated by four large carriers, representing approximately 85% of total seats. Accordingly, if these carriers discount their fares or enter into our core markets, we must match those fares in order to maintain our load factors, often resulting in year-over-year decreases

in our yields. We will defend our core markets vigorously and, if necessary, redeploy capacity to better match supply with demand. We believe the restructuring we've completed over the past decade has decreased our costs, enabling us to offer competitive fares while still earning appropriate returns for our shareholders.

Routes served, flight schedules, codesharing and interline relationships, and frequent flier programs

We also compete with other airlines based on markets served, the frequency of service to those markets, and frequent flier opportunities. Some airlines have more extensive route structures than we do, and they offer significantly more international routes. In order to expand opportunities for our customers, we enter into codesharing and interline relationships with other airlines that provide reciprocal frequent flier mileage credit and redemption privileges. These relationships allow us to offer our customers access to more destinations than we can on our own, gain exposure in markets we don't serve and allow our customers more opportunities to earn and redeem frequent flier miles. Our Mileage PlanTM offers one of the most comprehensive benefits to our members with the ability to earn and redeem miles on 16 of our partner carriers.

In addition to domestic or foreign airlines that we compete with on most of our routes, we compete with ground transportation in our short-haul markets. Both carriers, to some extent, also compete with technology such as video conferencing and internet-based meeting tools that have changed the need for, or frequency of face-to-face business meetings.

TICKET DISTRIBUTION

Our tickets are distributed through three primary channels:

- *Alaskaair.com:* It is less expensive for us to sell through this direct channel and, as a result, we continue to take steps to drive more business to our website. In addition, we believe this channel is preferable from a branding and customer-relationship standpoint in that we can establish ongoing communication with the customer and tailor offers accordingly.
- Traditional and online travel agencies: Both traditional and online travel agencies typically use Global Distribution Systems (GDS) to obtain their fare and inventory data from airlines. Bookings made through these agencies result in a fee that is charged to the airline. Many of our large corporate customers require us to use these agencies. Some of our competitors do not use this distribution channel and, as a result, have lower ticket distribution costs.
- *Reservation call centers:* These call centers are located in Phoenix, AZ, Kent, WA, and Boise, ID. We generally charge a \$15 fee for booking reservations through these call centers.

Our sales by channel are as follows:

	2015	2014	2013	2012	2011
Alaskaair.com	60%	57%	55%	54%	51%
Traditional agencies	23%	25%	27%	27%	28%
Online travel agencies	11%	12%	13%	13%	13%
Reservation call centers	6%	6%	5%	6%	8%
Total	100%	100%	100%	100%	100%

SEASONALITY AND OTHER FACTORS

Our results of operations for any interim period are not necessarily indicative of those for the entire year because our business is subject to seasonal fluctuations. Our profitability is generally lowest during the first and fourth quarters due principally to fewer departures and passengers. Profitability typically increases in the second quarter and then reaches its highest level during the third quarter as a result of vacation travel, including increased activity in the state of Alaska. However, we have taken steps over the past few years to better manage the seasonality of our operations by adding flights to leisure destinations, like Hawaii, and expanding to cities in the mid-continental and eastern U.S.

In addition to passenger loads, factors that could cause our quarterly operating results to vary include:

- general economic conditions and resulting changes in passenger demand,
- changes in fuel costs,
- · pricing initiatives by us or our competitors,
- · increases in competition at our primary airports, and
- increases or decreases in passenger and volume-driven variable costs.

Many of the markets we serve experience inclement weather conditions in the winter, causing increased costs associated with deicing aircraft, canceling flights, and accommodating displaced passengers. Due to our geographic area of operations, we can be more susceptible to adverse weather conditions, particularly in the state of Alaska and the Pacific Northwest, than some of our competitors, who may be better able to spread weather-related risks over larger route systems.

No material part of our business or that of our subsidiaries is dependent upon a single customer, or upon a few high-volume customers.

EMPLOYEES

Our business is labor intensive. As of December 31, 2015, we employed 15,143 (11,614 at Alaska and 3,529 at Horizon) active full-time and part-time employees. Wages and benefits, including variable incentive pay, represented approximately 41% of our total non-fuel operating expenses in both 2015 and

2014.

Most major airlines, including ours, have employee groups that are covered by collective bargaining agreements. Airlines with unionized work forces generally have higher labor costs than carriers without unionized work forces, and they may not have the ability to adjust labor costs downward quickly enough to respond to new competition or slowing demand. At December 31, 2015, labor unions represented 83% of Alaska's and 44% of Horizon's employees. Our relations with U.S. labor organizations are governed by the Railway Labor Act (RLA). Under this act, collective bargaining agreements do not expire but instead become amendable as of a stated date. If either party wishes to modify the terms of any such agreement, it must notify the other party in the manner prescribed by the RLA and/or described in the agreement. After receipt of such notice, the parties must meet for direct negotiations, and if no agreement is reached, either party may request the National Mediation Board (NMB) to initiate a process including mediation, arbitration, and a potential "cooling off" period that must be followed before either party may engage in self-help.

Alaska's union contracts at December 31, 2015 were as follows:

Union	Employee Group	Number of Employees	Contract Status
Air Line Pilots Association International (ALPA)	Pilots	1,697	Amendable 03/31/2018
Association of Flight Attendants (AFA)	Flight attendants	3,660	Amendable 12/17/2019
International Association of Machinists and Aerospace Workers (IAM)	Ramp service and stock clerks	625	Amendable 7/19/2018
IAM	Clerical, office and passenger service	2,921	Amendable 1/1/2019
Aircraft Mechanics Fraternal Association (AMFA)	Mechanics, inspectors and cleaners	665	Amendable 10/17/2016
Mexico Workers Association of Air Transport	Mexico airport personnel	85	Amendable 9/29/2016
Transport Workers Union of America (TWU)	Dispatchers	44	Amendable 3/24/2019

Horizon's union contracts at December 31, 2015 were as follows:

Union	Employee Group	Number of Employees	Contract Status
International Brotherhood of Teamsters (IBT)	Pilots ⁽¹⁾	643	Amendable 12/14/2018
AFA	Flight attendants (1)	596	Amendable 07/18/2018
IBT	Mechanics and related classifications	272	Amendable 12/16/2020
National Automobile, Aerospace, Transportation and General Workers	Station personnel in Vancouver and Victoria, BC, Canada	42	Amendable 8/26/2018
Transportation Workers Union of America	Dispatchers	17	Amendable 2/14/2016

⁽¹⁾ Horizon pilots and flight attendants ratified new agreements subsequent to December 31, 2015. The Flight Attendant agreement now becomes amendable in July 2019 and the Pilot agreement becomes amendable in December 2024.

9

EXECUTIVE OFFICERS

The executive officers of Alaska Air Group, Inc. and executive officers of Alaska and Horizon who have significant decision-making responsibilities, their positions and their respective ages are as follows:

Name	Position	Age	Air Group or Subsidiary Officer Since
Bradley Tilden	Chairman, President and Chief Executive Officer of Alaska Air Group, Inc. and Alaska Airlines, Inc. and Chief Executive Officer of Horizon Air Industries, Inc.	55	1994
Benito Minicucci	Executive Vice President/Operations and Chief Operating Officer of Alaska Airlines, Inc.	49	2004
Brandon Pedersen	Executive Vice President/Finance and Chief Financial Officer of Alaska Air Group, Inc. and Alaska Airlines, Inc.	49	2003
Andrew Harrison	Executive Vice President and Chief Commercial Officer of Alaska Airlines, Inc.	45	2008
David Campbell	President and Chief Operating Officer of Horizon Air Industries, Inc.	54	2014
Herman Wacker	Former Vice President of Legal and General Counsel of Alaska Air Group, Inc. and Alaska Airlines, Inc., and Chief Ethics and Compliance officer at Alaska Air Group, Inc.	67	2014
Kyle Levine	Vice President Legal and General Counsel of Alaska Air Group, Inc. and Alaska Airlines, Inc. and Chief Ethics and Compliance Officer of Alaska Air Group, Inc.	44	2016

Mr. Tilden joined Alaska Airlines in 1991, became Controller of Alaska Air Group and Alaska Airlines in 1994, Vice President/Finance in January 1999, Executive Vice President/Finance and Chief Financial Officer in January 2002, Executive Vice President/Finance and Planning in April 2007, and President of Alaska Airlines in December 2008. He leads Air Group's Management Executive Committee and was elected to the Air Group Board in 2010. He was elected Chief Executive Officer of Alaska Airlines and Horizon Air in May 2012, and became Chairman of the Board in January 2014.

Mr. Minicucci joined Alaska Airlines in 2004 as Staff Vice President of Maintenance and Engineering and was promoted to Vice President of Seattle Operations in June 2008. He was elected Executive Vice President/Operations and Chief Operating Officer of Alaska Airlines in December 2008. He is a member of Air Group's Management Executive Committee.

Mr. Pedersen joined Alaska Airlines in 2003 as Staff Vice President/Finance and Controller of Alaska Air Group and Alaska Airlines and was elected Vice President/Finance and Controller for both entities in 2006. He was elected Chief Financial Officer of Alaska Air Group and Alaska Airlines in June 2010, and elected as Executive Vice President/Finance and Chief Financial Officer in 2014. He is a member of Air Group's Management Executive Committee.

Mr. Harrison joined Alaska Airlines in 2003 as the Managing Director of Internal Audit and was elected Vice President of Planning and Revenue Management in 2018. He was elected Senior Vice President of Planning and Revenue Management in 2014. He was elected Executive Vice President and Chief Commercial Officer in February 2015. He is a member of Air Group's Management Executive Committee.

Mr. Campbell joined Horizon Air in 2014 as President and Chief Operating Officer. Prior to joining Horizon Air, Mr. Campbell served more than 25 years in maintenance and flight operations. Most recently, he served as the vice president of maintenance and engineering at jetBlue Airways from January 2014 to August 2014, and prior to that, he served as vice president of safety and operational performance at American Airlines. He joined American in 1988 after serving for four years in the U.S. Air

10

Force and has overseen maintenance, quality, technical operations and safety. He is a member of Air Group's Management Executive Committee.

Mr. Wacker joined Alaska Airlines in 2007 as Managing Director of Labor & Employment Law and Associate General Counsel. Mr. Wacker was elected Vice President of Legal at Alaska Air Group from February 2014 to December 2015, and General Counsel from October 2014 to December 2015. He was also appointed Chief Ethics and Compliance Officer at Air Group from May 2014 to December 2015. He was a member of Air Group's Management Executive Committee until his retirement in December 2015.

Mr. Levine was elected Vice President Legal and General Counsel of Alaska Air Group and Alaska Airlines effective January 1, 2016 and is a member of Air Group's Management Executive Committee. He joined Alaska Airlines in February 2006 as a Senior Attorney. At Alaska Airlines, he also served as Deputy General Counsel and Managing Director of Legal from February 2011 to January 2016, and as Associate General Counsel and Managing Director Commercial Law and General Litigation from July 2009 to February 2011. He was appointed Assistant Corporate Secretary of Air Group and Alaska Airlines in February 2015.

REGULATION

GENERAL

The airline industry is highly regulated, most notably by the federal government. The Department of Transportation (DOT), the Federal Aviation Administration (FAA) and the Transportation Security Administration (TSA) exercise significant regulatory authority over air carriers.

- DOT: In order to provide passenger and cargo air transportation in the U.S., a domestic airline is required to hold a certificate of public convenience and necessity issued by the DOT. Subject to certain individual airport capacity, noise and other restrictions, this certificate permits an air carrier to operate between any two points in the U.S. Certificates do not expire, but may be revoked for failure to comply with federal aviation statutes, regulations, orders or the terms of the certificates. While airlines are permitted to establish their own fares without governmental regulation, the DOT has jurisdiction over the approval of international codeshare agreements, marketing alliance agreements between major domestic carriers, international and some domestic route authorities, Essential Air Service market subsidies, carrier liability for personal or property damage, and certain airport rates and charges disputes. International treaties may also contain restrictions or requirements for flying outside of the U.S. and impose different carrier liability limits than those applicable to domestic flights. The DOT has been active in implementing a variety of "consumer protection" regulations, covering subjects such as advertising, passenger communications, denied boarding compensation and tarmac delay response. Airlines are subject to enforcement actions that are brought by the DOT from time to time for alleged violations of consumer protection and other economic regulations. We are not aware of any enforcement proceedings that could either materially affect our financial position or impact our authority to operate.
- *FAA*: The FAA, through Federal Aviation Regulations (FARs), generally regulates all aspects of airline operations, including establishing personnel, maintenance and flight operation standards. Domestic airlines are required to hold a valid air carrier operating certificate issued by the FAA. Pursuant to these regulations we have established, and the FAA has approved, our operations specifications and a maintenance program for each type of aircraft we operate. The maintenance program provides for the ongoing maintenance of such aircraft, ranging from frequent routine inspections to major overhauls. From time to time the FAA issues airworthiness directives (ADs) that must be incorporated into our aircraft maintenance program and operations. All airlines are subject to enforcement actions that are brought by the FAA from time to time for alleged violations of FARs or ADs. At this time, we are not aware of any enforcement proceedings that could either materially affect our financial position or impact our authority to operate.
- *TSA*: Airlines serving the U.S. must operate a TSA-approved Aircraft Operator Standard Security Program (AOSSP), and comply with TSA Security Directives (SDs) and regulations. Airlines are subject to enforcement actions that are brought by the TSA from time to time for alleged violations of the AOSSP, SDs or security regulations. We are not aware of any enforcement proceedings that could either materially affect our financial position or impact our authority to operate. Under TSA authority, we are required to collect a September 11 Security Fee of \$5.60 per one-way trip from passengers and remit that sum to the government to fund aviation security measures.

The Department of Justice and DOT have jurisdiction over airline antitrust matters. The U.S. Postal Service has jurisdiction over certain aspects of the transportation of mail and related services. Labor relations in the air transportation industry are regulated under the Railway Labor Act. To the extent we continue to fly to foreign countries and pursue alliances with international carriers, we may be subject to certain regulations of foreign agencies and international treaties.

ENVIRONMENTAL AND OCCUPATIONAL SAFETY MATTERS

We are subject to various laws and government regulations concerning environmental matters and employee safety and health in the U.S. and other countries. U.S. federal laws that have a particular effect on us include the Airport Noise and Capacity Act of 1990, the Clean Air Act, the Resource Conservation and Recovery Act, the Clean Water Act, the Safe Drinking Water Act, and the Comprehensive Environmental Response, Compensation and Liability Act, Superfund Amendments and Reauthorization Act, and the Oil Pollution Control Act. We are also subject to the oversight of the Occupational Safety and Health Administration (OSHA) concerning employee safety and health matters. The U.S. Environmental Protection Agency, OSHA, and other federal agencies have been authorized to create and enforce regulations that have an impact on our operations. In addition to these federal activities, various states have been delegated certain authorities under these federal statutes. Many state and local governments have adopted environmental and employee safety and health laws and regulations. We maintain our safety, health and environmental programs in order to meet or exceed these requirements.

We expect there will be legislation in the future to reduce carbon and other greenhouse gas emissions. Alaska and Horizon have transitioned to more fuelefficient aircraft fleets.

The Airport Noise and Capacity Act recognizes the rights of airport operators with noise problems to implement local noise abatement programs so long as they do not interfere unreasonably with interstate or foreign commerce or the national air transportation system. Authorities in several cities have established aircraft noise reduction programs, including the imposition of nighttime curfews. We believe we have sufficient scheduling flexibility to accommodate local noise restrictions.

Although we do not currently anticipate that these regulatory matters, individually or collectively, will have a material effect on our financial condition, results of operations or cash flows, new regulations or compliance issues that we do not currently anticipate could have the potential to harm our financial

condition, results of operations or cash flows in future periods.

INSURANCE

We carry insurance of types customary in the airline industry and in amounts deemed adequate to protect our interests and property and to comply both with federal regulations and certain credit and lease agreements. The insurance policies principally provide coverage for Airline Hull, Spares and Comprehensive Legal Liability Insurance, War and Allied Perils, and Workers' Compensation. In addition, we currently carry a Cyber Liability policy in the event of security breaches from malicious parties.

We believe that our emphasis on safety and our state-of-the-art flight deck safety technology help to control the cost of our insurance.

WHERE YOU CAN FIND MORE INFORMATION

Our filings with the Securities and Exchange Commission, including our annual report on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and amendments to those reports are available on our website at *www.alaskaair.com*, free of charge, as soon as reasonably practicable after the electronic filing of these reports with the Securities and Exchange Commission. The information contained on our website is not a part of this annual report on Form 10-K.



ITEM 1A. RISK FACTORS

If any of the following occurs, our business, financial condition and results of operations could suffer. In such case, the trading price of our common stock could also decline. We operate in a continually changing business environment. In this environment, new risks may emerge and already identified risks may vary significantly in terms of impact and likelihood of occurrence. Management cannot predict such developments, nor can it assess the impact, if any, on our business of such new risk factors or of events described in any forward-looking statements.

We have adopted an enterprise wide Risk Analysis and Oversight Program designed to identify the various risks faced by the organization, assign responsibility for managing those risks to individual executives as well as align these risks with Board oversight. These enterprise-level identified risks have been aligned to the risk factors discussed below.

SAFETY, COMPLIANCE AND OPERATIONAL EXCELLENCE

Our reputation and financial results could be harmed in the event of an airline accident or incident.

An accident or incident involving one of our aircraft or an aircraft operated by one of our codeshare partners or CPA carriers could involve a significant loss of life and result in a loss of confidence in our airlines by the flying public and/or aviation authorities. We could experience significant claims from injured passengers, by-standers and surviving relatives, as well as costs for the repair or replacement of a damaged aircraft and its consequential temporary or permanent loss from service. We maintain liability insurance in amounts and of the type generally consistent with industry practice, as do our codeshare partners and CPA carriers. However, the amount of such coverage may not be adequate to fully cover all claims and we may be forced to bear substantial economic losses from an accident. Substantial claims resulting from an accident in excess of our related insurance coverage would harm our business and financial results. Moreover, any aircraft accident or incident, even if fully insured and even if it does not involve one of our aircraft, could cause a public perception that our airlines or the aircraft we or our partners fly are less safe or reliable than other transportation alternatives, which would harm our business.

Our operations are often affected by factors beyond our control, including delays, cancellations, and other conditions, which could harm our business, financial condition and results of operations.

Like other airlines, our operations often are affected by delays, cancellations and other conditions caused by factors largely beyond our control.

Other conditions that might impact our operations include:

- lack of operational approval (e.g. new routes, aircraft deliveries, etc.);
- congestion and/or space constraints at airports or air traffic control problems;
- adverse weather conditions;
- increased security measures or breaches in security;
- contagious illness and fear of contagion;
- changes in international treaties concerning air rights;
- international or domestic conflicts or terrorist activity; and
- other changes in business conditions.

Due to our concentration of flights in the Pacific Northwest and Alaska, we believe a large portion of our operation is more susceptible to adverse weather conditions. A general reduction in airline passenger traffic as a result of any of the above-mentioned factors could harm our business, financial condition and results of operations.

Changes in government regulation imposing additional requirements and restrictions on our operations could increase our operating costs and result in service delays and disruptions.

Airlines are subject to extensive regulatory and legal requirements, both domestically and internationally, that involve significant compliance costs. In the last several years, Congress has passed laws, and the U.S. DOT, the TSA and the FAA have issued regulations that have required significant expenditures relating to the maintenance and operation of airlines and establishment of consumer protections.

Similarly, there are a number of legislative and regulatory initiatives and reforms at the federal, state, and local level, including increasingly stringent laws protecting the environment, minimum wage requirements, and health care mandates that could affect our relationship with our workforce and the vendors that serve our airlines, and cause our expenses to increase without an ability to pass through these costs.

Almost all commercial service airports are owned and/or operated by units of local or state governments. Airlines are largely dependent on these governmental entities to provide adequate airport facilities and capacity at an affordable cost. Many airports have increased their rates and charges to air carriers related to higher security costs, increased costs related to updated infrastructures, and other costs. Additional laws, regulations, taxes, and airport rates and charges have been proposed from time to time that could significantly increase the cost of airline operations or reduce the demand for air travel. Although lawmakers may impose these additional fees and view them as "pass-through" costs, we believe that a higher total ticket price will influence consumer purchase and travel decisions and may result in an overall decline in passenger traffic, which would harm our business.

The airline industry continues to face potential security concerns and related costs.

Terrorist attacks, the fear of such attacks or other hostilities involving the U.S. could have a significant negative effect on the airline industry, including us, and could:

- significantly reduce passenger traffic and yields as a result of a potentially dramatic drop in demand for air travel;
- significantly increase security and insurance costs;
- make war risk or other insurance unavailable or extremely expensive;
- increase fuel costs and the volatility of fuel prices;
- · increase costs from airport shutdowns, flight cancellations and delays resulting from security breaches and perceived safety threats; and
- result in a grounding of commercial air traffic by the FAA.

The occurrence of any of these events would harm our business, financial condition and results of operations.

We rely on third-party vendors for certain critical activities.

We have historically relied on outside vendors for a variety of services and functions critical to our business, including airframe and engine maintenance, ground handling, fueling, computer reservation system hosting, telecommunication systems, and information technology infrastructure and services. As part of our cost-reduction efforts, our reliance on outside vendors has increased and may continue to do so in the future, especially since we rely on timely and effective third-party performance in conjunction with many of our technology-related initiatives.

Even though we strive to formalize agreements with these vendors that define expected service levels, our use of outside vendors increases our exposure to several risks. In the event that one or more vendors go into bankruptcy, ceases operation or fails to perform as promised, replacement services may not be readily available at competitive rates, or at all. If one of our vendors fails to perform adequately, we may experience increased costs, delays, maintenance issues, safety issues or negative public perception of our airline. Vendor bankruptcies, unionization, regulatory compliance issues or significant changes in the competitive marketplace among suppliers could adversely affect vendor services or force us to renegotiate existing agreements on less favorable terms. These events could result in disruptions in our operations or increases in our cost structure.



STRATEGY

The airline industry is highly competitive and susceptible to price discounting and changes in capacity, which could have a material adverse effect on the Company. If we cannot successfully compete in the marketplace, our business, financial condition and operating results will be materially adversely affected.

The U.S. airline industry is characterized by substantial price competition. In recent years, the market share held by low-cost carriers and so-called ultra lowcost carriers has increased significantly and is expected to continue to increase. Airlines also compete for market share by increasing or decreasing their capacity, including route systems and the number of markets served. Several of our competitors have increased their capacity in markets we serve, particularly on the West Coast and in our Seattle hub, therefore increasing competition for those destinations. This increased competition in both domestic and international markets may have a material adverse effect on the Company's results of operations, financial condition or liquidity.

We continue to strive toward aggressive cost-reduction goals that are an important part of our business strategy of offering the best value to passengers through competitive fares while achieving acceptable profit margins and return on capital. If we are unable to reduce our costs over the long-term and achieve sustained targeted returns on invested capital, we will likely not be able to grow our business in the future or weather industry downturns and therefore our financial results may suffer.

The airline industry may undergo further restructuring, consolidation, or the creation or modification of alliances or joint ventures, any of which could have a material adverse effect on our business, financial condition and results of operations.

We continue to face strong competition from other carriers due to restructuring, consolidation, and the creation and modification of alliances and joint ventures. Since deregulation, both the U.S. and international airline industries have experienced consolidation through a number of mergers and acquisitions. Carriers may also improve their competitive positions through airline alliances, slot swaps/acquisitions, and/or joint ventures. Certain airline joint ventures further competition by allowing airlines to coordinate routes, pool revenues and costs, and enjoy other mutual benefits, achieving many of the benefits of consolidation.

We depend on a few key markets to be successful.

Our strategy is to focus on serving a few key markets, including Seattle, Los Angeles, Anchorage, Portland, and Hawaii . A significant portion of our flights occur to and from our Seattle hub. In 2015, passengers to and from Seattle accounted for 61% of our total passengers.

We believe that concentrating our service offerings in this way allows us to maximize our investment in personnel, aircraft, and ground facilities, as well as to gain greater advantage from sales and marketing efforts in those regions. As a result, we remain highly dependent on our key markets. Our business could be harmed by any circumstances causing a reduction in demand for air transportation in our key markets. An increase in competition in our key markets could also cause us to reduce fares or take other competitive measures that if sustained could harm our business, financial condition and results of operations.

Economic uncertainty or another recession would likely impact demand for our product and could harm our financial condition and results of operations.

The airline industry, which is subject to relatively high fixed costs and highly variable and unpredictable demand, is particularly sensitive to changes in economic conditions. We are also highly dependent on U.S. consumer confidence and the health of the U.S. economy. Unfavorable U.S. economic conditions have historically driven changes in travel patterns and have resulted in reduced spending for both leisure and business travel. For some consumers, leisure travel is a discretionary expense, and shorthaul travelers, in particular, have the option to replace air travel with surface travel. Businesses are able to forgo air travel by using communication alternatives such as videoconferencing or may be more likely to purchase less expensive tickets to reduce costs, which can result in a decrease in average revenue per seat. Unfavorable economic conditions also hamper the ability of airlines to raise fares to counteract increased fuel, labor, and other costs. Unfavorable or even uncertain economic conditions could negatively affect our financial condition and results of operations.

We are dependent on a limited number of suppliers for aircraft and parts.

Alaska is dependent on Boeing as its sole supplier for aircraft and many aircraft parts. Horizon is similarly dependent on Bombardier. Additionally, each carrier is dependent on sole suppliers for aircraft engines. As a result, we are more vulnerable to any problems associated with the supply of those aircraft and parts, including design defects, mechanical problems, contractual performance by the manufacturers, or adverse perception by the public that would result in customer avoidance or in actions by the FAA resulting in an inability to operate our aircraft.

We rely on partner airlines for codeshare and frequent flier marketing arrangements.

Alaska and Horizon are parties to marketing agreements with a number of domestic and international air carriers, or "partners," including, but not limited to, American Airlines and Delta Air Lines. These agreements provide that certain flight segments operated by us are held out as partner "codeshare" flights and that certain partner flights are held out for sale as Alaska codeshare flights. In addition, the agreements generally provide that members of Alaska's Mileage PlanTM program can earn miles on or redeem miles for partner flights and vice versa. We receive revenue from flights sold under codeshare and from interline arrangements. In addition, we believe that the frequent flier arrangements are an important part of our Mileage PlanTM program. The loss of a significant partner through bankruptcy, consolidation, or otherwise, could have a negative effect on our revenues or the attractiveness of our Mileage PlanTM, which we believe is a source of competitive advantage.

There is ongoing speculation that further airline consolidation or reorganization could occur in the future. We routinely engage in analysis and discussions regarding our own strategic position, including alliances, codeshare arrangements, interline arrangements, frequent flier program enhancements, and may have future discussions with other airlines regarding similar activities. If other airlines participate in consolidation or reorganization, those airlines may significantly improve their cost structures or revenue generation capabilities, thereby potentially making them stronger competitors of ours and potentially impairing our ability to realize expected benefits from our own strategic relationships.

INFORMATION TECHNOLOGY

We rely heavily on automated systems to operate our business, and a failure to invest in new technology, or a disruption of our current systems or their operators could harm our business.

We depend on automated systems to operate our business, including our airline reservation system, our telecommunication systems, our website, our maintenance systems, our check-in kiosks, mobile devices and other systems. Substantially all of our tickets are issued to passengers as electronic tickets and the majority of our customers check in using our website or our airport kiosks. We depend on our reservation system to be able to issue, track and accept these electronic tickets. In order for our operations to work efficiently, we must continue to invest in new technology to ensure that our website, reservation system, and check-in systems are able to accommodate a high volume of traffic, maintain secure information, and deliver important flight information. Substantial or repeated website, reservations system or telecommunication systems failures or service disruptions could reduce the attractiveness of our services and cause our customers to do business with another airline. In addition, we rely on other automated systems for crew scheduling, flight dispatch, and other operational needs. In 2016, we expect to migrate to a new crew management system. We also plan to move our primary data center location. Disruptions, failed migration, untimely recovery, or a breach of these systems or the data center could result in the loss of important data, an increase of our expenses, an impact on our operational performance, or a possible temporary cessation of our operations.

If we do not maintain the privacy and security of our information, we could damage our reputation and incur substantial legal and regulatory costs.

We accept, store, and transmit information about our customers, our employees, our business partners and our business. In addition, we frequently rely on third-party hosting sites and data processors, including cloud providers. Our sensitive information relies on secure transmission over public and private networks. A compromise of our systems, the security of our infrastructure, or those of other business partners that result in our information being accessed or stolen by unauthorized persons could adversely affect our operations and our reputation.



FINANCIAL CONDITION AND FINANCIAL MARKETS

Our business, financial condition, and results of operations are substantially exposed to the volatility of jet fuel prices. Significant increases in jet fuel costs would harm our business.

Fuel costs constitute a significant portion of our total operating expenses Future increases in the price of jet fuel may harm our business, financial condition and results of operations, unless we are able to increase fares and fees, or add additional ancillary services to attempt to recover increasing fuel costs.

Certain of the Company's financing agreements have covenants that impose operating and financial restrictions on the Company and its subsidiaries.

Certain of our credit facilities and indentures governing our secured borrowings impose certain operating and financial covenants on us. Such covenants require us to maintain, depending on the particular agreement, minimum liquidity and/or minimum collateral coverage ratios, and other negative covenants customary for such financings. A decline in the value of collateral could result in a situation where we may not be able to maintain the required collateral coverage ratio.

Our ability to comply with these covenants may be affected by events beyond our control, including the overall industry revenue environment and the level of fuel costs, and we may be required to seek waivers or amendments of covenants, repay all or a portion of the debt or find alternative sources of financing.

Our maintenance costs will increase as our fleet ages, and we will periodically incur substantial maintenance costs due to the maintenance schedules of our aircraft fleet.

As of December 31, 2015, the average age of our NextGen aircraft (B737-800, -900, -900ERs) was approximately 6.4 years, and the average age of our Q400 aircraft was approximately 9 years. Our relatively new aircraft require less maintenance now than they will in the future. Any significant increase in maintenance expenses could have a material adverse effect on our results of operations.

BRAND AND REPUTATION

As we evolve our brand to appeal to a changing demographic and grow into new markets, we will engage in strategic initiatives that may not be favorably received by all customers.

We continue to focus on strategic initiatives designed to increase our brand appeal to a diverse and evolving demographic of airline travelers. These efforts could include significant improvements to our in-airport and on-board environments, increasing our direct customer relationships through improvements to our purchasing portals (digital and mobile), and optimization of our customer loyalty programs.

In pursuit of these efforts we may negatively affect our reputation with some of our existing customer base.

LABOR RELATIONS AND LABOR STRATEGY

A significant increase in labor costs, unsuccessful attempts to strengthen our relationships with union employees, or loss of key personnel could adversely affect our business and results of operations.

Labor costs are a significant component of our total expenses. Each of our represented employee groups has a separate collective bargaining agreement, and could make demands that would increase our operating expenses and adversely affect our financial performance if we agree to them. The same result could apply if we experience a significant increase in vendor labor costs, including wage rate increases, that ultimately flow through to us.

As of December 31, 2015, labor unions represented approximately 83% of Alaska's and 44% of Horizon's employees. Although we have been successful in maturing communications, negotiating approaches, and other strategies to enhance workforce engagement in the Company's long-term vision, future uncertainty around open contracts could be a distraction, affecting employee focus in our business and diverting management's attention from other projects and issues.

We compete against the major U.S. airlines and other businesses for labor in many highly skilled positions. If we are unable to hire, train and retain qualified employees at a reasonable cost, sustain employee engagement in the Company's strategic vision,



or if we are unsuccessful at implementing succession plans for our key staff, we may be unable to grow or sustain our business. In recent years, there have been pilot shortages in the regional market. Attrition beyond normal levels could negatively impact our operating results and our business prospects could be harmed.

ITEM 1B. UNRESOLVED STAFF COMMENTS

None

ITEM 2.	PROPERTIES
AIRCRAFT	

The following table describes the aircraft we operate and their average age at December 31, 2015:

					Average Age in
Aircraft Type	Seats	Owned	Leased	Total	Years
B737 Freighters & Combis	0/72	6		6	22.2
B737-400/700	144/124	17	17	34	17.7
B737-800/900/900ER	163/181/181	97	10	107	6.4
Total Mainline Fleet		120	27	147	9.7
Q400	76	37	15	52	9.0
E175	76	—	5	5	0.5
CRJ-700 ^(a)	70	2	6	8	13.3
Total Regional Fleet		39	26	65	8.9
Total		159	53	212	9.5

(a) In addition to the CRJ-700s in our operating fleet, we have eight leased CRJ-700s currently subleased to a third party operated for other carriers.

"Management's Discussion and Analysis of Financial Condition and Results of Operations" discusses future orders and options for additional aircraft.

70 of our owned aircraft secure long-term debt arrangements or collateralize our revolving credit facility. See further discussion in "Liquidity and Capital Resources."

Alaska's leased B737 aircraft have lease expiration dates between 2016 and 2023. Horizon's leased Q400 aircraft have expiration dates in 2018. The leases on the six CRJ-700 aircraft have expiration dates between 2018 and 2020, and the leased E175 aircraft are through our capacity purchase arrangement with SkyWest. Alaska and Horizon have the option to extend most of the leases for additional periods, or the right to purchase the aircraft at the end of the lease term, usually at the then-fair-market value of the aircraft.

GROUND FACILITIES AND SERVICES

We own terminal buildings in various cities in the state of Alaska and several buildings located at or near Seattle-Tacoma International Airport (Sea-Tac) near Seattle, WA. These include a multi-bay hangar and shops complex (used primarily for line maintenance), a flight operations and training center, an air cargo facility, an information technology office and datacenter, and various other commercial office buildings.

We lease ticket counters, gates, cargo and baggage space, ground equipment, office space, and other support areas at the majority of the airports we serve. We also lease operations, training, and aircraft maintenance facilities in Portland and Spokane, as well as line maintenance stations in Boise, Bellingham, Eugene, San Jose, Medford, Redmond, Seattle, and Spokane. Further, we lease call center facilities in Phoenix and Boise.

ITEM 3. LEGAL PROCEEDINGS

We are a party to routine litigation matters incidental to our business. Management believes the ultimate disposition of these matters is not likely to materially affect our financial position or results of operations. This forward-looking statement is based on management's current understanding of the relevant law and facts, and it is subject to various contingencies, including the potential costs and risks associated with litigation and the actions of judges and juries.

ITEM 4. MINE SAFETY DISCLOSURES

Not applicable.

ITEM 5. MARKET FOR THE REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS, AND ISSUER PURCHASES OF EQUITY SECURITIES

As of December 31, 2015, there were 128,442,099 shares of common stock of Alaska Air Group, Inc. issued and 125,175,325 shares outstanding and 2,348 shareholders of record. In 2015, we paid quarterly dividends of \$0.200 per share in March, June, September, and December. Our common stock is listed on the New York Stock Exchange (symbol: ALK). The following table shows the trading range of Alaska Air Group, Inc. common stock on the New York Stock Exchange:

		2015			2014			
	High Low			High		Low		
First Quarter	\$	70.83	\$	57.73	\$	46.97	\$	36.28
Second Quarter		68.68		58.15		50.47		43.92
Third Quarter		82.75		62.59		50.10		41.85
Fourth Quarter		87.16		73.00		60.93		40.70

SALES OF NON-REGISTERED SECURITIES

None

PURCHASES OF EQUITY SECURITIES BY THE ISSUER AND AFFILIATED PURCHASERS

	Total Number of Shares Purchased	Average Price Paid per Share	Total Number of Shares (or units) Purchased as Part of Publicly Announced Plans or Programs	Maximum remaining dollar value of shares that can be purchased under the plan (in millions)
October 1, 2015 - October 2, 2015 ^(a)	41,246	\$ 78.83	41,246	
October 3, 2015 – October 31, 2015 ^(b)	531,609	\$ 76.69	531,609	
November 1, 2015 – November 30, 2015 ^(b)	484,454	78.16	484,454	
December 1, 2015 – December 31, 2015 ^(b)	501,214	82.44	501,214	
Total	1,558,523	\$ 79.10	1,558,523	\$ 880

(a) Purchased pursuant to the completed \$650 million repurchase program authorized by the Board of Directors in May 2014.

^(b) Purchased pursuant to the \$1 billion repurchase plan authorized by the Board of Directors in August 2015.

PERFORMANCE GRAPH

The following graph compares our cumulative total stockholder return since December 31, 2010 with the S&P 500 Index and the Dow Jones U.S. Airlines Index. The graph assumes that the value of the investment in our common stock and each index (including reinvestment of dividends) was \$100 on December 31, 2010.



Comparison of 5 year Cumulative Total Return

ITEM 6. SELECTED FINANCIAL AND OPERATING DATA

	 2015	2014	2013		2012	2011
CONSOLIDATED OPERATING RESULTS (audited)						
Year Ended December 31 (in millions, except per-share amounts):						
Operating Revenues ^(a)	\$ 5,598	\$ 5,368	\$ 5,156	\$	4,657	\$ 4,318
Operating Expenses	4,300	4,406	4,318		4,125	3,869
Operating Income	 1,298	 962	 838		532	 449
Nonoperating income (expense), net of interest capitalized ^(b)	14	13	(22)		(18)	(55)
Income before income tax	 1,312	 975	 816		514	 394
Net Income	\$ 848	\$ 605	\$ 508	\$	316	\$ 245
Average basic shares outstanding	 128.373	 135.445	 139.910	_	141.416	 143.510
Average diluted shares outstanding	129.372	136.801	141.878		143.568	146.842
Basic earnings per share	\$ 6.61	\$ 4.47	\$ 3.63	\$	2.23	\$ 1.71
Diluted earnings per share	\$ 6.56	\$ 4.42	\$ 3.58	\$	2.20	\$ 1.66
Cash dividends declared per share	\$ 0.80	0.50	0.20			_
CONSOLIDATED FINANCIAL POSITION (audited)						
At End of Period (in millions):						
Total assets	\$ 6,533	\$ 6,064	\$ 5,725	\$	5,357	\$ 5,061
Long-term debt, including current portion	\$ 686	\$ 803	\$ 871	\$	1,032	\$ 1,307
Shareholders' equity	\$ 2,411	\$ 2,127	\$ 2,029	\$	1,421	\$ 1,174
OPERATING STATISTICS (unaudited)						
Consolidated: ^(c)						
Revenue passengers (000)	31,883	29,278	27,414		25,896	24,790
Revenue passenger miles (RPM) (000,000) "traffic"	33,578	30,718	28,833		27,007	25,032
Available seat miles (ASM) (000,000) "capacity"	39,914	36,078	33,672		31,428	29,627
Load factor	84.1%	85.1%	85.6%		85.9%	84.5%
Yield	14.27¢	14.91¢	14.80¢		14.92¢	14.81¢
Passenger revenues per ASM (PRASM)	12.01¢	12.69¢	12.67¢		12.82¢	12.51¢
Operating revenues per ASM (RASM) ^(d)	14.03¢	14.88¢	14.74¢		14.82¢	14.57¢
Operating expenses per ASM, excluding fuel and noted items (CASMex) ^(d)	8.30¢	8.36¢	8.47¢		8.48¢	8.55¢
Mainline:						
Revenue passengers (000)	22,869	20,972	19,737		18,526	17,810
RPMs (000,000) "traffic"	30,340	27,778	26,172		24,417	22,586
ASMs (000,000) "capacity"	35,912	32,430	30,411		28,180	26,517
Load factor	84.5%	85.7%	86.1%		86.6%	85.2%
Yield	12.98¢	13.58¢	13.33¢		13.45¢	13.26¢
PRASM	10.97¢	11.64¢	11.48¢		11.65¢	11.29¢
CASMex ^(d)	7.39¢	7.45¢	7.54¢		7.56¢	7.60¢
Regional:						
Revenue passengers (000)	9,015	8,306	7,677		7,371	6,980
RPMs (000,000) "traffic"	3,238	2,940	2,661		2,590	2,446
ASMs (000,000) "capacity"	4,002	3,648	3,261		3,247	3,110
Load factor	80.9%	80.6%	81.6%		79.8%	78.6%
Yield	26.37¢	27.40¢	29.20¢		28.81¢	29.13¢
PRASM	21.34¢	22.08¢	23.83¢		22.98¢	22.94¢

(a) In the third quarter of 2013, the Company adopted Accounting Standards Update 2009-13, "Multiple-Deliverable Revenue Arrangements - a consensus of the FASB Emerging Issues Task Force" (ASU 2009-13).

(b) Capitalized interest was \$34 million, \$20 million, \$18 million, and \$12 million for 2015, 2014, 2013, 2012, and 2011, respectively.

(c) Includes flights under Capacity Purchase Agreements operated by SkyWest and PenAir.

(d) See reconciliation of RASM and CASMex to the most directly related GAAP measure in the "Results of Operations" section.

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following Management's Discussion and Analysis of Financial Condition and Results of Operations (MD&A) is intended to help the reader understand the Company, our operations and our present business environment. MD&A is provided as a supplement to – and should be read in conjunction with – our consolidated financial statements and the accompanying notes. All statements in the following discussion that are not statements of historical information or descriptions of current accounting policy are forward-looking statements. Please consider our forward-looking statements in light of the risks referred to in this report's introductory cautionary note and the risks mentioned in Part I, "Item 1A. Risk Factors." This overview summarizes the MD&A, which includes the following sections:

- Year in Review—highlights from 2015 outlining some of the major events that happened during the year and how they affected our financial performance.
- *Results of Operations*—an in-depth analysis of our revenues by segment and our expenses from a consolidated perspective for the three years presented in our consolidated financial statements. To the extent material to the understanding of segment profitability, we more fully describe the segment expenses per financial statement line item. Financial and statistical data is also included here. This section includes forward-looking statements regarding our view of 2016.
- Liquidity and Capital Resources—an overview of our financial position, analysis of cash flows, sources and uses of cash, contractual obligations, and commitments and off-balance sheet arrangements.
- Critical Accounting Estimates—a discussion of our accounting estimates that involve significant judgment and uncertainties.

YEAR IN REVIEW

The Company had a record year in nearly every respect. We posted our 12th consecutive annual profit on an adjusted basis, which is a testament to the hard work of our 15,000 employees and the successful execution of our strategic initiatives. Our 2015 consolidated pretax income was a record \$1.3 billion, a significant increase over the \$975 million in 2014. The \$337 million improvement was driven by an increase of \$230 million in revenues and a substantial decrease of \$464 million in our fuel expense. Partially offsetting these benefits was an increase in operating expenses, excluding fuel and special items, of \$296 million, or 10%, to support the increased capacity of 11%. We recorded \$32 million of special charges in the current year related to a non-cash pension expense associated with the buyout of the obligation for certain terminated, vested plan participants, and a litigation-related matter.

The growth in revenues of \$230 million was driven by the growth in our business. We launched 20 new markets in 2015, including 10 new cities such as Charleston, Nashville, Raleigh-Durham, Milwaukee as well as San Jose and Liberia in Costa Rica. We now have over 1,000 departures daily across our network on peak days - a significant milestone for our airlines. We are providing more utility than ever before in our primary hub of Seattle, giving our customers substantially more non-stop destinations out of Seattle than any other carrier. The significant decline in fuel expense this year was driven by declines in fuel prices, while the increase in non-fuel operating expenses are primarily growth-related increases as we increased our capacity 11%.

See "Results of Operations" below for further discussion of changes in revenues and operating expenses and our reconciliation of Non-GAAP measures to the most directly comparable GAAP measure.

Accomplishments and Highlights

Financial highlights from 2015 include:

Reported record full-year net income, excluding special items, of \$842 million, a 47% increase over 2014. Adjusted diluted earnings per share of \$6.51 was a 56.0% increase compared to 2014. See reconciliation of these non-GAAP measures to comparable GAAP figures in Results of Operations.



- Announced a 38% increase in the quarterly dividend, from \$0.20 per share to \$0.275 per share in January 2016. This is the third time the company has raised the dividend since initiating the quarterly dividend in July 2013, with a cumulative increase of 175% since that time.
- Paid \$0.20 per-share quarterly cash dividend in the fourth quarter, bringing total dividend payments in 2015 to \$102 million.
- Repurchased 7,208,328 shares of common stock for an average price of \$70 during 2015 for \$505 million, or approximately 6% of market capitalization at the beginning of 2015. Since 2007, Air Group has used \$1.3 billion to repurchase 56 million shares at an average price of \$23.66, representing about 35% of the Company's outstanding shares on December 31, 2006.
- Generated nearly \$1.6 billion of operating cash flow and \$753 million of free cash flow in 2015. Since the beginning of 2010 Air Group has generated \$5.6 billion of operating cash flow, and \$2.6 billion of free cash flow.
- Grew passenger revenues by 6% compared to the fourth quarter of 2014, and by 5% compared to full-year 2014.
- Reached a new long term agreement with Bank of America for the Alaska Airlines Visa credit card. The new agreement adds customer benefits, such as no foreign transaction fees, and is expected to generate an incremental \$60 million of revenue in 2016.
- Generated record full-year adjusted pretax margin of 24.0% in 2015, compared to 17.2% in 2014.
- Achieved return on invested capital of 25.2% in 2015, compared to 18.6% in 2014.
- Lowered adjusted debt-to-total capitalization ratio to 27% as of December 31, 2015. Air Group currently has no net debt.
- Lowered consolidated unit costs excluding fuel and special items for the sixth consecutive year, to the lowest level ever. Mainline unit costs excluding fuel have declined 13 of the last 14 years.
- Held \$1.3 billion in unrestricted cash and marketable securities as of December 31, 2015.

2015 Accomplishments and Highlights:

Recognition and Awards

- Ranked the best airline in the U.S. by The Wall Street Journal's "Middle Seat" scorecard for three consecutive years.
- Ranked "Highest in Customer Satisfaction Among Traditional Network Carriers" by J.D. Power and Associates for the eighth year in a row.
- Ranked highest by frequent fliers in the J.D. Power Airline Loyalty/Rewards Program Satisfaction Report for the second year in a row.
- Rated the #1 Airline Rewards Program by U.S. News and World Report.
- Named the "Airline Market Leader" by Air Transport World, becoming the only U.S. airline honored by the magazine in its 2016 Industry Achievement Awards.
- Named No. 1 on-time carrier in North America for the sixth year in a row by FlightStats in January 2016.
- Named Top Performing Airline among mid-sized carriers worldwide by Aviation Week magazine.
- Ranked as the most fuel efficient airline for U.S. airlines by the International Council on Clean Transportation for the fifth year in a row.
- Awarded Fast Travel Platinum status from the International Air Transport Association, which is awarded to airlines offering four or more Fast Travel
 options to at least 80% of their passengers.
- Ranked as a top 100 of America's Best Employers by Forbes Magazine.

19

• Ranked first in the Leadership 500 Excellence Awards, recognizing the success of Alaska's Gear Up leadership training.

Our People

- Awarded a record \$120 million in incentive pay to employees for 2015, or more than one month's pay for most employees. Over the last five years, employees have earned more than \$500 million in incentive pay, averaging 8.6% of annual pay.
- Reached new long-term agreements with Horizon's pilots and flight attendants on contracts that will position Horizon for future growth. These
 contracts were ratified in January 2016 subsequent to year end. The flight attendant agreement becomes amendable in 2019 and the pilot agreement
 becomes amendable in 2024. Each contract includes a signing bonus upon ratification, which is expected to be approximately \$3.5 million, in
 aggregate, in the first quarter of 2016.
- Signed a four-year agreement with Alaska Airline's dispatchers in December 2015.
- Completed "Gear Up 2" for over 1,200 leaders at Alaska and Horizon a continuation of our award-winning leadership training workshop.
- Delivered our "Beyond Service" customer service training to nearly 9,000 customer-facing employees.
- Received a perfect score of 100% for workplace equality on the 2016 Corporate Equality Index (CEI).

Our Customers and Product

- Launched Preferred Plus Seating, providing customers the ability to select bulkhead and exit-row seating 24 hours in advance of the flight. Preferred Plus Seating also includes priority boarding and complimentary beer, wine or cocktail.
- Announced plans to introduce Premium Class seating in 2016, which will provide customers greater leg room, early boarding, and premium onboard amenities, among other things.
- Became the launch customer of Boeing's new, innovative, high-capacity 737 Space Bins, which will increase bag capacity in the cabin by 50%.
- Added a free first checked bag as a permanent feature of the Alaska Airlines Visa Signature affinity credit card.
- Added 11 Boeing 737-900ERs and one Bombardier Q400 aircraft to the operating fleet in 2015.
- Added five Embraer 175 (E175) regional jets in 2015, and committed for future positions to grow the number of E175s to 23 by the end of 2017, including E175s that will replace the eight CRJ700 regional jets operating in our regional network. Furthermore, we may order an additional 30 regional jets with deliveries starting in 2017 that will likely be operated by Horizon.
- Added 20 new markets and 10 new cities to our growing network in 2015.
- Increased fuel efficiency (as measured by seat-miles per gallon) by 2.2% over 2014.

Our Communities

- Donated nearly \$12 million to support local communities, including job training for workers at the Seattle-Tacoma airport, STEM-focused education programs at Seattle's Museum of Flight, the Alaska Native Science and Engineering Program, and Seattle's bicycle sharing program.
- Announced a 10-year sponsorship agreement with the University of Washington which includes, among other things, exclusive naming rights for Alaska Airlines Field at Husky Stadium and Alaska Airlines Arena.

Capital Allocation

In 2015, we repurchased 7,208,328 shares of our common stock for \$505 million under the share repurchase programs authorized by our Board of Directors. Since 2007, we have repurchased 56 million shares of common stock under such programs for \$1.3 billion for an average price of approximately \$23.66 per share. In 2015, we increased our quarterly dividend 60% from \$0.125 per share to \$0.20 per share, and subsequent to December 31, 2015, we announced a 38% increase to \$0.275 per share. Overall, we returned \$607 million to shareholders during 2015 and expect to exceed that amount in 2016.

Outlook

Our primary focus every year is to run a safe, compliant and reliable operation at our airlines. In addition to our primary objective, we remain focused on providing a hassle-free experience for our customers, and building a compelling brand to support growing our network. We recently enhanced our affinity credit card product offering, and in 2016 we are introducing a new premium class of service on our airplanes. Our refreshed brand and bold new brand expressions will roll out throughout our major airports and on our airplanes in 2016 as well.

Similar to the past several years, we expect to continue our growth plans over the next year. Currently, we expect to grow our system-wide capacity by approximately 8% in 2016, compared to 10.6% in 2015. Over the past few years, we have seen competitive capacity increase significantly in our markets, especially in our hometown of Seattle. We expect to see even more competitive capacity in 2016. Current schedules indicate competitive capacity will be 13% higher in the first quarter and 14% higher in the second quarter of 2016. We believe that our product, our operation, our engaged employees and award-winning service, combined with our strong balance sheet give us the ability to compete vigorously in our markets. Because of our strong financial position, low costs, and high number of unencumbered aircraft, we have an ability to flex our fleet to meet demands and allocate capacity in the markets that meet our return objectives.

With our growth plans and the expectation for lower non-fuel unit costs and steady fuel costs for 2016, we believe our financial performance will continue to be strong.

RESULTS OF OPERATIONS

ADJUSTED (NON-GAAP) RESULTS AND PER-SHARE AMOUNTS

We believe disclosure of earnings excluding the impact of mark-to-market gains or losses or other individual revenues or expenses is useful information to investors because:

- By eliminating fuel expense and certain special items from our unit metrics, we believe that we have better visibility into the results of operations and our non-fuel cost-reduction initiatives. Our industry is highly competitive and is characterized by high fixed costs, so even a small reduction in non-fuel operating costs can result in a significant improvement in operating results. In addition, we believe that all domestic carriers are similarly impacted by changes in jet fuel costs over the long run, so it is important for management (and thus investors) to understand the impact of (and trends in) company-specific cost drivers such as labor rates and productivity, airport costs, maintenance costs, etc., which are more controllable by management.
- 2013 Operating revenue per ASM (RASM) excludes a favorable, one-time, non-cash Special mileage plan revenue item of \$192 million primarily
 related to our modified affinity card agreement with Bank of America, executed in July 2013. In accordance with accounting standards, we recorded
 this item in the third quarter of 2013, and it reflects a non-cash adjustment of the value of miles outstanding in the program. We believe it is
 appropriate to exclude this special revenue item from recurring revenues from operations.
- Cost per ASM (CASM) excluding fuel and certain special items is one of the most important measures used by management and by the Air Group Board of Directors in assessing quarterly and annual cost performance.
- Adjusted income before income tax and CASM excluding fuel (and other items as specified in our plan documents) are important metrics for the employee incentive plan that covers all Air Group employees.
- CASM excluding fuel and certain special items is a measure commonly used by industry analysts, and we believe it is the basis by which they compare our airlines to others in the industry. The measure is also the subject of frequent questions from investors.
- Disclosure of the individual impact of certain noted items provides investors the ability to measure and monitor performance both with and without these special items. We believe that disclosing the impact of certain items, such as mark-to-market hedging adjustments, is important because it provides information on significant items that are not necessarily indicative of future performance. Industry analysts and investors consistently measure our performance without these items for better comparability between periods and among other airlines.
- Although we disclose our passenger unit revenues, we do not (nor are we able to) evaluate unit revenues excluding the impact that changes in fuel costs have had on ticket prices. Fuel expense represents a large percentage of our total operating expenses. Fluctuations in fuel prices often drive changes in unit revenues in the mid-to-long term. Although we believe it is useful to evaluate non-fuel unit costs for the reasons noted above, we would caution readers of these financial statements not to place undue reliance on unit costs excluding fuel as a measure or predictor of future profitability because of the significant impact of fuel costs on our business.

Although we are presenting these non-GAAP amounts for the reasons above, investors and other readers should not necessarily conclude that these amounts are non-recurring, infrequent, or unusual in nature.

2015 COMPARED WITH 2014

Our consolidated net income for 2015 was \$848 million, or \$6.56 per diluted share, compared to net income of \$605 million, or \$4.42 per diluted share, in

2014.

Excluding the impact of mark-to-market fuel hedge adjustments, and special items, our adjusted consolidated net income for 2015 was \$842 million, or \$6.51 per diluted share, compared to an adjusted consolidated net income of \$571 million, or \$4.18 per share, in 2014.

	Twelve Months Ended December 31,							
		2015				20		
(in millions, except per-share amounts)	Ē	ollars	D	iluted EPS		Dollars		Diluted EPS
Net income and diluted EPS as reported	\$	848	\$	6.56	\$	605	\$	4.42
Mark-to-market fuel hedge adjustments, net of tax		—		_		(15)		(0.11)
Special items, net of tax		20		0.15		(19)		(0.13)
Special income tax benefit		(26)		(0.20)				—
Non-GAAP adjusted income and per-share amounts	\$	842	\$	6.51	\$	571	\$	4.18

Our operating costs per ASM (CASM) are summarized below:

	Twelve Mo	Twelve Months Ended December 31,					
	2015	2014	% Change				
Consolidated:							
Total operating expenses per ASM (CASM)	10.77¢	12.21¢	(11.8)				
Less the following components:							
Aircraft fuel, including hedging gains and losses	2.39	3.93	(39.2)				
Special items	0.08	(0.08)	NM				
CASM, excluding fuel and fleet transition costs	8.30¢	8.36¢	(0.7)				
Mainline:							
Total mainline operating expenses per ASM (CASM)	9.77¢	11.15¢	(12.4)				
Less the following components:							
Aircraft fuel, including hedging gains and losses	2.29	3.79	(39.6)				
Special items	0.09	(0.09)	NM				
CASM, excluding fuel	7.39¢	7.45¢	(0.8)				
NM - Not meaningful							

22

OPERATING STATISTICS SUMMARY (unaudited) Alaska Air Group, Inc.

Below are operating statistics we use to measure performance. We often refer to unit revenues and adjusted unit costs, which is a non-GAAP measure.

	Twelve Months Ended December 31,								
	2015	2014	Change	2013	Change				
Consolidated Operating Statistics: ^(a)									
Revenue passengers (000)	31,883	29,278	8.9%	27,414	6.8%				
RPMs (000,000) "traffic"	33,578	30,718	9.3%	28,833	6.5%				
ASMs (000,000) "capacity"	39,914	36,078	10.6%	33,672	7.1%				
Load factor	84.1%	85.1%	(1.0) pts	85.6%	(0.5) pts				
Yield	14.27¢	14.91¢	(4.3)%	14.80¢	0.7%				
PRASM	12.01¢	12.69¢	(5.4)%	12.67¢	0.2%				
RASM ^(b)	14.03¢	14.88¢	(5.7)%	14.74¢	0.9%				
CASM excluding fuel and fleet transition costs ^(b)	8.30¢	8.36¢	(0.7)%	8.47¢	(1.3)%				
Economic fuel cost per gallon ^(b)	\$1.88	\$3.08	(39.0)%	\$3.30	(6.7)%				
Fuel gallons (000,000)	508	469	8.3%	447	4.9%				
ASM's per gallon	78.6	76.9	2.2%	75.3	2.1%				
Average number of full-time equivalent employees (FTEs)	13,858	12,739	8.8%	12,163	4.7%				
Mainline Operating Statistics:									
Revenue passengers (000)	22,869	20,972	9.0%	19,737	6.3%				
RPMs (000,000) "traffic"	30,340	27,778	9.2%	26,172	6.1%				
ASMs (000,000) "capacity"	35,912	32,430	10.7%	30,411	6.6%				
Load factor	84.5%	85.7%	(1.2) pts	86.1%	(0.4) pts				
Yield	12.98¢	13.58¢	(4.4)%	13.33¢	1.9%				
PRASM	10.97¢	11.64¢	(5.8)%	11.48¢	1.4%				
CASM excluding fuel ^(b)	7.39¢	7.45¢	(0.8)%	7.54¢	(1.2)%				
Economic fuel cost per gallon ^(b)	\$1.87	\$3.07	(39.1)%	\$3.30	(7.0)%				
Fuel gallons (000,000)	439	407	7.9%	393	3.6%				
ASM's per gallon	81.8	79.7	2.6%	77.4	3.0%				
Average number of FTE's	10,750	9,910	8.5%	9,493	4.4%				
Aircraft utilization	10.8	10.5	2.9%	10.6	(0.9)%				
Average aircraft stage length	1,195	1,182	1.1%	1,177	0.4%				
Mainline operating fleet at period-end	147 a/c	137 a/c	10 a/c	131 a/c	6 a/c				
Regional Operating Statistics: ^(c)									
Revenue passengers (000)	9,015	8,306	8.5%	7,677	8.2%				
RPMs (000,000) "traffic"	3,238	2,940	10.1%	2,661	10.5%				
ASMs (000,000) "capacity"	4,002	3,648	9.7%	3,261	11.9%				
Load factor	80.9%	80.6%	0.3 pts	81.6%	(1.0) pts				
Yield	26.37¢	27.40¢	(3.8)%	29.20¢	(6.2)%				
PRASM	21.34¢	22.08¢	(3.4)%	23.83¢	(7.3)%				

(a) Except for FTEs, data includes information related to regional CPA flying with Horizon, SkyWest and PenAir.

(b) See reconciliation of this measure to the most directly related GAAP measure in the "Results of Operations" section.

(c) Data presented includes information related to regional CPAs.

OPERATING REVENUES

Total operating revenues increased \$230 million, or 4%, during 2015 compared to the same period in 2014. The changes are summarized in the following table:

	1ber 31,			
	2015	2014		% Change
\$	3,939	\$	3,774	4
	854		805	6
\$	4,793	\$	4,579	5
	108		114	(5)
	697		675	3
\$	5,598	\$	5,368	4
	\$ \$ \$ \$	2015 \$ 3,939 854 \$ 4,793 108 697	2015 \$ 3,939 \$ 854 \$ \$ 4,793 \$ 108 697	\$ 3,939 \$ 3,774 854 805 \$ 4,793 \$ 4,579 108 114 697 675

Passenger Revenue – Mainline

Mainline passenger revenue for 2015 increased by 4% on a 10.7% increase in capacity, partially offset by a 5.8% decrease in PRASM compared to 2014. The increase in capacity was driven by new routes, larger aircraft added to our fleet, and increased utilization of our aircraft. The decrease in PRASM was driven by a 4.4% decrease in ticket yield, combined with a 1.2 point decrease in load factor compared to the prior year. The decline in ticket yield was primarily due to increased competitive capacity in the markets we serve, and our own growth. Furthermore, the significant decline in fuel prices has had an impact on lower ticket prices. The decline in load factor was also a result of increased capacity.

We expect competitive pressures on unit revenues to continue into 2016. However, we expect total passenger revenue will increase from 2015, although likely less than our growth rate in capacity.

Passenger Revenue – Regional

Regional passenger revenue increased by \$49 million, or 6%, compared to 2014 due to a 9.7% increase in capacity, partially offset by a 3.4% decrease in PRASM compared to 2014. The increase in capacity is due to an increase in departures, and average aircraft stage length. The decrease in PRASM was due to a 3.8% decrease in ticket yield, partially offset by an increase in load factor of 0.3 points. The decline in yield was due to an increase in competitive capacity in our regional markets and our own growth as we strengthen our network utility in the Pacific Northwest.

We expect Regional passenger revenue to increase in 2016, primarily due to the annualization of new routes introduced in 2015, and additional regional routes scheduled to be added in 2016.

Other – Net

Other—net revenue increased \$22 million, or 3%, from 2014, due to increases in Mileage Plan revenue and food and beverage sales, partially offset by lower bag fee revenues. Mileage Plan revenue increased \$34 million or 12%, due to increased miles sold. Food and beverage sales were higher due to the 8.9% increase in passengers and selling more premium offerings, such as Tom Douglas signature meals. Bag fee revenue was lower due to promotions launched in January to offer a free first checked bag to our Mileage Plan members, and to all Alaska Airlines Signature Visa credit card holders beginning in February. This decline was partially offset by incremental revenue from our affinity card bank partner.

We expect our Other—net revenue to experience an increase at a pace higher than the expected increase in passengers in 2016, due primarily to the extended agreement with our affinity credit card bank partner effective January 1, 2016. We expect this agreement will result in an incremental \$60 million of revenue in 2016.



OPERATING EXPENSES

Total operating expenses decreased \$106 million, or 2%, compared to 2014, primarily as a result of lower fuel costs. We believe it is useful to summarize operating expenses as follows, which is consistent with the way expenses are reported internally and evaluated by management:

	Twelve Months Ended December 31,						
(in millions)	2015		2014	% Change			
Fuel expense	\$ 954	\$	1,418	(33)			
Non-fuel expenses	3,314		3,018	10			
Special items	32		(30)	NM			
Total Operating Expenses	\$ 4,300	\$	4,406	(2)			

NM - Not Meaningful

Significant operating expense variances from 2014 are more fully described below.

Wages and Benefits

Wages and benefits increased during 2015 by \$118 million, or 10%, compared to 2014. The primary components of wages and benefits are shown in the following table:

	 Twelve Months Ended December 31,						
(in millions)	2015	2014		% Change			
Wages	\$ 945	\$	862	10			
Medical and other benefits	153		150	2			
Defined contribution plans	60		53	13			
Pension - Defined benefit plans	28		9	211			
Payroll taxes	68		62	10			
Total wages and benefits	\$ 1,254	\$	1,136	10			

Wages increased 10%, primarily due to an 8.8% increase in FTEs and the annualization of new labor contracts that included higher rates. The increase in FTEs was to support the growth in our business.

Defined contribution plans increased 13% due to increased contributions throughout all labor groups and an increased matched percentage as a part of recent labor contracts.

Pension expense increased \$19 million, compared to the same period in the prior year. The increase is due to higher amortization of actuarial losses from previous years due primarily to a lower discount rate used to value the pension obligation at December 31, 2014.

We expect wages and benefits to be higher in 2016 compared to 2015 on a 4% to 5% increase in FTEs.

Variable Incentive Pay

Variable incentive pay expense increased to \$120 million in 2015 from \$116 million in 2014. The increase is due to actual results exceeding our targets for financial performance more so than in the prior year, coupled with a higher wage base.

Aircraft Fuel

Aircraft fuel expense includes both *raw fuel expense* (as defined below) plus the effect of mark-to-market adjustments to our fuel hedge portfolio included in our consolidated statement of operations as the value of that portfolio increases and decreases. Aircraft fuel expense can be volatile, even between quarters, because it includes these gains or losses in the value of the underlying instrument as crude oil prices and refining margins increase or decrease. *Raw fuel expense* is defined as the price that we generally pay at the airport, or the "into-plane" price, including taxes and fees. Raw fuel prices are impacted by world oil prices and refining costs, which can vary by region in the U.S. *Raw fuel expense* approximates cash paid to suppliers and does not reflect the effect of our fuel hedges.

Aircraft fuel expense decreased \$464 million, or 33% compared to 2014. The elements of the change are illustrated in the following table:

	Twelve Months Ended December 31,									
	2015				2014					
(in millions, except for per gallon amounts)	Dollars		Cost/Gal		Dollars		Cost/Gal			
Raw or "into-plane" fuel cost	\$	935	\$	1.84	\$	1,400	\$	2.99		
Losses on settled hedges		19		0.04		41		0.09		
Consolidated economic fuel expense	\$	954	\$	1.88	\$	1,441	\$	3.08		
Mark-to-market fuel hedge adjustments		_		_		(23)		(0.05)		
GAAP fuel expense	\$	954	\$	1.88	\$	1,418	\$	3.03		
Fuel gallons		508				469				

Fuel gallons consumed increased 8.3% in line with the 10.6% increase in capacity, partially offset by a 2.2% improvement in fuel efficiency as measured by ASMs per gallon.

The raw fuel price per gallon decreased 38.5% as a result of lower West Coast jet fuel prices. West Coast jet fuel prices are impacted by both the price of crude oil, as well as refining margins associated with the conversion of crude oil to jet fuel. The decrease in raw fuel price per gallon during 2015 was due to a decline in crude oil prices of 48% and a decrease in refining margins of 11%, when compared to the prior year.

We also evaluate economic fuel expense, which we define as raw fuel expense adjusted for the cash we receive from, or pay to, hedge counterparties for hedges that settle during the period, and for the premium expense that we paid for those contracts. A key difference between aircraft fuel expense and economic fuel expense is the timing of gain or loss recognition on our hedge portfolio. When we refer to economic fuel expense, we include gains and losses only when they are realized for those contracts that were settled during the period based on their original contract terms. We believe this is the best measure of the effect that fuel prices are currently having on our business because it most closely approximates the net cash outflow associated with purchasing fuel for our operations. Accordingly, many industry analysts evaluate our results using this measure, and it is the basis for most internal management reporting and incentive pay plans.

Losses recognized for hedges that settled during the year were \$19 million in 2015, compared to losses of \$41 million in 2014. These amounts represent cash paid for premium expense, offset by any cash received from those hedges at settlement. The decrease in losses on settled hedges is primarily due to our increased use of "out of the money" call options as well as purchasing shorter-dated options, both of which reduce the premium cost we pay.

We currently expect our economic fuel price per gallon to be approximately 37% lower in the first quarter of 2016 than the first quarter of 2015 due to lower West Coast jet fuel prices and the decrease in premium costs related to our fuel hedge program. As both oil prices and refining margins are volatile, we are unable to forecast the full-year cost with any certainty.

Aircraft Maintenance

Aircraft maintenance costs increased by \$24 million, or 10%, compared to the prior year. Maintenance costs increased due to more scheduled engine maintenance events that were more expensive due to replacing life-limited parts, and heavier airframe checks.



We expect aircraft maintenance to be 5% to 8% higher in 2016 primarily due to the timing of maintenance events as our fleet ages and from lease return costs as we return 13 B737-400 aircraft to lessors.

Landing Fees and Other Rentals

Landing fees and other rental expenses increased \$17 million, or 6%, primarily due to increased flying in 2015, as we increased capacity and entered new markets.

We expect landing fees and other rental expenses to increase in 2016 in line with the expected capacity growth and higher rates at the airports where we operate.

Contracted Services

Contracted services increased \$18 million, or 9%, when compared to 2014. The increase is primarily due to increased flying at stations where we use vendors to assist us with passenger and ramp handling.

We expect contracted services to be higher in 2016 due to higher rates and volumes as we continue to expand into new markets utilizing vendor services.

Selling Expenses

Selling expenses increased by \$12 million, or 6%, compared to 2014, mostly due to increased promotional and advertising activities in Seattle and increased credit card commissions due to higher revenue.

We expect selling expense to increase approximately 10% in 2016, as we refresh our brand and increase promotion and advertising in our core markets, and as revenues increase.

Depreciation and Amortization

Depreciation and amortization expenses increased by \$26 million, or 9%, due to the increased number of aircraft in our fleet, and cabin upgrades made in the past 18 months.

We expect depreciation and amortization expense to increase approximately 15% in 2016 as we continue to purchase aircraft for replacement of leased aircraft and for growth.

Food and Beverage Service

Food and beverage service expenses increased by \$20 million, or 22%, due to the increased number of passengers, and upgrades to our onboard menu, offering higher quality food and beverage products.

We expect food and beverage expenses to increase approximately 5% in 2016 primarily because of the expected increase in passengers.

Third-party regional carrier expense

Third-party regional carrier expense, which represents payments made to SkyWest and PenAir under our CPA agreements, increased \$14 million, or 24%, in 2015 compared to 2014. The increase is primarily due to the five E175 aircraft operated by SkyWest added to our regional operation in 2015.

We expect third-party regional carrier expense to increase in 2016 as we continue to expand our regional network.

Other Operating Expenses

Other operating expenses increased \$48 million, or 16%, compared to 2014. The increase is primarily due to professional services with regard to our brand refresh and network strategy, personnel costs for our flight crews, and IT-related costs.

We expect other operating expenses to increase 3% to 5% in 2016, primarily as a result of IT-related investments.

27

Special Items

In the fourth quarter of 2015, we recorded special items of \$32 million (\$20 million after tax). This is due to a \$14 million million non-cash pension settlement expense related to lump-sum payments made to terminated, vested participants which removed them from participation in the pension plan, and an \$18 million expense related to ongoing litigation.

Consolidated Nonoperating Income (Expense)

During 2015, we recorded nonoperating income of \$14 million, compared to \$13 million in 2014. In the prior year, we recognized gains on the sale of certain equity securities. In the current year, we have capitalized more of our interest expense on an increasing balance of prepaid aircraft deposits.

Operating Costs per Available Seat Mile

We are presenting our line-item expenses below both in absolute dollars and on an ASM basis to highlight areas in which costs have increased or decreased either more or less than capacity.

			Twelve	Months Ended Decen	mber 31,	
	 2015		2014	2015	2014	%Change
(in millions, except CASM)	 Amount		Amount	CASM	CASM	CASM
Wages and benefits	\$ 1,254	\$	1,136	3.14¢	3.16¢	(0.6)%
Variable incentive pay	120		116	0.30	0.32	(6.3)%
Aircraft maintenance	253		229	0.64	0.63	1.6 %
Aircraft rent	105		110	0.26	0.30	(13.3)%
Landing fees and other rentals	296		279	0.74	0.77	(3.9)%
Contracted services	214		196	0.54	0.54	<u> </u>
Selling expenses	211		199	0.53	0.55	(3.6)%
Depreciation and amortization	320		294	0.80	0.81	(1.2)%
Food and beverage service	113		93	0.28	0.26	7.7 %
Other	356		308	0.89	0.86	3.5 %
Third-party regional carrier expense	72		58	0.18	0.16	12.5 %
Non-fuel Expenses ^(a)	\$ 3,314	\$	3,018	8.30¢	8.36¢	(0.7)%

^(a) Excludes special items recorded in 2015 and 2014.

Additional Segment Information

Refer to the Notes of the Consolidated Financial Statements for a detailed description of each segment. Below is a summary of each segments' profitability.

Alaska Mainline

Pretax profit for Alaska Mainline was \$1.2 billion in 2015 compared to \$834 million in 2014. The \$165 million increase in Mainline passenger revenue is described previously. Mainline operating expense excluding fuel increased by \$236 million, due to higher wages to support our growth, higher ramp and passenger handling associated with increased flying, higher depreciation related to our fleet growth, and increased food and beverage costs. Economic fuel cost, as defined above, decreased due to lower raw fuel costs, and increased fuel efficiency, slightly offset by an increase in consumption.

Alaska Regional

Pretax profit for Alaska Regional was \$105 million in 2015 compared to \$74 million in 2014. The \$49 million increase in Alaska Regional passenger revenue is described above. The increase in revenue and the significant decline in fuel costs were partially offset by higher expenses to support additional departures.

Horizon

Pretax profit for Horizon was \$28 million in 2015 compared to \$17 million in 2014. CPA Revenues (100% of which are from Alaska and eliminated in consolidation) increased due to additional capacity and higher rates. The \$26 million increase in Horizon's non-fuel operating expenses was largely driven by increased engine maintenance and other expenses to support the increase in capacity.

2014 COMPARED WITH 2013

Our consolidated net income for 2014 was \$605 million, or \$4.42 per diluted share, compared to net income of \$508 million, or \$3.58 per diluted share, in 2013. Significant items impacting the comparability between the periods are as follows:

- Both periods include adjustments to reflect the timing of net unrealized mark-to-market gains or losses related to our fuel hedge positions. For 2014, we recognized net mark-to-market losses of \$23 million (\$15 million after tax, or \$0.11 per diluted share) compared to gains of \$8 million (\$5 million after tax, or \$0.03 per share) in 2013.
- In 2014, we recognized a one-time, non-cash benefit from the curtailment of certain post-retirement benefit plans of \$20 million and a one-time gain associated with the settlement of a legal matter of \$10 million. The aggregate \$30 million (\$19.0 million in aggregate after tax, or \$0.13 per diluted share) is included in Special items in the consolidated statement of operations.
- In 2013, we recognized a one-time, non-cash Special mileage plan revenue item of \$192 million (\$120 million after tax, or \$0.85 per diluted share) that resulted from the application of new accounting rules associated with the modified Bank of America Affinity Card Agreement, and the effect of an increase in the estimate of the number of frequent flier miles expected to expire unused.

Excluding the mark-to-market fuel hedge adjustments, special items, and the one-time Special mileage plan revenue item, our adjusted consolidated net income for 2014 was \$571 million, or \$4.18 per diluted share, compared to an adjusted consolidated net income of \$383 million, or \$2.70 per share, in 2013.

	Twe										
	2014					2013					
(in millions, except per-share amounts)	Dollars Diluted EPS				Dollars			Diluted EPS			
Net income and diluted EPS as reported	\$	605	\$	4.42	\$	508	\$	3.58			
Mark-to-market fuel hedge adjustments, net of tax		(15)		(0.11)		(5)		(0.03)			
Special items, net of tax		(19)		(0.13)		—		—			
Special mileage plan revenue, net of tax		—		—		(120)		(0.85)			
Non-GAAP adjusted income and per-share amounts	\$	571	\$	4.18	\$	383	\$	2.70			

Revenues adjusted for the one-time Special mileage plan item is as follows:

		Twelve Months Ended December 31,							
	-		2014		2013	% Change			
Total operating revenues		\$	5,368	\$	5,156	4.1			
Less: Special mileage plan revenue					192	NM			
Adjusted Revenue	1	\$	5,368	\$	4,964	8.1			
Consolidated ASMs			36,078		33,672	7.1			
RASM	-		14.88¢		14.74¢	0.9			
	=			-					

NM - Not meaningful

Our operating costs per ASM are summarized below:

	Twelve Mo	nths Ended Decemb	er 31,
	2014	2013	% Change
Consolidated:			
Total operating expenses per ASM (CASM)	12.21¢	12.82¢	(4.8)
Less the following components:			
Aircraft fuel, including hedging gains and losses	3.93	4.35	(9.7)
Special items	(0.08)	_	NM
CASM, excluding fuel and fleet transition costs	8.36¢	8.47¢	(1.3)
Mainline:			
Total mainline operating expenses per ASM (CASM)	11.15¢	11.77¢	(5.3)
Less the following components:			
Aircraft fuel, including hedging gains and losses	3.79	4.23	(10.4)
Special items	(0.09)	_	NM
CASM, excluding fuel	7.45¢	7.54¢	(1.2)
NM - Not Meaningful			

OPERATING REVENUES

Total operating revenues increased \$212 million, or 4%, during 2014 compared to the same period in 2013. Adjusted for the Special mileage plan revenue item recognized in 2013, operating revenues increased \$404 million, or 8%, during 2014. The changes are summarized in the following table:

		Twelve Months Ended December 31,							
(in millions)		2014			2013	% Change			
Passenger	_								
Mainline	S	\$	3,774	\$	3,490	8			
Regional			805		777	4			
Total passenger revenue	5	\$	4,579	\$	4,267	7			
Freight and mail			114		113	1			
Other - net			675		584	16			
Special mileage plan revenue			_		192	NM			
Total operating revenues	5	\$	5,368	\$	5,156	4			
NM - Not meaningful	=								

Passenger Revenue – Mainline

Mainline passenger revenue for 2014 increased by 8% on a 6.6% increase in capacity and a 1.4% increase in PRASM compared to 2013. The increase in capacity was driven by new routes, seats added to our existing fleet as part of our cabin improvement project, and delivery of 10 737-900ERs in 2014. The increase in PRASM was driven by a 1.9% increase in ticket yield, partially offset by a 0.4-point decrease in load factor compared to the prior year. Increase in yield was due to reallocation of capacity to markets with stronger demand and by a change in revenue allocation between Mainline and Regional service because of certain industry pricing changes. Without the industry change, Mainline yields would have increased by 0.9%.

Passenger Revenue – Regional

Regional passenger revenue increased by \$28 million, or 4%, compared to 2013 on an 11.9% increase in capacity, partially offset by a 7.3% decline in PRASM compared to 2013. The decrease in PRASM was due to a 6.2% decrease in ticket yield coupled with a 1.0-point decrease in load factor compared to the prior year. The decline in yield was primarily driven by a change in revenue allocation between Mainline and Regional service because of certain industry pricing changes. Without the revenue allocation adjustment, yield would have decreased 1.7%. Additionally, the average trip length for our Regional flights increased 3% in 2014, which also put downward pressure on yields.

Other – Net

Other-net revenue increased \$91 million, or 16%, from 2013. This is primarily due to an increase in our Mileage PlanTM revenues of \$39 million, or 15%, due to an increase in miles sold and an increase in cash received per mile. Additionally, bag fees and ticket change fees are up 23% and 12%, respectively, due to changes in our fee structure that took effect in November 2013.

Special Mileage Plan Revenue

In 2013, we modified and extended our co-branded credit card agreement with BAC. In connection with this agreement and as a result of applying related accounting standards, we recorded a one-time, non-cash Special mileage plan revenue item of \$192 million primarily related to our revaluation of the deferred revenue liability related to miles previously sold to Bank of America Corporation (referred to herein as BAC).

OPERATING EXPENSES

Total operating expenses increased \$88 million, or 2%, compared to 2013, primarily driven by higher non-fuel costs due to increased capacity. We believe it is useful to summarize operating expenses as follows, which is consistent with the way expenses are reported internally and evaluated by management:

	Twelve Months Ended December 31							
(in millions)		2014	% Change					
Fuel expense	\$	1,418	\$	1,467	(3)			
Non-fuel expenses		3,018		2,851	6			
Special items		(30)			NM			
Total Operating Expenses	\$	4,406	\$	4,318	2			

Significant operating expense variances from 2013 are more fully described below.

Wages and Benefits

Wages and benefits increased during 2014 by \$50 million, or 5%, compared to 2013. The primary components of wages and benefits are shown in the following table:

		Twelve Months Ended December 31,								
(in millions)	_	2014		2013	% Change					
Wages	\$	862	\$	788	9					
Medical and other benefits		150		145	3					
Defined contribution plans		53		44	20					
Pension - Defined benefit plans		9		50	(82)					
Payroll taxes		62		59	5					
Total wages and benefits	\$	1,136	\$	1,086	5					

Wages increased 9%, primarily due to annualization of new labor contracts that included higher rates, a 4.7% increase in full-time equivalent employees, and an \$8 million signing bonus paid to Alaska's flight attendants in December 2014 in connection with the ratification of a new collective bargaining agreement. the increase in FTEs was to support the growth in our business.

Defined contribution plan expense increased 20% due to an increase in the number of employees participating in the plans and an increase in the employer contribution for non-union employees previously in the pension plan.

Pension expense decreased 82%, compared to the same period in the prior year. The decline is due to having a lower accumulated loss to amortize as a result of higher plan assets, a higher discount rate at December 31, 2013 compared to December 31, 2012, and the freezing of plan benefits for our non-union employees beginning January 1, 2014.

Variable Incentive Pay

Variable incentive pay expense increased from \$105 million in 2013 to \$116 million in 2014. The increase is due to actual results exceeding our target results of financial and operational performance more so than in the prior year.

Aircraft Fuel

Aircraft fuel expense decreased \$49 million, or 3% compared to 2013. The elements of the change are illustrated in the following table:

]	welve Months E	nded	December 31,			
	 20	014		2013				
(in millions, except for per gallon amounts)	 Dollars Cost/Gal				Dollars	Cost/Gal		
Raw or "into-plane" fuel cost	\$ 1,400	\$	2.99	\$	1,423	\$	3.19	
Gains on settled hedges	41		0.09		52		0.11	
Consolidated economic fuel expense	\$ 1,441	\$	3.08	\$	1,475	\$	3.30	
Mark-to-mark fuel hedge adjustments	 (23)		(0.05)		(8)		(0.02)	
GAAP fuel expense	\$ 1,418	\$	3.03	\$	1,467	\$	3.28	
Fuel gallons	 469				447			

Fuel gallons consumed increased 4.9% in line with the increase in departures and capacity, partially offset by a 2.1% improvement in fuel efficiency as measured by ASMs per gallon.

The raw fuel price per gallon decreased 6.3% as a result of lower West Coast jet fuel prices. The decrease in raw fuel price per gallon during 2014 was due to a decrease in average crude oil prices of 5% and a decline in refining margins of 16%, as compared to the prior year.

Losses recognized for hedges that settled during the year were \$41 million in 2014, compared to losses of \$52 million in 2013. These amounts represent the cash received, or paid, net of the premium expense recognized for those hedges.

Aircraft Maintenance

Aircraft maintenance decreased by \$18 million, or 7%, compared to the prior year. The decrease is primarily due to a \$22 million reduction in our power-bythe-hour (PBH) expense, \$11 million in lower lease return costs, and five fewer unscheduled engine checks for our Q400 aircraft. Offsetting these decreases was an \$11 million increase in engine maintenance expense primarily related to our 737-400 engines, and slightly higher airframe checks for both our 737 and Q400 fleet.

The decrease in our PBH expense is due to fewer engines covered by the contracts in the current year, along with reduced flying on the engines that are still under the current contract. The decrease is return costs is due to the four aircraft we returned during the current year and two aircraft we returned at the end of the prior year with no lease return costs expected for lease returns in 2015.

Landing Fees and Other Rentals

Landing fees and other rentals increased \$17 million, or 6%, primarily due to increased flying in 2014 as we increased capacity and entered into new markets.

Contracted Services

Contracted services increased \$20 million, or 11%, primarily due to \$15 million increase in contract ramp and passenger handling costs resulting from new stations and rate increases in Seattle.

Selling Expenses

Selling expenses increased by \$20 million, or 11%, compared to 2013, mostly due to increased promotional and advertising activities in Seattle and increased credit card commissions from higher revenue.



Depreciation and Amortization

Depreciation and amortization expenses increased by \$24 million, or 9%, due to increased number of aircraft in our fleet.

Food and Beverage Service

Food and beverage service expenses increased by \$9 million, or 11%, due to the increased number of passengers, and more premium product offerings.

Third-party regional carrier expense

Third-party regional carrier expense, which represents payments made to SkyWest and PenAir under our CPA agreements, increased \$13 million, or 29%, in 2014 compared to 2013. The increase is primarily due to the three additional lines of flying performed by SkyWest in 2014 compared to 2013.

Other Operating Expenses

Other operating expenses increased \$30 million, or 11%, compared to 2013. The increase is primarily driven by IT project costs, higher professional fees, and flight crew hotel costs.

Special Items

In the fourth quarter of 2014, we recorded special items for \$30 million. This is primarily due to a \$20 million non-cash curtailment gain related to certain post-retirement benefits that were reduced in 2014. The remaining gain is related to a one-time cash settlement related to a legal matter.

Consolidated Nonoperating Income (Expense)

During 2014, we recorded nonoperating income of \$13 million, compared to an expense of \$22 million in 2013. The \$35 million favorable change is due to gains recorded in the current year related to the sale of certain equity securities and reduced interest expense due to lower average debt levels. Additionally, in the prior year, we incurred costs of \$12 million to overhaul and repair three aircraft that were previously subleased to another carrier.

Operating Costs per Available Seat Mile

		Twelve Months Ended December 31,											
		2014		2013	2014	2013	%Change						
(in millions, except CASM)	I	Amount		Amount	CASM	CASM	CASM						
Wages and benefits	\$	1,136	\$	1,086	3.16¢	3.23¢	(2.2)%						
Variable incentive pay		116		105	0.32	0.31	3.2 %						
Aircraft maintenance		229		247	0.63	0.73	(13.7)%						
Aircraft rent		110		119	0.30	0.35	(14.3)%						
Landing fees and other rentals		279		262	0.77	0.78	(1.3)%						
Contracted services		196		176	0.54	0.52	3.8 %						
Selling expenses		199		179	0.55	0.53	3.8 %						
Depreciation and amortization		294		270	0.81	0.80	1.3 %						
Food and beverage service		93		84	0.26	0.25	4.0 %						
Other		308		278	0.86	0.83	3.6 %						
Third-party regional carrier expense		58		45	0.16	0.14	14.3 %						
Non-fuel expenses	\$	3,018	\$	2,851	8.36¢	8.47¢	(1.3)%						

Additional Segment Information

Refer to the Notes of the Consolidated Financial Statements for a detailed description of each segment. Below is a summary of each segments' profitability.



Alaska Mainline

Pretax profit for Alaska Mainline was \$834 million in 2014 compared to \$530 million in 2013. The \$284 million increase in Mainline passenger revenue is described previously. Mainline operating expense excluding fuel increased by \$124 million, due to increased capacity, departures, expanding to new locations, and higher advertising and promotional activity in Seattle and our new locations. Additionally, we increased spending on IT infrastructure projects, and incurred more depreciation as we continue to purchase aircraft. Economic fuel cost as defined above decreased due to a decline in the economic price per gallon, and increased fuel efficiency, slightly offset by an increase in consumption.

Alaska Regional

Pretax profit for Alaska Regional was \$74 million in 2014 compared to \$69 million in 2013. The \$28 million increase in Alaska Regional passenger revenue is described previously. The increased Regional revenue was offset by higher expenses to support additional capacity. Additionally, we recorded a \$12 million loss in 2013 related to overhaul and repair of three aircraft that were previously subleased to another carrier.

Horizon

Pretax profit for Horizon was \$17 million in 2014 compared to \$20 million in 2013. CPA Revenues (100% of which are from Alaska and eliminated in consolidation) increased due to additional capacity in the state of Alaska. The \$8 million increase in Horizon's non-fuel operating expenses was driven by increased wages to support additional aircraft in the fleet, higher pilot training costs, and increased depreciation and amortization due to the three additional Q400 aircraft added in Q4 of 2013.

LIQUIDITY AND CAPITAL RESOURCES

Our primary sources of liquidity are:

- Our existing cash and marketable securities balance of \$1.3 billion, and our expected cash from operations;
- Our 89 unencumbered aircraft in the operating fleet as of December 31, 2015, that could be financed, if necessary;
- Our combined \$200 million bank line-of-credit facilities, with none currently outstanding.

In 2015, we took free and clear delivery of 11 B737-900ER aircraft. We made debt payments totaling \$116 million. In addition, we continued to return capital to our shareholders by repurchasing \$505 million of our common stock in 2015, and paid dividends totaling \$102 million. Because of our strong balance sheet and financial performance, we are one of only two airlines in the U.S. with investment grade credit ratings. We will continue to focus on preserving a strong liquidity position and evaluate our cash needs as conditions change.

We believe that our current cash and marketable securities balance combined with future cash flows from operations and other sources of liquidity will be sufficient to fund our operations for the foreseeable future.

In our cash and marketable securities portfolio, we invest only in securities that meet our primary investment strategy of maintaining and securing investment principal. The portfolio is managed by reputable firms that adhere to our investment policy that sets forth investment objectives, approved and prohibited investments, and duration and credit quality guidelines. Our policy and the portfolio managers are continually reviewed to ensure that the investments are aligned with our strategy.

34

The table below presents the major indicators of financial condition and liquidity:

(in millions, except per share and debt-to-capital amounts)	December 31, 2015 December 31, 2014		_	Change	
Cash and marketable securities	\$	1,328	\$ 1,217	\$	111
Cash, marketable securities, and unused lines of credit as a percentage of trailing twelve months revenue		28%	26%		2 pts
Long-term debt, net of current portion		571	686		(115)
Shareholders' equity		2,411	2,127		284
Long-term debt-to-capital ratio ^(a)		27%:73%	31%:69%) (4pts

(a) Calculated using the present value of remaining aircraft lease payments for aircraft that are in our operating fleet as of the balance sheet date.

The following discussion summarizes the primary drivers of the increase in our cash and marketable securities balance and our expectation of future cash requirements.

ANALYSIS OF OUR CASH FLOWS

Cash Provided by Operating Activities

In 2015, we generated \$1.6 billion in operating cash flows compared to \$1.0 billion in 2014. The increase of \$554 million was primarily attributable to improved operating results, driven largely by lower jet fuel costs.

Cash provided by operating activities was \$1 billion in 2014, compared to \$981 million in 2013. The \$49 million increase is primarily due to an increase in earnings, and an increase in our advance ticket sales, partially offset by a \$177 million increase in payments for income taxes.

We typically generate positive cash flows from operations, and expect to use a portion to invest in capital expenditures and increasing shareholder value through the repurchase of our common stock and dividends.

Cash Used in Investing Activities

Cash used in investing activities was \$930 million during 2015, compared to \$541 million in 2014. Our capital expenditures were \$831 million, or \$137 million higher than in 2014. This is due to the delivery of 11 B737-900ERs, one Q400 aircraft, and advance purchase deposits on 32 B737 aircraft that will be delivered over the next 24 months. This compares to the delivery of ten B737-900ERs in the prior year. In 2015, we replaced restricted cash on deposit of \$52 million with a line of credit facility, resulting in an increase in investing cash flows.

As of December 31, 2015, we had firm commitments for 68 B737 aircraft through 2022 with options to acquire up to 46 additional 737 NextGen (NG) aircraft and MAX aircraft in 2018 through 2024. We also have firm commitments for two Q400s and options to acquire five Q400 aircraft with deliveries from 2018 to 2019, although we do not currently expect to take delivery of the firm orders or exercise the options. In addition to the five E175 aircraft currently operating in our regional fleet, we have commitments to lease 18 E175 aircraft with start dates in 2016 through 2017. We also have options to lease eight E175 aircraft with positions starting in late 2017 and early 2018. The options for all fleet types give us the flexibility, but not the obligation, to grow the fleet assuming profitability and return on invested capital targets can be met.

Additionally, we may purchase up to 30 regional jets to be operated by Horizon with deliveries starting in 2017. At this time, no order has been placed nor have any commercial terms been agreed upon.

The table below reflects total expected capital expenditures and the additional expenditures if options were exercised. Additional options will be exercised only if we believe return on invested capital targets can be met:

	2015							
	Actuals	2	2016	2017	1	2018	1	2019
Aircraft and aircraft purchase deposits - firm ^{(a)(b)}	\$ 681	\$	450	\$ 470	\$	400	\$	345
Other flight equipment	79		40	45		40		40
Other property and equipment	71		150	105		85		85
Total property and equipment additions (c)	\$ 831	\$	640	\$ 620	\$	525	\$	470
Option aircraft and aircraft deposits, if exercised	\$ _	\$	65	\$ 130	\$	235	\$	350

(a) Excludes contractual payments for two Q400 aircraft in 2018 for which we do not expect to take delivery.

(b) We may order 30 regional jets for deliveries over the next several years, for which none of the related expenditures are included in the table above. No commercial terms have been reached, however, if the order is placed in 2016, we would expect additional capital expenditures of approximately \$40 million in 2016.

(c) 2015 actual capital expenditures include capitalized interest. Future expected capital expenditures exclude capitalized interest and represent only contractual or expected cash outlay for capital assets.

Cash used in investing activities was \$541 million during 2014, compared to \$698 million in 2013. Our capital expenditures were \$694 million in 2014, or \$128 million higher than in 2013. This is due to the delivery of ten B737-900ER aircraft, the completion of our B737 cabin improvement project, and the exercise of 16 B737 options, two Q400 options, and deposits for an incremental Q400 in 2014.

Cash Used by Financing Activities

Cash used by financing activities was \$688 million during 2015, compared to \$462 million in 2014. During the current year, we made debt payments of \$116 million, stock repurchases of \$505 million, and cash dividend payments of \$102 million. In 2014, we made debt payments of \$119 million, stock repurchases of \$348 million, and cash dividend payments of \$68 million, partially offset by proceeds from debt of \$51 million. In 2013, we made debt payments of \$161 million and stock repurchases of \$159 million.

We plan to meet our future capital and operating commitments through our cash and investments on hand, internally generated cash from operations, along with additional debt financing if necessary.

Bank Lines of Credit

The Company has two \$100 million credit facilities and a \$52 million credit facility. Information about these facilities can be found in Note 5 in the Notes to Consolidated Financial Statements in Item 8 of this Form 10-K. The Company has no immediate plans to borrow using any of these facilities.



CONTRACTUAL OBLIGATIONS AND COMMITMENTS

Aircraft Purchase Commitments

At December 31, 2015, we had firm orders to purchase, or lease, 70 aircraft. We also have options to acquire 46 B737 aircraft and options to acquire five Q400 aircraft, although we do not expect to exercise the Q400 options. In addition to the five E175 aircraft currently operated by SkyWest in our regional fleet, we have commitments to lease 18 E175 aircraft with start dates in 2016 and 2017. We have options for eight additional E175 aircraft with positions starting in late 2017 and early 2018.

As mentioned previously, we may order 30 regional jets to be operated by Horizon with deliveries starting in 2017. As no commercial terms have been reached at this time, delivery positions and future capital expenditures are excluded from the fleet table and commitment table below. However, if the order is placed in 2016, we would expect additional capital expenditures of approximately \$40 million in 2016.

The following table summarizes our projected fleet count by year, as of February 11, 2016:

	Actual Fl	eet Count	Expected Fleet Activity ^(a)							
Aircraft	Dec 31, 2014	Dec 31, 2015	2016 Changes	Dec 31, 2016	Dec 31, 2018					
B737 Freighters & Combis	6	6	_	6	(3)	3				
B737 Passenger Aircraft ^(b)	131	141	3	144	12	156				
Total Mainline Fleet	137	147	3	150	9	159				
Q400	51	52	_	52	(15)	37				
E175 (c)		5	10	15	8	23				
CRJ700 (c)	8	8	(8)	_	—	_				
Total Regional Fleet	59	65	2	67	(7)	60				
Total	196	212	5	217	2	219				

(a) The expected fleet counts at December 31, 2016, 2017, and 2018 are subject to change.

(b) 2016 changes include the expected delivery of 19 Boeing 737-900ER aircraft, offset by the return of 13 leased 737-400 aircraft and the removal of three 737-700 aircraft from our operating fleet to be converted into freighter aircraft and return to the fleet in 2017.

(c) Aircraft are operated under capacity purchase agreements with a third party.

For future firm orders and option exercises, we may finance the aircraft through internally generated cash, long-term debt, or lease arrangements.

Future Fuel Hedge Positions

All of our future oil positions are call options, which are designed to effectively cap the cost of the crude oil component of our jet fuel purchases. With call options, we are hedged against volatile crude oil price increases; and, during a period of decline in crude oil prices, we only forfeit cash previously paid for hedge premiums. Our crude oil positions are as follows:

	Approximate % of Expected Fuel Requirements	Weighted-Average Crude Oil Price per Barrel	Average Premium Cost per Barrel
First Quarter 2016	50%	\$74	\$2
Second Quarter 2016	50%	\$66	\$3
Third Quarter 2016	40%	\$66	\$3
Fourth Quarter 2016	30%	\$65	\$3
Full Year 2016	42%	\$68	\$3
First Quarter 2017	20%	\$60	\$3
Second Quarter 2017	10%	\$58	\$3
Full Year 2017	7%	\$59	\$3

Contractual Obligations

The following table provides a summary of our principal payments under current and long-term debt obligations, operating lease commitments, aircraft purchase commitments and other obligations as of December 31, 2015.

(in millions)	2	016	2017	2018	2019	2020	Beyond 2020	Total
Current and long-term debt obligations	\$	115	\$ 121	\$ 151	\$ 114	\$ 116	\$ 69	\$ 686
Operating lease commitments ^(a)		205	192	139	131	119	609	1,395
Aircraft purchase commitments (b)		505	549	444	390	327	418	2,633
Interest obligations ^(c)		32	27	21	13	7	5	105
Other obligations ^(d)		71	62	64	68	72	574	911
Total	\$	928	\$ 951	\$ 819	\$ 716	\$ 641	\$ 1,675	\$ 5,730

(a) Operating lease commitments generally include aircraft operating leases, airport property and hangar leases, office space, and other equipment leases. Included here are E175 aircraft that are operated by SkyWest under a capacity purchase agreement.

(b) Includes payments for two Q400 aircraft deliveries in 2018 that are currently contracted. At this time, however, we do not expect to take delivery of those aircraft.
 (c) For variable-rate debt, future obligations are shown above using interest rates in effect as of December 31, 2015.

(d) Includes minimum obligations associated with third-party CPAs with SkyWest and PenAir. Refer to the "Commitments" note in the consolidated financial statements for further information

Defined Benefit Pensions

The table above excludes contributions to our various pension plans, for which there are no minimum required contributions. The unfunded liability for our qualified defined-benefit pension plans was \$161 million at December 31, 2015, compared to a \$133 million unfunded position at December 31, 2014. This results in a 92% funded status on a projected benefit obligation basis compared to 94% funded as of December 31, 2014.

Credit Card Agreements

We have agreements with a number of credit card companies to process the sale of tickets and other services. Under these agreements, there are material adverse change clauses that, if triggered, could result in the credit card companies holding back a reserve from our credit card receivables. Under one such agreement, we could be required to maintain a reserve if our credit rating is downgraded to or below a rating specified by the agreement or our cash and marketable securities balance fell below \$500 million. Under another such agreement, we could be required to maintain a reserve if our cash and marketable securities balance fell below \$500 million. Under another such agreement, we could be required to maintain a reserve if our cash and marketable securities balance fell below \$500 million. We are not currently required to maintain any reserve under these agreements, but if we were, our financial position and liquidity could be materially harmed.

Deferred Income Taxes

For federal income tax purposes, the majority of our assets, as measured by value, are fully depreciated over a seven-year life using an accelerated depreciation method or bonus depreciation if available. For financial reporting purposes, the majority of our assets are depreciated over 15 to 20 years to an estimated salvage value using the straight-line basis. This difference has created a significant deferred tax liability. At some point in the future the depreciation basis will reverse, potentially resulting in an increase in income taxes paid.

While it is possible that we could have material cash obligations for this deferred liability at some point in the future, we cannot estimate the timing of longterm cash flows with reasonable accuracy. Taxable income and cash taxes payable in the short term are impacted by many items, including the amount of book income generated, which can be volatile depending on revenue and fuel prices, level of pension funding (which is generally not known until late each year), whether "bonus depreciation" provisions are available, as well as other legislative changes that are beyond our control.

In 2015, we made tax payments, net of refunds, totaling \$349 million, and had an effective tax rate of 35.4%. We expect our effective tax rate to be in the range of 36% to 38% for 2016 and the effective rate of cash paid for income taxes to be in the range of 30% to 35% of book income, although these estimates are subject to change. We believe that we will have the liquidity available to make our future tax payments.

CRITICAL ACCOUNTING ESTIMATES

The discussion and analysis of our financial position and results of operations in this MD&A are based upon our consolidated financial statements. The preparation of these financial statements requires us to make estimates and judgments that affect our financial position and results of operations. See Note 1 to the consolidated financial statements for a description of our significant accounting policies.

Critical accounting estimates are defined as those that reflect significant management judgment and uncertainties and that potentially may result in materially different results under varying assumptions and conditions. Management has identified the following critical accounting estimates and has discussed the development, selection and disclosure of these policies with our audit committee.

MILEAGE PLAN

Our Mileage $Plan^{TM}$ loyalty program awards miles to members who fly on our airlines and our airline partners. We also sell services, including miles for transportation, to non-airline partners, such as hotels, car rental agencies, and a major bank that offers Alaska Airlines affinity credit cards. In either case, the outstanding miles may be redeemed for travel on our airlines or any of our airline partners. As long as the Mileage $Plan^{TM}$ is in existence, we have an obligation to provide this future travel.

For miles earned by passengers who fly on us or our airline partners, we recognize a liability and a corresponding selling expense representing the incremental cost associated with the obligation to provide travel in the future. For services sold through one of our non-airline partners, the sales proceeds

that represent award transportation and certificates for discounted companion travel are deferred and recognized when the transportation is delivered, and the remaining components are recorded as commission in other-net revenue in the period the services are performed. Commission revenue recognized for the years ended December 31, 2015, 2014 and 2013 was \$280 million, \$261 million and \$213 million, respectively. The deferred revenue is recognized as passenger revenue when the award travel occurs, or the miles expire, and as other-net revenue for awards issued and flown on partner airlines.

At December 31, 2015, we had approximately 161 billion miles outstanding, resulting in an aggregate liability and deferred revenue balance of \$814 million. Both the liability and the deferred revenue are determined based on several assumptions that require significant management judgment to estimate and formulate. There are uncertainties inherent in these estimates; therefore, different assumptions could affect the amount and/or timing of revenue recognition or Mileage PlanTM expenses. The most significant assumptions in accounting for the Mileage PlanTM are described below.

1. The rate at which we defer sales proceeds related to services sold through non-airline partners:

We use the relative selling price method for the deferral of sales proceeds. For contracts that were modified after the effective date of Accounting Standards Update 2009-13, "Multiple-Deliverable Revenue Arrangements - a consensus of the FASB Emerging Issues Task Force" (ASU 2009-13), we determined our best estimate of selling price by considering multiple inputs and methods including, but not limited to, the estimated selling price of comparable travel, discounted cash flows, brand value, published selling prices, number of miles awarded and the number of miles redeemed. We estimated the selling prices and volumes over the terms of the agreements in order to determine the allocation of proceeds to each of the multiple deliverables. This relative allocation is evaluated annually and updated according to changes in the assumptions of the volume of related deliverables. A 1% shift between the allocation of cash proceeds to travel deliverables from marketing deliverables would defer the timing of revenue recognition by approximately \$6 million.

2. The number of miles that will not be redeemed for travel (breakage):

The liability for outstanding Mileage PlanTM mileage credits includes all mileage credits that are expected to be redeemed, including mileage credits earned by members whose mileage account balances have not yet reached the minimum mileage credit level to redeem an award. Our estimate of the number of miles that will not be redeemed (breakage) considers historical activity in our members' accounts and other factors. Based on statistical analysis of historical data, our current breakage rate is 17.4%. A hypothetical 1% change in our estimate of breakage has approximately a \$7 million effect on the liability.

3. The number of miles used per award:

We estimate how many miles will be used per award. For example, our members may redeem credit for award travel to various locations or choose between a highly restricted award and an unrestricted award. Our estimates are based on the current requirements in our Mileage PlanTM program and historical award redemption patterns.

4. The number of awards redeemed for travel on our airlines versus other airlines:

The cost for us to carry an award passenger is typically lower than the cost we will pay to our travel partners. We estimate the number of awards that will be redeemed on our airlines versus on our travel partners, and accrue the estimated costs based on historical redemption patterns. If the number of awards redeemed on our travel partner is higher or lower than estimated, we may need to adjust our liability and corresponding expense.

5. The costs that will be incurred to provide award travel:

When a frequent flier travels on his or her award ticket on one of our airlines, incremental costs such as food, fuel and insurance are incurred to carry that passenger. We estimate what these costs will be (excluding any contribution to overhead and profit) and accrue a liability. If the passenger travels on another airline on an award ticket, we often must pay the other airline for carrying the passenger. The other airline costs are based on negotiated agreements and are often substantially higher than the costs we would incur to carry that passenger. We estimate how much we will pay to other airlines for future travel awards based on historical redemptions and settlements with other carriers and accrue a liability accordingly. The costs actually incurred by us or paid to other airlines may be higher or lower than the costs that were estimated and accrued, and therefore we may need to adjust our liability and recognize a corresponding expense.

We regularly review significant Mileage Plan[™] assumptions and change our assumptions if facts and circumstances indicate that a change is necessary. Any such change in assumptions could have a significant effect on our financial position and results of operations.

PENSION PLANS

Our actuarial estimates of pension liability, the related expense, and the associated significant assumptions are discussed in Note 7.

The calculation of pension expense and the corresponding liability requires the use of a number of important assumptions, including the expected long-term rate of return on plan assets and the assumed discount rates to be used in the calculation of the projected benefit obligation and the interest and service cost. Changes in these assumptions can result in different expense and liability amounts, and future actual experience can differ from these assumptions.

Pension liability and future pension expense decrease as the discount rate used for the obligation increases. We discounted future pension obligations using rates between 4.55% and 4.69% at December 31, 2015 and a rate of 4.20% at December 31, 2014. The discount rates were determined using current rates earned on high-quality, long-term bonds with maturities that correspond with the estimated cash distributions from each of the four defined-benefit pension plans. The discount rates we use are based on a yield curve that uses a pool of higher-yielding bonds estimated to be more in line with settlement rates, as we have taken steps to ultimately terminate or settle plans that are frozen and move toward freezing benefits in active plans in the future. If the discount rate decreased by 0.5%, our projected benefit obligation at December 31, 2015 would increase by approximately \$128 million and our estimated 2016 pension expense would increase by approximately \$8 million.

Pension liability and future pension expense can increase or decrease as assumptions in the actuarial data changes. In 2015, we engaged our third-party actuary to update demographic assumptions used in the valuation of the defined-benefit pension liabilities. These assumptions included updates such as estimated salary increases, employee turnover and retirement rates, among other items. The changes in these assumptions decreased our projected benefit obligation by approximately \$50 million at December 31, 2015. The primary cause of the decline was the change in expected participant retirement rates. The study found that eligible employees are retiring later than they had in our previous assumption update, resulting in lower expected cash flows post retirement.

Pension expense normally increases as the expected rate of return on pension plan assets decreases. As of December 31, 2015, we estimate that the pension plan assets will generate long-term rates of return between 6.00% and 6.50%, which compares to 6.5% at December 31, 2014. We regularly review the actual asset allocation and periodically rebalance investments as appropriate. This expected long-term rate of return on plan assets at December 31, 2015 is based on an allocation of U.S. and non-U.S. equities, U.S. fixed-income securities, and real estate. A decrease in the expected long-term rate of return of 0.5% would increase our estimated 2016 pension expense by approximately \$8 million.

All of our defined-benefit pension plans are now closed to new entrants. Additionally, benefits in our non-union defined-benefit plans were frozen January 1, 2014. Furthermore, in 2015, we offered lump sum payments to certain terminated, vested participants in our defined-benefit pension plans in order to eliminate a significant portion of our outstanding liability. As a result, we reduced our projected benefit obligation by \$62 million and recorded a settlement charge of \$14 million in 2015, which is included in Special Items on the consolidated statement of operations.

Future changes in plan asset returns, assumed discount rates and various other factors related to the participants in our pension plans will impact our future pension expense and liabilities. We cannot predict what these factors will be in the future.

LONG-LIVED ASSETS

As of December 31, 2015, we had approximately \$4.8 billion of property and equipment and related assets, net of accumulated depreciation. In accounting for these long-lived assets, we make estimates about the expected useful lives of the assets, changes in fleet plans, the expected residual values of the assets, and the potential for impairment based on the fair value of the assets and the cash flows they generate. Factors indicating potential impairment include, but are not limited to, significant decreases in the market value of the long-lived assets, management decisions regarding the future use of the assets, a significant change in the long-lived assets condition, and operating cash flow losses associated with the use of the long-lived asset.

There is inherent risk in estimating the fair value of our aircraft and related parts and their salvage values at the time of impairment. Actual proceeds upon disposition of the aircraft or related parts could be materially less than expected, resulting in additional loss. Our estimate of salvage value at the time of disposal could also change, requiring us to increase the depreciation expense on the affected aircraft.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURE ABOUT MARKET RISK

We have interest-rate risk on our variable-rate debt obligations and our available-for-sale marketable investment portfolio, and commodity-price risk in jet fuel required to operate our aircraft fleet. We purchase the majority of our jet fuel at prevailing market prices and seek to manage market risk through execution of our hedging strategy and other means. We have market-sensitive instruments in the form of fixed-rate debt instruments, and financial derivative instruments used to hedge our exposure to jet-fuel price increases and interest-rate increases. We do not purchase or hold any derivative financial instruments for trading purposes.

Aircraft Fuel

Currently, our fuel-hedging portfolio consists of crude oil call options. Call options effectively cap our pricing for the crude oil, limiting our exposure to increasing fuel prices for about half of our planned fuel consumption. With call options, we are hedged against volatile crude oil price increases, and during a period of decline in crude oil prices, we only forfeit cash previously paid for hedge premiums. We believe there is risk in not hedging against the possibility of fuel price increases. We estimate that a 10% change in the forward curve for crude oil prices as of December 31, 2015 would increase or decrease the fair value of our crude oil hedge portfolio to approximately \$8 million or \$2 million, respectively.

Our portfolio value of fuel hedge contracts was \$4 million at December 31, 2015 compared to a portfolio value of \$7 million at December 31, 2014. We do not have any collateral held by counterparties to these agreements as of December 31, 2015.

We continue to believe that our fuel hedge program is an important part of our strategy to reduce our exposure to volatile fuel prices. We expect to continue to enter into these types of contracts prospectively, although significant changes in market conditions could affect our decisions. For more discussion, see the "Derivative Instruments" note in our consolidated financial statements.

Interest Rates

We have exposure to market risk associated with changes in interest rates related primarily to our debt obligations and short-term investment portfolio. Our debt obligations include variable-rate instruments, which have exposure to changes in interest rates. This exposure is somewhat mitigated through our variable-rate investment portfolio. A hypothetical 10% change in the average interest rates incurred on variable-rate debt during 2015 would correspondingly change our net earnings and cash flows associated with these items by less than \$1 million. In order to help mitigate the risk of interest rate fluctuations, we have fixed the interest rates on certain existing variable-rate debt agreements. Our variable-rate debt is approximately 24% of our total long-term debt at December 31, 2015 and December 31, 2014.

We also have investments in marketable securities, which are exposed to market risk associated with changes in interest rates. If short-term interest rates were to average 1% more than they did in 2015, interest income would increase by approximately \$14 million.

ITEM 8. CONSOLIDATED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

SELECTED QUARTERLY CONSOLIDATED FINANCIAL INFORMATION (unaudited)

	 First Quarter				Second Quarter			Third Quarter				Fourth Quarter			
(in millions, except per share)	2015		2014		2015 2014		2014		2015 201		2014	2014 2015		2014	
Operating revenues	\$ 1,269	\$	1,222	\$	1,437	\$	1,375	\$	1,515	\$	1,465	\$	1,377	\$	1,306
Operating income	238		141		372		263		433		316		255		242
Net income	149		94		234		165		274		198		191		148
Basic earnings per share ^(a)	1.13		0.69		1.80		1.20		2.15		1.47		1.52		1.12
Diluted earnings per share ^(a)	1.12		0.68		1.79		1.19		2.14		1.45		1.51		1.11

^(a) For earnings per share, the sum of the quarters may not equal the total for the full year due to rounding.

The Board of Directors and Shareholders Alaska Air Group, Inc.:

We have audited the accompanying consolidated balance sheets of Alaska Air Group, Inc. and subsidiaries as of December 31, 2015 and 2014, and the related consolidated statements of operations, comprehensive operations, shareholders' equity, and cash flows for each of the years in the three-year period ended December 31, 2015. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Alaska Air Group, Inc. and subsidiaries as of December 31, 2015 and 2014, and the results of their operations and their cash flows for each of the years in the three-year period ended December 31, 2015, in conformity with U.S. generally accepted accounting principles.

As discussed in Note 10 to the consolidated financial statements, the Company's affinity card agreement was materially modified effective July 2, 2013. As a result, the Company changed its method of accounting for consideration received under this agreement in accordance with Accounting Standards Update No. 2009-13, Multiple Deliverable Revenue Arrangements, in 2013.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), Alaska Air Group, Inc.'s internal control over financial reporting as of December 31, 2015, based on criteria established in *Internal Control - Integrated Framework* (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO), and our report dated February 11, 2016 expressed an unqualified opinion on the effectiveness of the Company's internal control over financial reporting.

/s/ KPMG LLP

Seattle, Washington February 11, 2016

CONSOLIDATED BALANCE SHEETS

As of December 31 (in millions)	2015	2014
ASSETS		
Current Assets		
Cash and cash equivalents	\$ 73	\$ 107
Marketable securities	1,255	1,110
Total cash and marketable securities	1,328	1,217
Receivables - less allowance for doubtful accounts of \$1 and \$1	212	259
Inventories and supplies - net	51	58
Prepaid expenses and other current assets	72	105
Total Current Assets	1,663	1,639
Property and Equipment		
Aircraft and other flight equipment	5,690	5,165
Other property and equipment	955	896
Deposits for future flight equipment	771	555
	7,416	6,616
Less accumulated depreciation and amortization	2,614	2,317
Total Property and Equipment - Net	4,802	4,299
Other Assets	68	126
Total Assets	\$ 6,533	\$ 6,064

See accompanying notes to consolidated financial statements.

CONSOLIDATED BALANCE SHEETS - (continued)

As of December 31 (in millions except share amounts)	2015	2014
LIABILITIES AND SHAREHOLDERS' EQUITY		-
Current Liabilities		
Accounts payable	63	62
Accrued wages, vacation and payroll taxes	298	276
Air traffic liability	669	631
Other accrued liabilities	661	585
Current portion of long-term debt	 115	117
Total Current Liabilities	 1,806	1,671
Long-Term Debt, Net of Current Portion	 571	686
Other Liabilities and Credits		
Deferred income taxes	682	633
Deferred revenue	431	374
Obligation for pension and postretirement medical benefits	270	246
Other liabilities	 362	327
	 1,745	1,580
Commitments and Contingencies		
Shareholders' Equity		-
Preferred stock, \$0.01 par value Authorized: 5,000,000 shares, none issued or outstanding	—	
Common stock, \$0.01 par value Authorized: 200,000,000 shares, Issued: 2015 - 128,442,099 shares; 2014 - 131,556,573 shares, Outstanding: 2015 - 125,175,325 shares; 2014 - 131,481,473 shares	1	1
Capital in excess of par value	73	296
Treasury stock (common), at cost: 2015 - 3,266,774 shares; 2014 - 75,100 shares	(250)	(4)
Accumulated other comprehensive loss	(303)	(310)
Retained earnings	2,890	2,144
	2,411	2,12
Total Liabilities and Shareholders' Equity	\$ 6,533	\$ 6,064

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF OPERATIONS

Year Ended December 31 (in millions, except per-share amounts)	2015		2014		2013
Operating Revenues					
Passenger					
Mainline	\$ 3,93	•	3,774	\$	3,490
Regional	85	1	805		777
Total passenger revenue	4,79	3	4,579		4,267
Freight and mail	10	8	114		113
Other - net	69	7	675		584
Special mileage plan revenue	-	-			192
Total Operating Revenues	5,59	3	5,368		5,156
Operating Expenses					
Wages and benefits	1,25	1	1,136		1,086
Variable incentive pay	12		116		105
Aircraft fuel, including hedging gains and losses	95	1	1,418		1,467
Aircraft maintenance	25	3	229		247
Aircraft rent	10	5	110		119
Landing fees and other rentals	29	6	279		262
Contracted services	21	1	196		176
Selling expenses	21	l	199		179
Depreciation and amortization	32)	294		270
Food and beverage service	11	3	93		84
Third-party regional carrier expense	7.	2	58		45
Other	35	6	308		278
Special items	3	2	(30)		_
Total Operating Expenses	4,30)	4,406		4,318
Operating Income	1,29		962		838
Nonoperating Income (Expense)					
Interest income	2	l	21		18
Interest expense	(4	2)	(48)		(56)
Interest capitalized	3	1	20		21
Other - net		l	20		(5)
	1	1	13		(22)
Income before income tax	1,31	2	975		816
Income tax expense	46	1	370		308
Net Income	\$ 84	8 \$	605	\$	508
Basic Earnings Per Share:	\$ 6.6	L \$	4.47	\$	3.63
Diluted Earnings Per Share:	\$ 6.5		4.42	\$	3.58
Shares used for computation:		Ψ	1.12	Ψ	5.50
Basic	128.37	3	135.445		139.910
Diluted	129.37		136.801		141.878
	<i>ا کہ لا</i> نی <u>ت</u>	-	150.001		111.070
Cash dividend declared per share	\$ 0.8) \$	0.50	\$	0.20

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF COMPREHENSIVE OPERATIONS

Year Ended December 31 (in millions)	 2015	2014	2013
Net Income	\$ 848	\$ 605	\$ 508
Other Comprehensive Income (Loss):			
Related to marketable securities:			
Unrealized holding gains (losses) arising during the period	(6)	2	(9)
Reclassification of (gains) losses into Other-net nonoperating income (expense)	1	(2)	(2)
Income tax benefit (expense)	2	—	4
Total	 (3)		(7)
Related to employee benefit plans:			
Actuarial gains/(losses) related to pension and other post retirement benefit plans	10	(210)	358
Reclassification of net pension expense into Wages and benefits	14	9	42
Income tax benefit (expense)	(14)	76	(150)
Total	 10	(125)	250
Related to interest rate derivative instruments:			
Unrealized holding gains (losses) arising during the period	(5)	(8)	10
Reclassification of (gains) losses into Aircraft rent	6	6	6
Income tax benefit (expense)	(1)	—	(6)
Total	 	(2)	10
Other Comprehensive Income (Loss)	7	(127)	253
Comprehensive Income	\$ 855	\$ 478	\$ 761

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY

(in millions)	Common Stock Outstanding	Common Stock	Capital in Excess of Par Value	Treasury Stock	Accumulated Other Comprehensive Income (Loss)	Retained Earnings	Total
Balances at December 31, 2012	140.754	\$ 1	\$ 729	\$ —	\$ (436)	\$ 1,127	\$ 1,421
2013 net income				_		508	508
Other comprehensive income/(loss)	—	—		—	253	—	253
Common stock repurchase	(4.984)	—	(157)	(2)	—		(159)
Stock-based compensation	—	—	16	—	—	—	16
Cash dividend declared	—	—		—		(28)	(28)
Stock issued for employee stock purchase plan	0.342	_	6	_		_	6
Stock issued under stock plans	1.380	—	12	—		—	12
Balances at December 31, 2013	137.492	1	606	(2)	(183)	1,607	2,029
2014 net income						605	605
Other comprehensive income/(loss)		_			(127)		(127)
Common stock repurchase	(7.317)		(346)	(2)	—		(348)
Stock-based compensation		_	16		_		16
Cash dividend declared	—	—		—	—	(68)	(68)
Stock issued for employee stock purchase plan	0.299	_	9	_	_	_	9
Stock issued under stock plans	1.007		11		_	_	11
Balances at December 31, 2014	131.481	1	296	(4)	(310)	2,144	2,127
2015 net income						848	848
Other comprehensive income/(loss)	_		_		7		7
Common stock repurchase	(7.208)		(259)	(246)	_	_	(505)
Stock-based compensation		_	17				17
Cash dividend declared	_				—	(102)	(102)
Stock issued for employee stock purchase plan	0.281	_	13	_	_	_	13
Stock issued under stock plans	0.621		6				6
Balances at December 31, 2015	125.175	\$ 1	\$ 73	\$ (250)	\$ (303)	\$ 2,890	\$ 2,411

See accompanying notes to consolidated financial statements.

CONSOLIDATED STATEMENTS OF CASH FLOWS

Year Ended December 31 (in millions)	2015		2014	2013
Cash flows from operating activities:				
Net income	\$ 1	848	\$ 605	\$ 508
Adjustments to reconcile net income to net cash provided by operating activities:				
Special mileage plan accounting adjustment		—	—	(192
Depreciation and amortization	2	320	294	270
Stock-based compensation and other		25	6	32
Changes in certain assets and liabilities:				
Changes in deferred tax provision		56	114	146
(Increase) decrease in accounts receivable		47	(110)	(19
Increase (decrease) in air traffic liability		38	67	29
Increase (decrease) in deferred revenue		57	40	84
Changes in pension and other postretirement benefits		36	(18)	62
Other - net	1	57	32	148
Pension contribution		—	_	(87
Net cash provided by operating activities	1,4	584	1,030	 981
Cash flows from investing activities:				
Property and equipment additions:				
Aircraft and aircraft purchase deposits	(681)	(498)	(487
Other flight equipment		(79)	(131)	(41
Other property and equipment		(71)	(65)	(38
Total property and equipment additions	(8	31)	(694)	 (566
Purchases of marketable securities	(1,3	327)	(949)	(1,218
Sales and maturities of marketable securities	1,1	75	1,092	1,089
Proceeds from disposition of assets and changes in restricted deposits		53	10	(3
Net cash used in investing activities	()	30)	(541)	 (698
Cash flows from financing activities:				
Proceeds from issuance of long-term debt			51	_
Long-term debt payments	(1	16)	(119)	(161
Common stock repurchases	(*	505)	(348)	(159
Cash dividend paid	(1	02)	(68)	(28
Other financing activities		35	22	23
Net cash used in financing activities	(588)	(462)	 (325
Net increase (decrease) in cash and cash equivalents		(34)	27	 (42
Cash and cash equivalents at beginning of year	1	07	80	122
Cash and cash equivalents at end of year	\$	73	\$ 107	\$ 80
Supplemental disclosure:				
Cash paid during the year for:				
Interest, net of amount capitalized	\$	8	\$ 28	\$ 35
Income taxes, net of refunds received	3	649	326	149

See accompanying notes to consolidated financial statements.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Alaska Air Group, Inc. December 31, 2015

NOTE 1. GENERAL AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Organization and Basis of Presentation

The consolidated financial statements include the accounts of Alaska Air Group, Inc. (Air Group or the Company) and its subsidiaries, Alaska Airlines, Inc. (Alaska) and Horizon Air Industries, Inc. (Horizon), through which the Company conducts substantially all of its operations. All significant intercompany balances and transactions have been eliminated. These financial statements have been prepared in conformity with accounting principles generally accepted in the United States of America and their preparation requires the use of management's estimates. Actual results may differ from these estimates.

Certain reclassifications have been made to prior year financial statements to conform to classifications used in the current year.

Cash and Cash Equivalents

Cash equivalents consist of highly liquid investments with original maturities of three months or less, such as money market funds, commercial paper and certificates of deposit. They are carried at cost, which approximates market value. The Company reduces cash balances when funds are disbursed. Due to the time delay in funds clearing the banks, the Company normally maintains a negative balance in its cash disbursement accounts, which is reported as a current liability. The amount of the negative cash balance was \$12 million and \$7 million at December 31, 2015 and 2014, respectively, and is included in accounts payable, with the change in the balance during the year included in other financing activities in the consolidated statements of cash flows.

The Company has restricted cash balances primarily used to guarantee various letters of credit, self-insurance programs, or other contractual rights. Restricted cash consists of highly liquid securities with original maturities of three months or less. They are carried at cost, which approximates fair value.

Marketable Securities

Investments with original maturities of greater than three months and remaining maturities of less than one year are classified as short-term investments. Investments with maturities beyond one year may be classified as short-term based on their highly liquid nature and because such marketable securities represent the investment of cash that is available for current operations. All cash equivalents and short-term investments are classified as available-for-sale and realized gains and losses are recorded using the specific identification method. Changes in market value, excluding other-than-temporary impairments, are reflected in accumulated other comprehensive loss (AOCL).

Investments are considered to be impaired when a decline in fair value is judged to be other-than-temporary. The Company uses a systematic methodology that considers available quantitative and qualitative evidence in evaluating potential impairment. If the cost of an investment exceeds its fair value, management evaluates, among other factors, general market conditions, credit quality of debt instrument issuers, the duration and extent to which the fair value is less than cost, our intent and ability to hold, or plans to sell, the investment. Once a decline in fair value is determined to be other-than-temporary, an impairment charge is recorded to Other-net in the consolidated statements of operations and a new cost basis in the investment is established.

Receivables

Receivables are due on demand and consist primarily of airline traffic (including credit card) receivables, Mileage PlanTM partners, amounts due from other airlines related to interline agreements, government tax authorities, and other miscellaneous amounts due to the Company, and are net of an allowance for doubtful accounts. Management determines the allowance for doubtful accounts based on known troubled accounts and historical experience applied to an aging of accounts.



Inventories and Supplies—net

Expendable aircraft parts, materials and supplies are stated at average cost and are included in inventories and supplies—net. An obsolescence allowance for expendable parts is accrued based on estimated lives of the corresponding fleet type and salvage values. The allowance for all non-surplus expendable inventories was \$37 million and \$34 million at December 31, 2015 and 2014, respectively. Inventory and supplies—net also includes fuel inventory of \$14 million and \$21 million at December 31, 2015 and 2014, respectively. Repairable aircraft parts inventories are included in flight equipment.

Property, Equipment and Depreciation

Property and equipment are recorded at cost and depreciated using the straight-line method over their estimated useful lives less an estimated salvage value, which are as follows:

Aircraft and related flight equipment:	
Boeing 737 aircraft	20 years
Bombardier Q400	15 years
Buildings	25-30 years
Minor building and land improvements	10 years
Capitalized leases and leasehold improvements	Shorter of lease term or estimated useful life
Computer hardware and software	3-5 years
Other furniture and equipment	5-10 years

Salvage values used for aircraft are 10% of the fair value, but as aircraft near the end of their useful lives, we update the salvage value estimates based on current market conditions and expected use of the aircraft. "Related flight equipment" includes rotable and repairable spare inventories, which are depreciated over the associated fleet life unless otherwise noted.

Capitalized interest is based on the Company's weighted-average borrowing rate, is added to the cost of the related asset, and is depreciated over the estimated useful life of the asset.

Maintenance and repairs, other than engine maintenance on some B737-700 and -900 engines, are expensed when incurred. Major modifications that extend the life or improve the usefulness of aircraft are capitalized and depreciated over their estimated period of use. Maintenance on some B737-700 and -900 engines is covered under power-by-the-hour agreements with third parties, whereby the Company pays a determinable amount, and transfers risk, to a third party. The Company expenses the contract amounts based on engine usage.

The Company evaluates long-lived assets to be held and used for impairment whenever events or changes in circumstances indicate that the total carrying amount of an asset or asset group may not be recoverable. The Company groups assets for purposes of such reviews at the lowest level for which identifiable cash flows of the asset group are largely independent of the cash flows of other groups of assets and liabilities. An impairment loss is considered when estimated future undiscounted cash flows expected to result from the use of the asset or asset group and its eventual disposition are less than its carrying amount. If the asset or asset group is not considered recoverable, a write-down equal to the excess of the carrying amount over the fair value will be recorded.

Internally Used Software Costs

The Company capitalizes costs to develop internal-use software that are incurred in the application development stage. Amortization commences when the software is ready for its intended use and the amortization period is the estimated useful life of the software, generally three to five years. Capitalized costs primarily include contract labor and payroll costs of the individuals dedicated to the development of internal-use software.

Deferred Revenue

Deferred revenue results primarily from the sale of Mileage PlanTM miles to third-parties. This revenue is recognized when award transportation is provided or over the term of the applicable agreement.

Operating Leases

The Company leases aircraft, airport and terminal facilities, office space, and other equipment under operating leases. Some of these lease agreements contain rent escalation clauses or rent holidays. For scheduled rent escalation clauses during the lease terms or for rental payments commencing at a date other than the date of initial occupancy, the Company records minimum rental expenses on a straight-line basis over the terms of the leases in the consolidated statements of operations.

Leased Aircraft Return Costs

Cash payments associated with returning leased aircraft are accrued when it is probable that a cash payment will be made and that amount is reasonably estimable. Any accrual is based on the time remaining on the lease, planned aircraft usage and the provisions included in the lease agreement, although the actual amount due to any lessor upon return will not be known with certainty until lease termination.

As leased aircraft are returned, any payments are charged against the established accrual. The accrual is part of other current and long-term liabilities, and was \$5 million and \$1 million as of December 31, 2015 and December 31, 2014, respectively.

Revenue Recognition

Passenger revenue is recognized when the passenger travels. Tickets sold but not yet used are reported as air traffic liability until travel or date of expiration. Air traffic liability includes approximately \$42 million and \$33 million related to credits for future travel, as of December 31, 2015 and December 31, 2014, respectively. These credits are recognized into revenue either when the passenger travels or the date of expiration, which is twelve months from issuance. Commissions to travel agents and related fees are expensed when the related revenue is recognized. Passenger traffic commissions and related fees not yet recognized are included as a prepaid expense. Taxes collected from passengers, including transportation excise taxes, airport and security fees and other fees, are recorded on a net basis within passenger revenue in the consolidated statements of operations. Due to complex pricing structures, refund and exchange policies, and interline agreements with other airlines, certain amounts are recognized as revenue using estimates regarding both the timing of the revenue recognition and the amount of revenue to be recognized. These estimates are based on the Company's historical data.

Freight and mail revenues are recognized when service is provided.

Other - net revenues are primarily related to the Mileage Plan[™] and they are recognized as described in the "Mileage Plan" paragraph below. Other - net also includes certain ancillary or non-ticket revenues, such as checked-bag fees, reservations fees, ticket change fees, on-board food and beverage sales, and to a much lesser extent commissions from car and hotel vendors, and from the sales of travel insurance. These items are recognized as revenue when the related services are provided. Airport lounge memberships are recognized as revenue over the membership period.

Mileage Plan

Alaska operates a frequent flier program ("Mileage Plan[™]") that provides travel awards to members based on accumulated mileage. For miles earned by flying on Alaska or Horizon and through airline partners, the estimated cost of providing award travel is recognized as a selling expense and accrued as a liability as miles are earned and accumulated.

Alaska also sells services, including miles for transportation, to non-airline partners, such as hotels, car rental agencies, and a major bank that offers Alaska Airlines affinity credit cards. The Company defers passenger revenue related to air transportation and certificates for discounted companion travel until the transportation is delivered. The deferred proceeds are recognized as passenger revenue for awards redeemed and flown on Alaska or Horizon, and as Other-net revenue for awards redeemed and flown on other airlines (less the cost paid to the other airlines based on contractual agreements). For the elements that represent use of the Alaska Airlines brand and access to frequent flier member lists and advertising, it is recognized as commission income in the period that those elements are sold and included in Other - net revenue in the consolidated statements of operations.



Alaska's Mileage Plan[™] deferred revenue and liabilities on the consolidated balance sheets (in millions):

	2015	2014
Current Liabilities:		
Other accrued liabilities	\$ 368	\$ 343
Other Liabilities and Credits:		
Deferred revenue	427	367
Other liabilities	19	20
Total	\$ 814	\$ 730

The amounts recorded in other accrued liabilities relate primarily to deferred revenue expected to be realized within one year, which includes Mileage PlanTM awards that have been issued but not yet flown for 37 million and 33 million at December 31, 2015 and 2014, respectively.

Alaska's Mileage Plan™ revenue included in the consolidated statements of operations (in millions):

	20	15	:	2014	2013
Passenger revenues	\$	267	\$	246	\$ 208
Other-net revenues		329		295	256
Special mileage plan revenue ^(a)		_		—	192
Total Mileage Plan revenues	\$	596	\$	541	\$ 656

(a) Refer to Note 10 for further information.

Other--net revenue includes commission revenues of \$280 million, \$261 million, and \$213 million in 2015, 2014, and 2013, respectively.

Selling Expenses

Selling expenses include credit card fees, global distribution systems charges, the estimated cost of Mileage Plan[™] travel awards earned through air travel, advertising, promotional costs, commissions, and incentives. Advertising production costs are expensed the first time the advertising takes place. Advertising expense was \$55 million, \$49 million, and \$28 million during the years ended December 31, 2015, 2014, and 2013, respectively.

Derivative Financial Instruments

The Company's operations are significantly impacted by changes in aircraft fuel prices and interest rates. In an effort to manage our exposure to these risks, the Company periodically enters into fuel and interest rate derivative instruments. These derivative instruments are recognized at fair value on the balance sheet and changes in the fair value is recognized in AOCL or in the consolidated statements of operations, depending on the nature of the instrument.

The Company does not hold or issue derivative fuel hedge contracts for trading purposes and does not apply hedge accounting. For cash flow hedges related to our interest rate swaps, the effective portion of the derivative represents the change in fair value of the hedge that offsets the change in fair value of the hedge ditem. To the extent the change in the fair value of the hedge does not perfectly offset the change in the fair value of the hedged item, the ineffective portion of the hedge is immediately recognized in interest expense.

Fair Value Measurements

Accounting standards define fair value as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. The standards also establish a fair value hierarchy, which requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. There are three levels of inputs that may be used to measure fair value:

Level 1 - Quoted prices in active markets for identical assets or liabilities.



Level 2 - Observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities, quoted prices in markets that are not active, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.

Level 3 - Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The Company has elected not to use the Fair Value Option for non-financial instruments, and accordingly those assets and liabilities are carried at amortized cost. For financial instruments, those assets and liabilities are carried at fair value and are determined based on the market approach or income approach depending upon the level of inputs used.

Income Taxes

The Company uses the asset and liability approach for accounting and reporting income taxes. Deferred tax assets and liabilities are recognized for future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases, and for operating loss and tax credit carryforwards. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in the period that includes the enactment date. A valuation allowance would be established, if necessary, for the amount of any tax benefits that, based on available evidence, are not expected to be realized. The Company accounts for unrecognized tax benefits in accordance with the accounting standards.

Stock-Based Compensation

Accounting standards require companies to recognize as expense the fair value of stock options and other equity-based compensation issued to employees as of the grant date. These standards apply to all stock awards that the Company grants to employees as well as the Company's Employee Stock Purchase Plan (ESPP), which features a look-back provision and allows employees to purchase stock at a 15% discount. All stock-based compensation expense is recorded in wages and benefits in the consolidated statements of operations.

Earnings Per Share (EPS)

Diluted EPS is calculated by dividing net income by the average common shares outstanding plus additional common shares that would have been outstanding assuming the exercise of in-the-money stock options and restricted stock units, using the treasury-stock method. In 2015, 2014, and 2013, antidilutive stock options excluded from the calculation of EPS were not material.

Recently Issued Accounting Pronouncements

In May 2014, the FASB issued Accounting Standard Update 2014-09, "Revenue from Contracts with Customers" (ASU 2014-09), which requires an entity to recognize the amount of revenue to which it expects to be entitled for the transfer of promised goods or services to customers. The ASU will replace most existing revenue recognition guidance in U.S. GAAP when it becomes effective. The new standard is effective for the Company on January 1, 2018. The standard permits the use of either the retrospective or cumulative effect transition method. At this time, the Company believes the most significant impact to the financial statements will be in Mileage Plan revenues and liabilities. The Company currently uses the incremental cost approach for miles earned through travel. This standard eliminates that option and the Company will be required to increase its liability for earned miles through a relative selling price model. The Company has not evaluated the full impact of the standard, although application is expected to result in a material increase to Deferred Revenue. The Company has not yet selected a transition method.

In July 2015, FASB issued ASU 2015-07, Disclosures for Investments in Certain Entities that Calculate Net Asset Value per Share, which removes the requirement for companies to disclose the fair value hierarchy for assets calculated at net asset value per share in common commingled trusts, for example. This standard is effective January 1, 2016 for the Company. However, early adoption is permitted and the Company has elected to adopt this standard in 2015. The Company's pension assets are invested in common commingled trusts and, as such, the Company has removed the disclosure of the fair value hierarchy for pension assets in common commingled trusts.

In November 2015, FASB issued ASU 2015-17, Balance Sheet Classification of Deferred Taxes, which simplifies the presentation of deferred income taxes. This ASU requires that net deferred tax assets and liabilities be classified as noncurrent



on the balance sheet. The Company has elected to early adopt ASU 2015-17 as of December 31, 2015, as permitted by the FASB, and has retrospectively applied the standard. This has resulted in a reduction of total current assets and corresponding reduction of other liabilities of \$117 million as of December 31, 2014.

NOTE 2. CASH, CASH EQUIVALENTS AND MARKETABLE SECURITIES

Components for cash, cash equivalents and marketable securities (in millions):

December 31, 2015	Cost Basis	Unrealized Gains	Unrealized Losses		Fair Value
Cash	\$ 4	\$ —	\$ —	\$	4
Cash equivalents	69		—		69
Cash and cash equivalents	 73	 —	_		73
U.S. government and agency securities	 254	 _	(1)		253
Foreign government bonds	31	—	—		31
Asset-backed securities	130	—	—		130
Mortgage-backed securities	117	—	(1)		116
Corporate notes and bonds	711	1	(4)		708
Municipal securities	 17	 			17
Marketable securities	1,260	1	(6)		1,255
Total	\$ 1,333	\$ 1	\$ (6)	\$	1,328

December 31, 2014	Cost Basis	Unrealized Gains		Unrealized Losses		Fair Value
Cash	\$ 4	\$ —	\$	_	\$	4
Cash equivalents	103			—		103
Cash and cash equivalents	 107	 —		_		107
U.S. government and agency securities	 166	 _		_		166
Foreign government bonds	25	—		—		25
Asset-backed securities	130	—		—		130
Mortgage-backed securities	127	—		(1)		126
Corporate notes and bonds	644	3		(2)		645
Municipal securities	 18	 				18
Marketable securities	 1,110	3		(3)	_	1,110
Total	\$ 1,217	\$ 3	\$	(3)	\$	1,217

Unrealized losses from fixed-income securities are primarily attributable to changes in interest rates. Management does not believe any remaining unrealized losses represent other-than-temporary impairments based on our evaluation of available evidence as of December 31, 2015.

Activity for marketable securities (in millions):

 2015	2014		2013
\$ 1,175	\$ 1,092	\$	1,089
2	4		4
(3)	(2)		(2)
\$	 	\$ 1,175 \$ 1,092 2 4	\$ 1,175 \$ 1,092 \$ 2 4

Maturities for marketable securities (in millions):

December 31, 2015	Cost Basis		Fair Value	
Due in one year or less	\$	182	\$	182
Due after one year through five years		1,055		1,050
Due after five years through 10 years		14		14
Due after 10 years		9		9
Total	\$	1,260	\$	1,255

NOTE 3. DERIVATIVE INSTRUMENTS

Fuel Hedge Contracts

The Company's operations are inherently dependent upon the price and availability of aircraft fuel. To manage economic risks associated with fluctuations in aircraft fuel prices, the Company periodically enters into call options for crude oil.

As of December 31, 2015, the Company had fuel hedge contracts outstanding covering 273 million gallons of crude oil that will be settled from January 2016 to June 2017. Refer to the contractual obligations and commitments section of Item 7 for further information.

Interest Rate Swap Agreements

The Company has interest rate swap agreements with a third party designed to hedge the volatility of the underlying variable interest rate in the Company's aircraft lease agreements for six Boeing 737-800 aircraft. The agreements stipulate that the Company pay a fixed interest rate over the term of the contract and receive a floating interest rate. All significant terms of the swap agreement match the terms of the lease agreements, including interest-rate index, rate reset dates, termination dates and underlying notional values. The agreements expire from February 2020 through March 2021 to coincide with the lease termination dates.

Fair Values of Derivative Instruments

Fair values of derivative instruments on the consolidated balance sheet (in millions):

	2015		_	2014
Derivative Instruments Not Designated as Hedges				
Fuel hedge contracts				
Prepaid expenses and other current assets	\$	2	\$	3
Other assets		2		4
Derivative Instruments Designated as Hedges				
Interest rate swaps				
Other accrued liabilities		(5)		(6)
Other liabilities		(13)		(13)
Losses in accumulated other comprehensive loss (AOCL)		(18)		(19)

The net cash received (paid) for new fuel hedge positions and settlements was \$(17) million, \$(9) million, and \$5 million during 2015, 2014, and 2013, respectively.

Pretax effect of derivative instruments on earnings and AOCL (in millions):

	2015	2014	2013
Derivative Instruments Not Designated as Hedges			
Fuel hedge contracts			
Gains (losses) recognized in aircraft fuel expense	\$ (19)	\$ (18)	\$ (44)
Derivative Instruments Designated as Hedges			
Interest rate swaps			
Gains (losses) recognized in aircraft rent	(6)	(6)	(6)
Gains (losses) recognized in other comprehensive income (OCI)	(5)	(8)	10

The amounts shown as recognized in aircraft rent for cash flow hedges (interest rate swaps) represent the realized losses transferred out of AOCL to aircraft rent. The amounts shown as recognized in OCI are prior to the losses recognized in aircraft rent during the period. The Company expects \$5 million to be reclassified from OCI to aircraft rent within the next twelve months.

Credit Risk and Collateral

The Company is exposed to credit losses in the event of non-performance by counterparties to these derivative instruments. To mitigate exposure, the Company periodically reviews the risk of counterparty nonperformance by monitoring the absolute exposure levels and credit ratings. The Company maintains security agreements with a number of its counterparties which may require the Company to post collateral if the fair value of the selected derivative instruments fall below specified thresholds. The posted collateral does not offset the fair value of the derivative instruments and is included in "Prepaid expenses and other current assets" on the consolidated balance sheet.

The amount posted as collateral for these contracts is not material to the consolidated balance sheets as of December 31, 2015, and 2014.

NOTE 4. FAIR VALUE MEASUREMENTS

Fair Value of Financial Instruments on a Recurring Basis

Fair values of financial instruments on the consolidated balance sheet (in millions):

December 31, 2015	 Level 1	Level 2		Total	
Assets					
Marketable securities					
U.S. government and agency securities	\$ 253	\$	_	\$	253
Foreign government bonds			31		31
Asset-backed securities	_		130		130
Mortgage-backed securities	_		116		116
Corporate notes and bonds	_		708		708
Municipal securities	_		17		17
Derivative instruments					
Fuel hedge contracts					
Call options	—		4		4
Liabilities					
Derivative instruments					
Interest rate swap agreements	—		(18)		(18)



December 31, 2014	Level 1	Level 2		Total
Assets				
Marketable securities				
U.S. government and agency securities	\$ 166	\$	—	\$ 166
Foreign government bonds	—		25	25
Asset-backed securities	—		130	130
Mortgage-backed securities	—		126	126
Corporate notes and bonds	—		645	645
Municipal securities	—		18	18
Derivative instruments				
Fuel hedge contracts				
Call options	—		7	7
Liabilities				
Derivative instruments				
Interest rate swap agreements	_		(19)	(19)

The Company uses the market and income approach to determine the fair value of marketable securities. U.S. government securities are Level 1 as the fair value is based on quoted prices in active markets. Foreign government bonds, asset-backed securities, mortgage-backed securities, corporate notes and bonds, and municipal securities are Level 2 as the fair value is based on standard valuation models that are calculated based on observable inputs such as quoted interest rates, yield curves, credit ratings of the security and other observable market information.

The Company uses the market approach and the income approach to determine the fair value of derivative instruments. Fuel hedge contracts that are not traded on a public exchange are Level 2 as the fair value is primarily based on inputs which are readily available in active markets or can be derived from information available in active markets. The fair value for call options is determined utilizing an option pricing model based on inputs that are readily available in active markets, or can be derived from information available in active markets, or can be derived from information available in active markets, or can be derived from information available in active markets. In addition, the fair value considers the exposure to credit losses in the event of non-performance by counterparties. Interest rate swap agreements are Level 2 as the fair value of these contracts is determined based on the difference between the fixed interest rate in the agreements and the observable LIBOR-based interest forward rates at period end, multiplied by the total notional value.

The Company has no other financial assets that are measured at fair value on a nonrecurring basis at December 31, 2015.

Fair Value of Other Financial Instruments

The Company used the following methods and assumptions to determine the fair value of financial instruments that are not recognized at fair value as described below.

Cash and Cash Equivalents: Carried at amortized costs which approximate fair value.

Debt: The carrying amounts of the Company's variable-rate debt approximate fair values. For fixed-rate debt, the Company uses the income approach to determine the estimated fair value, by discounting cash flows using borrowing rates for comparable debt over the weighted life of the outstanding debt. The estimated fair value of the fixed-rate debt is Level 3 as certain inputs used are unobservable.

Fixed-rate debt that is not carried at fair value on the consolidated balance sheet and the estimated fair value of long-term fixed-rate debt (in millions):

	2015	2014
Carrying Amount	\$ 520	\$ 614
Fair value	557	666

NOTE 5. LONG-TERM DEBT

Long-term debt obligations (in millions):

	2015		2014	
Fixed-rate notes payable due through 2024	\$	520	\$	614
Variable-rate notes payable due through 2025		166		189
Long-term debt		686		803
Less current portion		115		117
	\$	571	\$	686
Weighted-average fixed-interest rate		5.7%		5.7%
Weighted-average variable-interest rate		1.8%		1.6%

All of the Company's borrowings are secured by aircraft.

During 2015, the Company issued no new debt and made scheduled debt payments of \$116 million. As of December 31, 2015, none of the Company's borrowings were restricted by financial covenants.

Long-term debt principal payments for the next five years and thereafter (in millions):

	Total
2016	\$ 115
2017	121
2018	151
2019	114
2020	116
Thereafter	69
Total principal payments	\$ 686

Bank Line of Credit

The Company has two \$100 million credit facilities and one \$52 million credit facility. All three facilities have variable interest rates based on LIBOR plus a specified margin. One of the \$100 million facilities, which expires in September 2017, is secured by aircraft. The other \$100 million facility, which expires in March 2017, is secured by certain accounts receivable, spare engines, spare parts and ground service equipment. The \$52 million facility expires in October 2016 with a mechanism for annual renewal and is secured by two 737-800 aircraft. The Company has no immediate plans to borrow using any of these facilities. All three credit facilities have a requirement to maintain a minimum unrestricted cash and marketable securities balance of \$500 million. The Company was in compliance with this covenant at December 31, 2015.

NOTE 6. INCOME TAXES

Deferred Income Taxes

Deferred income taxes reflect the impact of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and such amounts for tax purposes. Primarily due to differences in depreciation rates for federal income tax purposes and for financial reporting purposes, the Company has generated a net deferred tax liability.



Deferred tax (assets) and liabilities comprise the following (in millions):

	2015	2014
Excess of tax over book depreciation	\$ 1,110	\$ 1,042
Other—net	23	22
Gross deferred tax liabilities	1,133	1,064
Mileage Plan	(208)	(206)
Inventory obsolescence	(22)	(20)
Deferred gains	(8)	(10)
Employee benefits	(167)	(166)
Fuel hedge contracts	(5)	(5)
Other—net	(41)	(24)
Gross deferred tax assets	(451)	(431)
Net deferred tax liabilities	\$ 682	\$ 633

The Company has concluded that it is more likely than not that its deferred tax assets will be realizable and thus no valuation allowance has been recorded as of December 31, 2015. This conclusion is based on the expected future reversals of existing taxable temporary differences, anticipated future taxable income, and the potential for future tax planning strategies to generate taxable income, if needed. The Company will continue to reassess the need for a valuation allowance during each future reporting period.

Components of Income Tax Expense

The components of income tax expense were as follows (in millions):

		2015	2014	2013
Current tax expense:				
Federal	:	\$ 397	\$ 229	\$ 145
State		30	27	17
Total current		427	256	162
	-			
Deferred tax expense:				
Federal		60	103	131
State		(23)	11	15
Total deferred		37	114	146
Income tax expense		\$ 464	\$ 370	\$ 308
	=			



Income Tax Rate Reconciliation

Income tax expense reconciles to the amount computed by applying the U.S. federal rate of 35% to income before income tax and accounting change as follows (in millions):

	_	2015	2014	2013
Income before income tax	\$	1,312	\$ 975	\$ 816
Expected tax expense		459	341	286
Nondeductible expenses		4	4	4
State income taxes		19	25	21
State income sourcing		(15)	—	—
Other—net		(3)		(3)
Actual tax expense	\$	464	\$ 370	\$ 308
Effective tax rate		35.4%	37.9%	37.7%

In the fourth quarter of 2015, the Company filed amended state tax returns for the years 2010 through 2013 to update the Company's position on income sourcing in various states. These positions were also taken on 2014 filings and will be taken going forward. The cumulative benefit of this change is \$26 million, net of federal impact and unrecognized tax benefits of \$18 million.

Uncertain Tax Positions

The Company has identified its federal tax return and its state tax returns in Alaska, Oregon, and California as "major" tax jurisdictions. A summary of the Company's jurisdictions and the periods that are subject to examination are as follows:

Jurisdiction	Period
Federal	2012 to 2014
Alaska	2012 to 2014
California	2010 to 2014
Oregon	2003 to 2014*

*The 2003, 2004, 2008-2010 and 2011 Oregon tax returns are subject to examination only to the extent of net operating loss carryforwards from those years that were utilized in 2010 and later years.

Changes in the liability for unrecognized tax benefits during 2015, 2014, and 2013 are as follows (in millions):

	2	015	2014	2013
Balance at January 1,	\$	3 \$	2	\$ 1
Additions based on tax positions and settlements related to the current year		19	1	1
Balance at December 31,	\$	22 \$	3	\$ 2

At December 31, 2015, the total amount of unrecognized tax benefits is recorded as a liability, all of which impacts the effective tax rate. The Company added \$19 million of reserves for uncertain tax positions in 2015, primarily due to changes in income sourcing for state income taxes. These uncertain tax positions could change as a result of the Company's ongoing audits, settlement of issues, new audits and status of other taxpayer court cases and we cannot predict the timing of these actions. Due to the positions being taken in various jurisdictions, at the current time, an estimate of the range of reasonably possible outcomes cannot be made, beyond amounts currently accrued. No interest or penalties related to these tax positions were accrued as of December 31, 2015.

NOTE 7. EMPLOYEE BENEFIT PLANS

Four defined-benefit and five defined-contribution retirement plans cover various employee groups of Alaska and Horizon. The defined-benefit plans provide benefits based on an employee's term of service and average compensation for a specified period of time before retirement. The qualified defined-benefit pension plans are closed to new entrants.

Accounting standards require recognition of the overfunded or underfunded status of an entity's defined-benefit pension and other postretirement plan as an asset or liability in the consolidated financial statements and requires recognition of the funded status in AOCL.

Qualified Defined-Benefit Pension Plans

The Company's pension plans are funded as required by the Employee Retirement Income Security Act of 1974 (ERISA). The defined-benefit plan assets consist primarily of marketable equity and fixed-income securities. The Company uses a December 31 measurement date for these plans.

Weighted average assumptions used to determine benefit obligations:

	2015	2014
Discount rates ^(a)	4.55% to 4.69%	4.20%
Rate of compensation increases ^(a)	2.06% to 2.65%	2.85% to 3.91%

^(a) Varies by plan and related work group.

Weighted average assumptions used to determine net periodic benefit cost:

	2015	2014	2013
Discount rate	4.20%	4.85%	3.95%
Expected return on plan assets	6.50%	6.75%	7.25%
Rate of compensation increases ^(a)	2.85% to 3.91%	2.90% to 3.93%	3.05% to 4.02%

^(a) Varies by plan and related work group.

The discount rate was determined using current rates earned on high-quality, long-term bonds with maturities that correspond with the estimated cash distributions from the pension plans. At December 31, 2015, the Company selected discount rates for each of the plans using a pool of higher-yielding bonds estimated to be more reflective of settlement rates, as management has taken steps to ultimately terminate or settle plans that are frozen and move toward freezing benefits in active plans in the future. In determining the expected return on plan assets, the Company assesses the current level of expected returns on risk-free investments (primarily government bonds), the historical level of the risk premium associated with the other asset classes in which the portfolio is invested and the expectations for future returns of each asset class. The expected return for each asset class is then weighted based on the target asset allocation to develop the expected long-term rate of return on assets assumption for the portfolio.

Plan assets are invested in common commingled trust funds invested in equity and fixed income securities and in certain real estate assets. The asset allocation of the funds in the qualified defined-benefit plans, by asset category, is as follows:

	2015	2014
Asset category:		
Domestic equity securities	28%	33%
Non-U.S. equity securities	12%	14%
Fixed income securities	55%	53%
Real estate	5%	%
Plan assets	100%	100%

The Company's investment policy focuses on achieving maximum returns at a reasonable risk for pension assets over a full market cycle. In 2015, the Company separated the management of plan assets for the defined-benefit plan that covers the Company's non-union, management participants. This plan has been closed to new participants for several years and benefits were frozen effective January 1, 2014. These assets have a higher allocation to fixed income securities than the other plans. The Company uses a fund manager and invests in various asset classes to diversify risk.

Target allocations for the primary asset classes based on current funded status are approximately:

Domestic equities:	21% - 33%
Non-U.S. equities:	7% - 17%
Fixed income:	48% - 67%
Real estate:	0% - 8%

The Company determines the strategic allocation between equities, fixed income, and real estate based on current funded status and other characteristics of the plans. As the funded status improves, the Company increases the fixed income allocation of the portfolio, and decreases the equity allocation. Actual asset allocations are reviewed regularly and periodically rebalanced as appropriate.

As of December 31, 2015, all assets other than real estate were invested in common commingled trust funds. The Company uses the net asset values of these funds to determine fair value as allowed using the practical expediency method outlined in the accounting standards. Fair value estimates for real estate are calculated using the present value of expected future cash flows based on independent appraisals, local market conditions, and current and projected operating performance.

Plan asset by fund category (in millions):

2015 2014		2014	Fair Value Hierarchy
491		634	(a)
208		272	(a)
953		190	(a)
_		821	(a)
\$ 1,652	\$	1,917	
85		_	Level 3
\$ 1,737	\$	1,917	
<u>\$</u> <u>\$</u>	491 208 953 	491 208 953	491 634 208 272 953 190 — 821 \$ 1,652 \$ 1,917 85 —

^(a) In accordance with Subtopic 820-10, certain investments that are measured at fair value using the net asset value per share practical expedient have not been classified in the fair value hierarchy.

Changes in our Level 3 plan assets for the year ended December 31, 2015 included:

Asset Category	December 31, 2014 Balance	Net Realized and Unrealized Gains/Losses	Net Purchases, Issuances and Settlements	Net Transfers Into/(Out of) Level 3	December 31, 2015 Balance
Real Estate	\$ _	5	80	_	\$ 85

The following table sets forth the status of the qualified defined-benefit pension plans (in millions):

	2015		2014	
Projected benefit obligation (PBO)				
Beginning of year	\$ 2,050	\$	1,709	
Service cost	41		33	
Interest cost	84		81	
Plan settlement	(62)			
Actuarial (gain) loss	(140)		298	
Benefits paid	(75)		(71)	
End of year	\$ 1,898	\$	2,050	
Plan assets at fair value				
Beginning of year	\$ 1,917	\$	1,769	
Actual return on plan assets	(43)		219	
Employer contributions	—		—	
Plan settlement	(62)		—	
Benefits paid	(75)		(71)	
End of year	\$ 1,737	\$	1,917	
Funded status (unfunded)	\$ (161)	\$	(133)	
Percent funded	92%		94%	

The accumulated benefit obligation for the combined qualified defined-benefit pension was \$1.8 billion and \$1.9 billion at December 31, 2015, and 2014, respectively.

The amounts recognized in the consolidated balance sheets (in millions):

	201	5	2014
Accrued benefit liability-long term	\$	173	\$ 133
Plan assets-long term (within noncurrent Other Assets)		(12)	—
Total liability recognized	\$	161	\$ 133

The amounts not yet reflected in net periodic benefit cost and included in AOCL:

	2015		2014
Prior service credit	\$	(11)	\$ (12)
Net loss		499	514
Amount recognized in AOCL (pretax)	\$	488	\$ 502

The expected amortization of prior service credit and net loss from AOCL in 2016 is \$(1) million and \$25 million, respectively, for the qualified definedbenefit pension plans. Net pension expense for the qualified defined-benefit plans included the following components (in millions):

	20	15	2014	2013
Service cost	\$	41	\$ 33	\$ 46
Interest cost		84	81	73
Expected return on assets		(122)	(117)	(111)
Amortization of prior service cost		(1)	(1)	(1)
Recognized actuarial loss		26	13	43
Settlement expense (special item)		14		—
Net pension expense	\$	42	\$ 9	\$ 50

In 2015, the Company recognized a settlement charge of \$14 million related to lump sum settlements offered to terminated, vested plan participants. The result was a reduction in the projected benefit obligation of \$62 million. The settlement charge reflects the remaining unamortized actuarial loss in AOCL associated with the settled obligation.

There are no current statutory funding requirements for the Company's plans in 2016, nor does the Company expect to contribute to the qualified definedbenefit pension plans during 2016.

Future benefits expected to be paid over the next ten years under the qualified defined-benefit pension plans from the assets of those plans (in millions):

2016	\$ 77
2017	86
2018	93
2019	96
2020	107
2021 - 2024	613

Nonqualified Defined-Benefit Pension Plan

Alaska also maintains an unfunded, noncontributory defined-benefit plan for certain elected officers. This plan uses a December 31 measurement date. The assumptions used to determine benefit obligations and the net period benefit cost for the nonqualified defined-benefit pension plan are similar to those used to calculate the qualified defined-benefit pension plan. The plan's unfunded status, PBO, accumulated benefit obligation is immaterial. The net pension expense in prior year and expected future expense is also immaterial.

Postretirement Medical Benefits

The Company allows certain retirees to continue their medical, dental, and vision benefits by paying all or a portion of the active employee plan premium until eligible for Medicare, currently age 65. This results in a subsidy to retirees, because the premiums received by the Company are less than the actual cost of the retirees' claims. The accumulated postretirement benefit obligation (APBO) for this subsidy is unfunded. The accumulated postretirement benefit obligation was \$64 million and \$81 million at December 31, 2015 and 2014, respectively. The net periodic benefit cost was not material to the statement of operations.

During 2014, the Company made changes to the postretirement medical benefits for non-union personnel and certain labor groups to sunset the postretirement medical benefits effective in 2015. As a result of these changes, the Company recognized a partial curtailment gain of \$25 million in 2014. The curtailment gain included \$5 million associated with an embedded sick leave subsidy. This subsidy was used to establish a new compensated absence liability. The net impact of the curtailment gain of \$20 million is included in special items in the income statement.

Defined-Contribution Plans

The defined-contribution plans are deferred compensation plans under section 401(k) of the Internal Revenue Code. All of these plans require Company contributions. Total expense for the defined-contribution plans was \$60 million, \$54 million, and \$44 million in 2015, 2014, and 2013, respectively.

The Company also has a noncontributory, unfunded defined-contribution plan for certain elected officers of the Company who are ineligible for the nonqualified defined-benefit pension plan. Amounts recorded as liabilities under the plan are not material to the consolidated balance sheet at December 31, 2015 and 2014.

Pilot Long-term Disability Benefits

Alaska maintains a long-term disability plan for its pilots. The long-term disability plan does not have a service requirement. Therefore, the liability is calculated based on estimated future benefit payments associated with pilots that were assumed to be disabled on a long-term basis as of December 31, 2015 and does not include any assumptions for future disability. The liability includes the discounted expected future benefit payments and medical costs. The total liability was \$19 million and \$16 million, which was recorded net of a prefunded trust account of \$2 million and \$2 million, and included in long-term other liabilities on the consolidated balance sheets as of December 31, 2015 and December 31, 2014, respectively.

Employee Incentive-Pay Plans

Alaska and Horizon have employee incentive plans that pay employees based on certain financial and operational metrics. These metrics are set and approved annually by the Compensation Committee of the Board of Directors. The aggregate expense under these plans in 2015, 2014 and 2013 was \$120 million, \$116 million, and \$105 million, respectively. The plans are summarized below:

- *Performance-Based Pay* (PBP) is a program that rewards all employees. The program is based on four separate metrics related to Air Group profitability, safety, achievement of unit-cost goals, and employee engagement as measured by customer satisfaction.
- The Operational Performance Rewards Program entitles all Air Group employees to quarterly payouts of up to \$300 per person if certain operational and customer service objectives are met.

NOTE 8. COMMITMENTS AND CONTINGENCIES

Future minimum fixed payments for commitments as of December 31, 2015 (in millions):

	Ai	rcraft Leases	Facility Leases	Airc	craft Commitments	(Capacity Purchase Agreements
2016	\$	113	\$ 92	\$	505	\$	67
2017		104	88		549		58
2018		98	41		444		60
2019		90	41		390		64
2020		81	38		327		68
Thereafter		467	142		418		554
Total	\$	953	\$ 442	\$	2,633	\$	871

Lease Commitments

At December 31, 2015, the Company had lease contracts for 27 B737 aircraft, 15 Q400 aircraft, 6 CRJ-700 aircraft (operated under the CPA with SkyWest) and 8 CRJ-700 aircraft that are subleased and operated by another carrier (i.e. not in the Company's fleet). The Company has 15 E175 aircraft lease commitments under the CPA with SkyWest, 5 of which are included in the fleet as of December 31, 2015. All lease contracts have remaining noncancelable lease terms ranging from 2016 to 2028. The Company has the option to increase capacity flown by SkyWest with eight additional E175 aircraft with 2017 and 2018 delivery dates.

In addition to the above lease commitments, the Company has contracted for eight E175 regional aircraft that will enter service in 2017.



The majority of airport and terminal facilities are leased. Total rent expense for aircraft and facility leases was \$295 million, \$288 million, and \$290 million, in 2015, 2014, and 2013, respectively.

Aircraft Commitments

As of December 31, 2015, the Company is committed to purchasing 68 B737 aircraft (31 B737-900ER aircraft and 37 B737 MAX aircraft, with deliveries in 2016 through 2022) and two Q400 aircraft, with deliveries in 2018. In addition, the Company has options to purchase 46 B737 aircraft and 5 Q400 aircraft.

Capacity Purchase Agreements (CPAs)

At December 31, 2015, Alaska had CPAs with three carriers, including the Company's wholly-owned subsidiary, Horizon. Horizon sells 100% of its capacity under a CPA with Alaska. In addition, Alaska has a CPA with PenAir to fly certain routes in the state of Alaska. Under these agreements, Alaska pays the carriers an amount which is based on a determination of their cost of operating those flights and other factors intended to approximate market rates for those services. Future payments (excluding Horizon) are based on minimum levels of flying by the third-party carriers, which could differ materially due to variable payments based on actual levels of flying and certain costs associated with operating flights such as fuel.

During the second quarter Alaska signed an amendment to the CPA with SkyWest to remove the eight CRJ-700 aircraft out of regional operations and replace them with eight E175 aircraft. Six of these CRJ-700 aircraft are leased by the Company and two of the aircraft are owned by the Company. The E175 aircraft will be introduced into service throughout 2016, at which time the CRJ-700 aircraft will be removed from service. The CPA with SkyWest is a service contract that, in accordance with GAAP, includes embedded leases related to the aircraft operated under the agreement.

Contingencies

The Company is a party to routine litigation matters incidental to its business and with respect to which no material liability is expected. Management believes the ultimate disposition of these matters is not likely to materially affect the Company's financial position or results of operations. This forward-looking statement is based on management's current understanding of the relevant law and facts, and it is subject to various contingencies, including the potential costs and risks associated with litigation and the actions of arbitrators, judges and juries.

NOTE 9. SHAREHOLDERS' EQUITY

Common Stock Changes

During the second quarter of 2014, shareholders voted to increase the number of authorized shares from 100 million to 200 million shares, and reduce the par value of common stock from \$1 per share to \$0.01 per share, and the Board of Directors declared a two-for-one stock split by means of a stock distribution. The additional shares were distributed on July 9, 2014, to the shareholders of record on June 23, 2014.

Dividends

During 2015, the Board of Directors declared dividends of \$0.80 per share. The Company paid dividends of \$102 million to shareholders of record during 2015.

Subsequent to year-end, the Board of Directors declared a quarterly cash dividend of \$0.275 per share to be paid on March 8, 2016 to shareholders of record as of February 23, 2016. This is a 38% increase from the most recent quarterly dividends of \$0.20 per share.

Common Stock Repurchase

In August 2015, the Board of Directors authorized a \$1 billion share repurchase program, which does not have a set expiration date. In May 2014, the Board of Directors authorized a \$650 million share repurchase program, which was completed in October 2015. In September 2012, the Board of Directors authorized a \$250 million share repurchase program, which was completed in July 2014. Share repurchase activity (in millions, except shares):

	2015			2		2013			
	Shares		Amount	Shares	Amount		Shares		Amount
2015 \$1 billion Repurchase Program	1,517,277	\$	120	_	\$	_		\$	—
2014 \$650 million Repurchase Program	5,691,051	\$	385	5,497,427	\$	265		\$	—
2012 \$250 million Repurchase Program	—		_	1,819,304		83	4,984,186		159
Total	7,208,328	\$	505	7,316,731	\$	348	4,984,186	\$	159

Retirement of Treasury Shares

In 2015, the Company retired 4,016,654 common shares that had been held in treasury. At December 31, 2015, the Company held 3,266,774 shares in treasury. Management does not anticipate retiring common shares held in treasury for the foreseeable future.

Accumulated Other Comprehensive Loss (AOCL)

AOCL consisted of the following (in millions, net of tax):

	2015	2014
Related to marketable securities	(3)	_
Related to employee benefit plans	(288)	(298)

Related to interest rate derivatives	(12)	(12)
	\$ (303)	\$ (310)

NOTE 10. SPECIAL ITEMS

Special Mileage Plan Revenue

In the third quarter of 2013, the Company modified its Affinity Card Agreement (Agreement) with Bank of America Corporation (BAC), through which the Company sells miles and other items to BAC and the Company's loyalty program members accrue frequent flyer miles based on purchases using credit cards issued by BAC. As a result of the execution of the Agreement, consideration received as part of this agreement is subject to Accounting Standards Update 2009-13, "Multiple-Deliverable Revenue Arrangements - a consensus of the FASB Emerging Issues Task Force" (ASU 2009-13).

The Company followed the rollforward transition approach of ASU 2009-13, which required that the Company's existing deferred revenue balance be adjusted to reflect the value, on a relative selling price basis, of any undelivered element remaining at the date of contract modification as if the Company had been applying ASU 2009-13 since inception of the Agreement. The relative selling price of the undelivered element (air transportation) is lower than the rate at which it had been deferred under the previous contract and the Company recorded a one-time, non-cash adjustment to decrease frequent flyer deferred revenue and increase Special mileage plan revenue. The amount recorded for the year ended December 31, 2013 was \$192 million.

Also during 2013, as part of the Company's ongoing evaluation of Mileage Plan program assumptions, the Company performed a statistical analysis of historical data, which refined its estimate of the amount of breakage in the mileage population. This new refinement enables the Company to better identify historical differences between certain of its mileage breakage estimates and the amounts that have actually been experienced. As a result, the Company increased its estimate of the number of frequent flyer miles expected to expire unused from 12% to 17.4%. Included in the Special mileage plan revenue item above is \$44 million of additional revenue related to the effect of the change on the deferred revenue balance.

Special Items

In 2015, the Company recognized special items of \$32 million in aggregate (\$20 million after tax, or \$0.15 per diluted share). The special items are comprised of the recognition of an expense of \$14 million due to lump sum settlements offered to terminated, vested participants in the qualified defined benefit pension plans, and a litigation-related matter. See Note 7. Employee Benefit Plans for more information regarding the pension settlement charge.

Refer to Note 7. Employee Benefit Plans for detailed information about the \$20 million benefit related to the curtailment of certain postretirement benefit plans recognized in 2014. Furthermore, in 2014, we recorded a one-time gain of \$10 million associated with the settlement of a legal matter.

NOTE 11. STOCK-BASED COMPENSATION PLANS

The table below summarizes the components of total stock-based compensation (in millions):

	2015	2014	2013
Stock options	\$ 2	\$ 3	\$ 3
Stock awards	11	10	10
Deferred stock awards	1	1	1
Employee stock purchase plan	3	2	2
Stock-based compensation	\$ 17	\$ 16	\$ 16
Tax benefit related to stock-based compensation	\$ 7	\$ 6	\$ 6

Unrecognized stock-based compensation for non-vested options and awards and the weighted-average period the expense will be recognized (in millions):

	Amount	Weighted- Average Period
Stock options	\$ 3	1.0
Stock awards	8	0.7
Unrecognized stock-based compensation	\$ 11	0.7

The Company has various equity incentive plans under which it may grant stock awards to directors, officers and employees. The Company also has an employee stock purchase plan (ESPP).

The Company is authorized to issue 36 million shares of common stock under these plans, of which 16,353,597 shares remain available for future grants of either options or stock awards as of December 31, 2015.

Stock Options

Stock options to purchase common stock are granted at the fair market value of the stock on the date of grant. The stock options granted have terms of up to ten years.

The fair value of each option grant was estimated on the date of grant using the Black-Scholes option-pricing model with the following weighted-average assumptions used for grants:

	201	5	2	014	2013
Expected volatility		53%		65%	 67%
Expected term	6	years		6 years	6 years
Risk-free interest rate		1.67%		1.87%	1.1%
Expected dividend yield		1.25%		1.25%	—
Weighted-average grant date fair value per share	\$	28.71	\$	21.70	\$ 14.74
Estimated fair value of options granted (millions)	\$	3	\$	3	\$ 3

The expected market price volatility is based on the historical volatility. The expected term is based on the estimated period of time until exercise based on historical experience. The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of the grant. The expected dividend yield is based on the estimated weighted average dividend yield over the expected term. The expected forfeiture rates are based on historical experience.

The tables below summarize stock option activity for the year ended December 31, 2015:

	Shares	Weighted- Average Exercise Price Per Share		Weighted- Average Contractual Life (Years)	Aggregate Intrinsic Value (in millions)
Outstanding, December 31, 2014	707,688	\$	21.57	6.4	\$ 27
Granted	93,660		65.30		
Exercised	(255,717)		15.97		
Canceled	(1)		38.76		
Forfeited or expired	(5,285)		44.56		
Outstanding, December 31, 2015	540,345	\$	31.58	6.3	\$ 26
Exercisable, December 31, 2015	195,873	\$	18.62	5.6	\$ 12
Vested or expected to vest, December 31, 2015	539,868	\$	31.57	6.3	\$ 26

(in millions)	2015		_	2014	 2013
Intrinsic value of option exercises	\$	14	\$	20	\$ 19
Cash received from stock option exercises		4		6	8
Tax benefit related to stock option exercises		5		7	7
Fair value of options vested		3		2	3

Stock Awards

Restricted Stock Units (RSUs) are awarded to eligible employees and entitle the grantee to receive shares of common stock at the end of the vest period. The fair value of the RSUs is based on the stock price on the date of grant. The RSUs "cliff vest" after three years, or the period from the date of grant to the employee's retirement eligibility, and expense is recognized accordingly. Performance Share Units (PSUs) are awarded to certain executives to receive shares of common stock if specific performance goals and market conditions are achieved. There are several tranches of PSUs which vest when performance goals and market conditions are met.

The following table summarizes information about outstanding stock awards:

	Number of Units	Weighted- Average Grant Date Fair Value		Weighted- Average Contractual Life (Years)	Aggregate Intrinsic Value (in millions)
Non-vested, December 31, 2014	1,027,390	\$	26.19	0.6	\$ 61
Granted	244,874		50.94		
Vested	(764,322)		26.33		
Forfeited	(37,227)		35.86		
Non-vested, December 31, 2015	470,715	\$	38.09	0.8	\$ 38

Deferred Stock Awards

Deferred Stock Units (DSUs) are awarded to members of its Board of Directors as part of their retainers. The underlying common shares are issued upon retirement from the Board, but require no future service period. As a result, the entire intrinsic value of the awards is expensed on the date of grant.

Employee Stock Purchase Plan (ESPP)

The ESPP allows employees to purchase common stock at 85% of the stock price on the first day of the offering period or the specified purchase date, whichever is lower. Employees may contribute up to 10% of their base earnings during the offering

period to purchase stock. Employees purchased 281,058, 298,283, and 171,227 shares in 2015, 2014, and 2013 under the ESPP.

NOTE 12. OPERATING SEGMENT INFORMATION

Air Group has two operating airlines - Alaska Airlines and Horizon Air. Each is a regulated airline with separate management teams. To manage the two operating airlines and the revenues and expenses associated with the CPAs, management views the business in three operating segments.

Alaska Mainline - The Boeing 737 part of Alaska's business.

Alaska Regional - Alaska's shorter distance network. In this segment, Alaska Regional records actual on board passenger revenue, less costs such as fuel, distribution costs, and payments made to Horizon, SkyWest and PenAir under CPAs. Additionally, Alaska Regional includes a small allocation of corporate overhead such as IT, finance and other administrative costs incurred by Alaska and on behalf of the regional operations.

Horizon - Horizon operates regional aircraft. All of Horizon's capacity is sold to Alaska under a CPA. Expenses included those typically borne by regional airlines such as crew costs, ownership costs, and maintenance costs.

Additionally, the following table reports "Air Group Adjusted," which is not a measure determined in accordance with GAAP. The Company's chief operating decision-makers and others in management use this measure to evaluate operational performance and determine resource allocations. Adjustments are further explained below in reconciling to consolidated GAAP results.

Operating segment information is as follows (in millions):

	Al	laska						
Year Ended December 31, 2015	Mainline	Regional	Horizon	Consolidati	ng	Air Group Adjusted ^(a)	Special Items ^(b)	Consolidated
Operating revenues								
Passenger								
Mainline	\$ 3,939	\$	\$	\$	—	\$ 3,939	\$	\$ 3,939
Regional	—	854	—		—	854	—	854
Total passenger revenues	3,939	854	_		_	4,793	_	4,793
CPA revenues	—	—	408	((408)	_	—	—
Freight and mail	103	5	—		—	108	—	108
Other-net	621	72	4		_	697	_	697
Total operating revenues	4,663	931	412	((408)	5,598	_	5,598
Operating expenses								
Operating expenses, excluding fuel	2,653	695	375	((409)	3,314	32	3,346
Economic fuel	823	131	_		_	954	_	954
Total operating expenses	3,476	826	375	((409)	4,268	32	4,300
Nonoperating income (expense)								
Interest income	19	_			2	21	_	21
Interest expense	(28)	_	(10)		(4)	(42)	_	(42)
Other	28	_	1		6	35	_	35
	19		(9)		4	14		14
Income (loss) before income tax	\$ 1,206	\$ 105	\$ 28	\$	5	\$ 1,344	\$ (32)	\$ 1,312

67

		laska	_				
Year Ended December 31, 2014	Mainline	Regional	Horizon	Consolidating	Air Group Adjusted ^(a)	Special Items ^(b)	Consolidated
Operating revenues							
Passenger							
Mainline	\$ 3,774	\$ —	\$	\$	\$ 3,774	\$ —	\$ 3,774
Regional	—	805	—	—	805	—	805
Total passenger revenues	3,774	805		_	4,579		4,579
CPA revenues	_	_	371	(371)	_		
Freight and mail	109	5	_		114	_	114
Other-net	592	78	5		675		675
Total operating revenues	4,475	888	376	(371)	5,368	_	5,368
Operating expenses							
Operating expenses, excluding fuel	2,417	623	349	(371)	3,018	(30)	2,988
Economic fuel	1,251	190	_	_	1,441	(23)	1,418
Total operating expenses	3,668	813	349	(371)	4,459	(53)	4,406
Nonoperating income (expense)							
Interest income	20	_	_	1	21	_	21
Interest expense	(32)	_	(12)	(4)	(48)		(48)
Other	39	(1)	2	_	40	—	40
	27	(1)	(10)	(3)	13		13
Income (loss) before income tax	\$ 834	\$ 74	\$ 17	\$ (3)	\$ 922	\$ 53	\$ 975

		A	laska	ı										
Year Ended December 31, 2013	N	lainline	_	Regional		Horizon		Consolidating		ir Group djusted ^(a)		Special Items ^(b)	0	Consolidated
Operating revenues														
Passenger														
Mainline	\$	3,490	\$		\$	—	\$	—	\$	3,490	\$		\$	3,490
Regional				777		_		_		777		_		777
Total passenger revenues		3,490		777		—		—		4,267		_		4,267
CPA revenues						368		(368)		_		_		—
Freight and mail		109		4		_		—		113		_		113
Other-net		513		66		5		—		584		192		776
Total operating revenues		4,112		847		373		(368)		4,964		192		5,156
Operating expenses														
Operating expenses, excluding fuel		2,293		585		341		(368)		2,851				2,851
Economic fuel		1,294		181				_		1,475		(8)		1,467
Total operating expenses		3,587		766		341		(368)		4,326		(8)		4,318
Nonoperating income (expense)														
Interest income		18								18				18
				_		(14)		(4)				_		
Interest expense Other		(38) 25		(12)		. /		(4)		(56) 16		_		(56) 16
Oule				(12)		2								
	<u>_</u>	5	0	(12)	•	(12)	•	(3)	0	(22)	0	-	¢	(22)
Income (loss) before income tax	\$	530	\$	69	\$	20	\$	(3)	\$	616	\$	200	\$	816

(a) The adjusted column represents the financial information that is reviewed by management to assess performance of operations and determine capital allocations and does not include certain income and charges.

(b) Includes accounting adjustments related to mark-to-market fuel-hedge accounting charges (all years), pension settlement charge (2015), litigation-related matter (2015), non-cash curtailment gain (2014), a gain related to a legal matter (2014), and Special mileage plan revenue (2013).

	2015	2014	2013
Depreciation:			
Alaska ^(a)	\$ 268	243	\$ 223
Horizon	52	51	47
Parent company	_	—	—
Consolidated	\$ 320	\$ 294	\$ 270
Capital expenditures:			
Alaska ^(a)	\$ 821	\$ 659	\$ 494
Horizon	10	35	72
Consolidated	\$ 831	\$ 694	\$ 566
Total assets at end of period:			
Alaska ^(a)	\$ 8,129	\$ 6,665	
Horizon	718	809	
Parent company	4,734	3,551	
Elimination of inter-company accounts	(7,048)	(4,961)	
Consolidated	\$ 6,533	\$ 6,064	

(a) There are no depreciation expenses, capital expenditures or assets associated with purchased capacity flying at Alaska Regional.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None

ITEM 9A. CONTROLS AND PROCEDURES

EVALUATION OF DISCLOSURE CONTROLS AND PROCEDURES

The Company's management, with the participation of the Principal Executive Officer and Principal Financial Officer, conducted an evaluation of the effectiveness of the Company's disclosure controls and procedures (as defined in Exchange Act Rule 13a-15(e)) as of the end of the period covered by this report. Based on that evaluation, the Principal Executive Officer and Principal Financial Officer concluded that the Company's disclosure controls and procedures were effective as of the end of the period covered by this report.

CHANGES IN INTERNAL CONTROL OVER FINANCIAL REPORTING

There have been no changes in the Company's internal controls over financial reporting during the fourth quarter of 2015 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

MANAGEMENT'S REPORT ON INTERNAL CONTROL OVER FINANCIAL REPORTING

Our management is responsible for establishing and maintaining adequate internal control over financial reporting, as such term is defined in Exchange Act Rule 13a-15(f). Under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, we conducted an evaluation of the effectiveness of our internal control over financial reporting based on the 2013 framework in Internal Control – Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (the COSO Framework). Based on our evaluation, our management concluded that our internal control over financial reporting was effective as of December 31, 2015.

We intend to regularly review and evaluate the design and effectiveness of our disclosure controls and procedures and internal control over financial reporting on an ongoing basis and to improve these controls and procedures over time and to correct any deficiencies that we may discover in the future. While we believe the present design of our disclosure controls and procedures and internal control over financial reporting are effective, future events affecting our business may cause us to modify our controls and procedures.

The Company's independent registered public accounting firm has issued an attestation report regarding its assessment of the effectiveness of the Company's internal control over financial reporting as of December 31, 2015.

70

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The Board of Directors and Shareholders Alaska Air Group, Inc.:

We have audited Alaska Air Group, Inc.'s internal control over financial reporting as of December 31, 2015, based on criteria established in *Internal Control*-*Integrated Framework* (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). Alaska Air Group, Inc.'s management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying Management's Report on Internal Control over Financial Reporting (included in Item 9A). Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audit also included performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, Alaska Air Group, Inc. maintained, in all material respects, effective internal control over financial reporting as of December 31, 2015, based on criteria established in *Internal Control - Integrated Framework* (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheets of Alaska Air Group, Inc. and subsidiaries as of December 31, 2015 and 2014, and the related consolidated statements of operations, comprehensive operations, shareholders' equity, and cash flows for each of the years in the three-year period ended December 31, 2015, and our report dated February 11, 2016 expressed an unqualified opinion on those consolidated financial statements.

/s/ KPMG LLP

Seattle, Washington February 11, 2016

71

ITEM 9B. OTHER INFORMATION

None.

PART III

ITEM 10. DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

See "Executive Officers of the Registrant" under Item 1, "Our Business," in Part I of this Form 10-K for information on the executive officers of Air Group and its subsidiaries. Except as provided herein, the remainder of the information required by this item is incorporated herein by reference from the definitive Proxy Statement for Air Group's 2016 Annual Meeting of Stockholders to be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year ended December 31, 2015 (hereinafter referred to as our "2016 Proxy Statement").

ITEM 11. EXECUTIVE COMPENSATION

The information required by this item is incorporated herein by reference from our 2016 Proxy Statement.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT, AND RELATED STOCKHOLDER MATTERS

Securities Authorized for Issuance Under Equity Compensation Plans

Plan category	Number of securities to be issued upon exercise of outstanding options, warrants and rights (a)	Weighted-average exercise price of outstanding options, warrants and rights (b)	Number of securities remaining available for future issuance under equity compensation plans (excluding securities reflected in column (a)) (c)
Equity compensation plans approved by security holders	1,405,741(1)	\$31.58(2)	12,813,915(3)
Equity compensation plans not approved by security holders	_	Not applicable	_
Total	1,405,741	\$31.58	12,813,915

(1) Of these shares, 540,345 were subject to options then outstanding under the 2008 Plan, and 865,396 were subject to outstanding restricted, performance and deferred stock unit awards granted under the 2008 Plan. Outstanding performance awards are reflected in the table assuming that the target level of performance will be achieved.

⁽²⁾ This number does not reflect the 865,396 shares that were subject to outstanding stock unit awards granted under the 2008 Plan.

(3) Of the aggregate number of shares that remained available for future issuance, 6,299,700 shares were available under the 2008 Plan and 6,514,215 shares were available under the ESPP. Subject to certain express limits of the 2008 Plan, shares available for award purposes under the 2008 Plan generally may be used for any type of award authorized under that plan including options, stock appreciation rights, and other forms of awards granted or denominated in shares of our common stock including, without limitation, stock bonuses, restricted stock, restricted stock units and performance shares.

Other information required by this item is set forth under the heading "Beneficial Ownership of Securities" in our 2016 Proxy Statement and is incorporated by reference.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

The information required by this item is incorporated herein by reference from our 2016 Proxy Statement.

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

The information required by this item is incorporated herein by reference from our 2016 Proxy Statement.

PART IV

ITEM 15. EXHIBITS

The following documents are filed as part of this report:

1. Exhibits: See Exhibit Index.



SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

ALASKA AIR GROUP, INC.

By: /s/ BRADLEY D. TILDEN

Date: February 11, 2016

Bradley D. Tilden President and Chief Executive Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on February 11, 2016 on behalf of the registrant and in the capacities indicated.

73

/s/ BRADLEY D. TILDEN	Chairman, President, and Chief Executive Officer
Bradley D. Tilden	(Principal Executive Officer)
/s/ BRANDON S. PEDERSEN	Executive Vice President/Finance and Chief Financial Officer
Brandon S. Pedersen	(Principal Financial Officer)
/s/ CHRISTOPHER M. BERRY	Controller, Alaska Airlines Managing Director, Accounting (Principal
Christopher M. Berry	Accounting Officer)
/s/ PATRICIA M. BEDIENT	Director
Patricia M. Bedient	_
/s/ MARION C. BLAKEY	Director
Marion C. Blakey	
/s/ PHYLLIS J. CAMPBELL	Director
Phyllis J. Campbell	
/s/ DHIREN R. FONSECA	Director
Dhiren R. Fonseca	
/s/ JESSIE J. KNIGHT, JR.	Director
Jessie J. Knight, Jr.	
/s/ DENNIS F. MADSEN	Director
Dennis F. Madsen	
/s/ HELVI K. SANDVIK	Director
Helvi K. Sandvik	
/s/ KATHERINE J. SAVITT	Director
Katherine J. Savitt	
/s/ J. KENNETH THOMPSON	Director
J. Kenneth Thompson	
/s/ ERIC K. YEAMAN	Director
Eric K. Yeaman	_

EXHIBIT INDEX

Certain of the following exhibits have been filed with the Securities and Exchange Commission and are incorporated by reference from the documents below. Certain others are filed with this Form 10-K. The exhibits are numbered in accordance with Item 601 of Regulation S-K.

Exhibit Number	Exhibit Description	Form	Date of First Filing	Exhibit Number	File Number
3.1	Amended and Restated Certificate of Incorporation of Registrant	10-Q	August 6, 2014	3.1	
3.2	Bylaws of Registrant, as amended December 9, 2015	8-K	December 15, 2015	3.2	
10.1#	Aircraft General Terms Agreement, dated June 15, 2005, between the Boeing Company and Alaska Airlines, Inc.	10-Q	August 5, 2005	10.1	
10.2#	Purchase Agreement No. 2497, dated June 15, 2005, between the Boeing Company and Alaska Airlines, Inc.	10-Q	August 5, 2005	10.2	
10.3#	Supplemental Agreement No. 23 to Purchase Agreement No. 2497 between The Boeing Company and Alaska Airlines, Inc.	10-Q/A	August 2, 2011	10.1	
10.4#	Supplemental Agreement No. 29 to Purchase Agreement No. 2497 between The Boeing Company and Alaska Airlines, Inc.	10-K	February 14, 2013	10.1	

10.5#	Purchase Agreement No. 3866 between The Boeing Company and Alaska Airlines, Inc.	10-K	February 14, 2013	10.2
10.6#	Supplemental Agreement No. 39 to Purchase Agreement No. 2497 between The Boeing Company and Alaska Airlines, Inc.	10-Q	May 7, 2015	10.1
10.7*	Alaska Air Group, Inc. 2008 Performance Incentive Plan, Form of Nonqualified Stock Option Agreement	10-Q	August 4, 2011	10.3
10.8*	Alaska Air Group, Inc. 2008 Performance Incentive Plan, Form of Performance Stock Unit Award Agreement	10-Q	August 4, 2011	10.4
10.9*	Alaska Air Group, Inc. 2008 Performance Incentive Plan, Form of Stock Unit Award Agreement	10-Q	August 4, 2011	10.5
10.10*†	Alaska Air Group, Inc. 2008 Performance Incentive Plan, Amended for Stock-Split	10-K	February 11, 2016	
10.11*†	Alaska Air Group, Inc. 2010 Employee Stock Purchase Plan, Amended for Stock-Split	10-K	February 11, 2016	
10.12*†	Alaska Air Group, Inc. Stock Deferral Plan for Non-Employee Directors	10-K	February 11, 2016	
10.13*	Alaska Air Group, Inc. Nonqualified Deferred Compensation Plan, as amended	10-Q	August 4, 2011	10.1
10.14*	1995 Elected Officers Supplementary Retirement Plan, as amended	10-Q	August 4, 2011	10.2
10.15*†	Form of Alaska Air Group, Inc. Change of Control Agreement for named executive officers, as amended and restated October 16, 2014	10-K	February 11, 2016	
21†	Subsidiaries of Registrant			
23.1†	Consent of Independent Registered Public Accounting Firm (KPMG LLP)			
31.1†	Certification of Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002			
31.2†	Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002			
32.1†	Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002			
32.2†	Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002			
101.INS†	XBRL Instance Document			
101.SCH†	XBRL Taxonomy Extension Schema Document			
101.CAL†	XBRL Taxonomy Extension Calculation Linkbase Document			
101.DEF†	XBRL Taxonomy Extension Definition Linkbase Document			
101.LAB†	XBRL Taxonomy Extension Label Linkbase Document			
101.PRE†	XBRL Taxonomy Extension Presentation Linkbase Document			
t	Filed herewith			
*	Indicates management contract or compensatory plan or arrangement.			

Pursuant to 17 CFR 240.24b-2, confidential information has been omitted and filed separately with the Securities and Exchange Commission pursuant to a Confidential Treatment Application filed with the Commission.

74

EXHIBIT 21

SUBSIDIARIES OF ALASKA AIR GROUP, INC.

Name	State of Incorporation
Alaska Airlines, Inc.	Alaska
Horizon Air Industries, Inc.	Washington
Air Group Leasing, Inc.	Delaware

EXHIBIT 23.1

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The Board of Directors Alaska Air Group, Inc.:

We consent to the incorporation by reference in the registration statements (Nos. 333-176051, 333-168293, and 333-151743) on Form S-8 of Alaska Air Group, Inc. of our reports dated February 11, 2016, with respect to the consolidated balance sheets of Alaska Air Group, Inc. as of December 31, 2015 and 2014, and the related consolidated statements of operations, comprehensive operations, shareholders' equity, and cash flows, for each of the years in the three-year period ended December 31, 2015, and the effectiveness of internal control over financial reporting as of December 31, 2015, which reports appear in the December 31, 2015 annual report on Form 10-K of Alaska Air Group, Inc.

Our report on the consolidated financial statements refers to a change in the method of accounting for consideration received under an affinity card agreement upon the adoption of Accounting Standards Update No. 2009-13, Multiple Deliverable Revenue Arrangements, in 2013.

/s/ KPMG LLP

Seattle, Washington February 11, 2016

EXHIBIT 31.1

CERTIFICATIONS

I, Bradley D. Tilden, certify that:

- 1. I have reviewed this annual report on Form 10-K of Alaska Air Group, Inc. for the period ended December 31, 2015;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's fourth fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- e) The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 11, 2016

By: /s/ BRADLEY D. TILDEN

Bradley D. Tilden President and Chief Executive Officer

EXHIBIT 31.2

CERTIFICATIONS

I, Brandon S. Pedersen, certify that:

- 1. I have reviewed this annual report on Form 10-K of Alaska Air Group, Inc. for the period ended December 31, 2015;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's fourth fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 11, 2016

By: /s/ BRANDON S. PEDERSEN

Brandon S. Pedersen Chief Financial Officer

EXHIBIT 32.1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Alaska Air Group, Inc. (the "Company") on Form 10-K for the period ended December 31, 2015 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Bradley D. Tilden, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

February 11, 2016

By: /s/ BRADLEY D. TILDEN

Bradley D. Tilden President and Chief Executive Officer

EXHIBIT 32.2

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Alaska Air Group, Inc. (the "Company") on Form 10-K for the period ended December 31, 2015 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Brandon S. Pedersen, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

February 11, 2016

By: /s/ BRANDON S. PEDERSEN

Brandon S. Pedersen Chief Financial Officer

ALASKA AIR GROUP, INC. 2008 PERFORMANCE INCENTIVE PLAN

Adopted by the Alaska Air Group, Inc. Board of Directors on March 13, 2008 Approved by Alaska Air Group, Inc. Shareholders on May 20, 2008 Amended by the Alaska Air Group, Inc. Board of Directors on February 9, 2011 Amended Plan Approved by Alaska Air Group, Inc. Shareholders on May 17, 2011

1. PURPOSE OF PLAN

The purpose of this Alaska Air Group, Inc. 2008 Performance Incentive Plan (this "**Plan**") of Alaska Air Group, Inc., a Delaware corporation (the "**Corporation**"), is to promote the success of the Corporation and to increase stockholder value by providing an additional means through the grant of awards to attract, motivate, retain and reward selected employees and other eligible persons.

2. ELIGIBILITY

The Administrator (as such term is defined in Section 3.1) may grant awards under this Plan only to those persons that the Administrator determines to be Eligible Persons. An "**Eligible Person**" is any person who is either: (a) an officer (whether or not a director) or employee of the Corporation or one of its Subsidiaries; (b) a director of the Corporation or one of its Subsidiaries; or (c) an individual consultant or advisor who renders or has rendered bona fide services (other than services in connection with the offering or sale of securities of the Corporation or one of its Subsidiaries) to the Corporation or one of its Subsidiaries and who is selected to participate in this Plan by the Administrator; provided, however, that a person who is otherwise an Eligible Person under clause (c) above may participate in this Plan only if such participation would not adversely affect either the Corporation's eligibility to use Form S-8 to register under the Securities Act of 1933, as amended (the "**Securities Act**"), the offering and sale of shares issuable under this Plan by the Corporation or the Corporation's compliance with any other applicable laws. An Eligible Person who has been granted an award (a "participant") may, if otherwise eligible, be granted additional awards if the Administrator shall so determine. As used herein, "**Subsidiary**" means any corporation or other entity a majority of whose outstanding voting stock or voting power is beneficially owned directly or indirectly by the Corporation; and "**Board**" means the Board of Directors of the Corporation.

3. PLAN ADMINISTRATION

3.1 The Administrator. This Plan shall be administered by and all awards under this Plan shall be authorized by the Administrator. The "Administrator" means the Board or one or more committees appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may

also delegate, to the extent permitted by Section 157(c) of the Delaware General Corporation Law and any other applicable law, to one or more officers of the Corporation, its powers under this Plan (a) to designate the officers and employees of the Corporation and its Subsidiaries who will receive grants of awards under this Plan, and (b) to determine the number of shares subject to, and the other terms and conditions of, such awards. The Board may delegate different levels of authority to different committees with administrative and grant authority under this Plan. Unless otherwise provided in the Bylaws of the Corporation or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute action by the acting Administrator.

With respect to awards intended to satisfy the requirements for performance-based compensation under Section 162(m) of the Internal Revenue Code of 1986, as amended (the "**Code**"), this Plan shall be administered by a committee consisting solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code); provided, however, that the failure to satisfy such requirement shall not affect the validity of the action of any committee otherwise duly authorized and acting in the matter. Award grants, and transactions in or involving awards, intended to be exempt under Rule 16b-3 under the Securities Exchange Act of 1934, as amended (the "**Exchange Act**"), must be duly and timely authorized by the Board or a committee consisting solely of two or more non-employee directors (as this requirement is applied under Rule 16b-3 promulgated under the Exchange Act). To the extent required by any applicable listing agency, this Plan shall be administered by a committee composed entirely of independent directors (within the meaning of the applicable listing agency).

3.2 *Powers of the Administrator*. Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of awards and the administration of this Plan (in the case of a committee or delegation to one or more officers, within the authority delegated to that committee or person(s)), including, without limitation, the authority to:

- (a) determine eligibility and, from among those persons determined to be eligible, the particular Eligible Persons who will receive an award under this Plan;
- (b) grant awards to Eligible Persons, determine the price at which securities will be offered or awarded and the number of securities to be offered or awarded to any of such persons, determine the other specific terms and conditions of such awards consistent with the express limits of this Plan, establish the installments (if any) in which such awards shall become exercisable or shall vest (which may include, without limitation, performance and/or time-based schedules), or determine that no delayed exercisability or vesting is required, establish any applicable performance targets, and establish the events of termination or reversion of such awards;
- (c) approve the forms of award agreements (which need not be identical either as to type of award or among participants);
- (d) construe and interpret this Plan and any agreements defining the rights and obligations of the Corporation, its Subsidiaries, and participants under this Plan,

further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the awards granted under this Plan;

- (e) cancel, modify, or waive the Corporation's rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding awards, subject to any required consent under Section 8.6.5;
- (f) accelerate or extend the vesting or exercisability or extend the term of any or all such outstanding awards (in the case of options or stock appreciation rights, within the maximum ten-year term of such awards) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a termination of employment or services or other events of a personal nature) subject to any required consent under Section 8.6.5;
- (g) adjust the number of shares of Common Stock subject to any award, adjust the price of any or all outstanding awards or otherwise change previously imposed terms and conditions, in such circumstances as the Administrator may deem appropriate, in each case subject to Sections 4 and 8.6 (and subject to the no repricing provision below);
- (h) determine the date of grant of an award, which may be a designated date after but not before the date of the Administrator's action (unless otherwise designated by the Administrator, the date of grant of an award shall be the date upon which the Administrator took the action granting an award);
- determine whether, and the extent to which, adjustments are required pursuant to Section 7 hereof and authorize the termination, conversion, substitution or succession of awards upon the occurrence of an event of the type described in Section 7;
- (j) acquire or settle (subject to Sections 7 and 8.6) rights under awards in cash, stock of equivalent value, or other consideration (subject to the no repricing provision below); and
- (k) determine the fair market value of the Common Stock or awards under this Plan from time to time and/or the manner in which such value will be determined.

Notwithstanding the foregoing and except for an adjustment pursuant to Section 7.1 or a repricing approved by stockholders, in no case may the Administrator (1) amend an outstanding stock option or SAR to reduce the exercise price or base price of the award, (2) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for cash or other awards for the purpose of repricing the award, or (3) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for an option or SAR with an exercise or base price that is less than the exercise or base price of the original award.

3.3 Binding Determinations. Any action taken by, or inaction of, the Corporation, any Subsidiary, or the Administrator relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board nor any

Board committee, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any award made under this Plan), and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.

3.4 Reliance on Experts. In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including employees and professional advisors to the Corporation. No director, officer or agent of the Corporation or any of its Subsidiaries shall be liable for any such action or determination taken or made or omitted in good faith.

3.5 *Delegation*. The Administrator may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Corporation or any of its Subsidiaries or to third parties.

4. SHARES OF COMMON STOCK SUBJECT TO THE PLAN; SHARE LIMITS

4.1 Shares Available. Subject to the provisions of Section 7.1, the capital stock that may be delivered under this Plan shall be shares of the Corporation's authorized but unissued Common Stock and any shares of its Common Stock held as treasury shares. For purposes of this Plan, "**Common Stock**" shall mean the common stock of the Corporation and such other securities or property as may become the subject of awards under this Plan, or may become subject to such awards, pursuant to an adjustment made under Section 7.1.

4.2 Share Limits. The maximum number of shares of Common Stock that may be delivered pursuant to awards granted to Eligible Persons under this Plan (the "Share Limit") is equal to the sum of the following:

- (1) 16,000,000 shares of Common Stock, plus
- (2) the number of shares of Common Stock available for additional award grant purposes under the Corporation's 2004 Long-Term Incentive Plan (the "2004 Plan") as of the date of stockholder approval of this Plan (the "Stockholder Approval Date") and determined immediately prior to the termination of the authority to grant new awards under the 2004 Plan as of the Stockholder Approval Date, plus
- (3) the number of any shares subject to stock options granted under the 2004 Plan and outstanding on the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date without being exercised (including any shares subject to stock options granted under the Corporation's 1999 Long-Term Equity Incentive Plan (the "1999 Plan") outstanding on the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date and would otherwise become eligible for award grant purposes under the 2004 Plan in accordance with the provisions of that plan);

(4) the number of any shares subject to restricted stock and restricted stock unit awards granted under the 2004 Plan that are outstanding and unvested on the Stockholder Approval Date that are forfeited, terminated, cancelled or otherwise reacquired by the Corporation without having become vested.

provided that in no event shall the Share Limit exceed 27,998,068 shares (which is the sum of the 16,000,000 shares set forth above, plus the number of shares available under the 2004 Plan for additional award grant purposes as of the Effective Date (as such term is defined in Section 8.6.1), plus the aggregate number of shares subject to awards previously granted and outstanding under the 2004 Plan and the 1999 Plan as of the Effective Date).

Shares issued in respect of any "Full-Value Award" granted under this Plan shall be counted against the foregoing Share Limit as 1.7 shares for every one share actually issued in connection with such award. (For example, if a stock bonus of 100 shares of Common Stock is granted under this Plan, 170 shares shall be charged against the Share Limit in connection with that award.) For this purpose, a "**Full-Value Award**" means any award under this Plan that is <u>not</u> a stock option grant or a stock appreciation right grant. The following limits also apply with respect to awards granted under this Plan:

- (a) The maximum number of shares of Common Stock that may be delivered pursuant to options qualified as incentive stock options granted under this Plan is 16,000,000 shares.
- (b) The maximum number of shares of Common Stock subject to those options and stock appreciation rights that are granted during any calendar year to any individual under this Plan is 1,200,000 shares.
- (c) Additional limits with respect to Performance-Based Awards are set forth in Section 5.2.3.

Each of the foregoing numerical limits is subject to adjustment as contemplated by Section 4.3, Section 7.1, and Section 8.10.

4.3 Awards Settled in Cash, Reissue of Awards and Shares. Except as provided in the next sentence, shares that are subject to or underlie awards granted under this Plan which expire or for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall again be available for subsequent awards under this Plan. Shares that are exchanged by a participant or withheld by the Corporation as full or partial payment in connection with any award under this Plan, as well as any shares exchanged by a participant or withheld by the Corporation or one of its Subsidiaries to satisfy the tax withholding obligations related to any award, shall not be available for subsequent awards under this Plan. To the extent that an award granted under this Plan is settled in cash or a form other than shares of Common Stock, the shares that would have been delivered had there been no such cash or other settlement shall not be counted against the shares available for issuance under this Plan. In the number of shares delivered with respect to the award shall be counted against the share limits of this Plan (including, for purposes of clarity, the limits of Section 4.2 of this Plan). (For purposes of clarity, if 1,000 dividend equivalent rights are granted and outstanding when the Corporation pays a dividend, and 50 shares are delivered in payment

of those rights with respect to that dividend, 85 shares (after giving effect to the Full-Value Award premium counting rules) shall be counted against the share limits of this Plan). To the extent that shares of Common Stock are delivered pursuant to the exercise of a stock appreciation right or stock option granted under this Plan, the number of underlying shares as to which the exercise related shall be counted against the applicable share limits under Section 4.2, as opposed to only counting the shares issued. (For purposes of clarity, if a stock appreciation right relates to 100,000 shares and is exercised at a time when the payment due to the participant is 15,000 shares, 100,000 shares shall be charged against the applicable share limits under Section 4.2 with respect to such exercise.) Refer to Section 8.10 for application of the foregoing share limits with respect to assumed awards. The foregoing adjustments to the share limits of this Plan are subject to any applicable limitations under Section 162(m) of the Code with respect to awards intended as performance-based compensation thereunder.

4.4 Reservation of Shares; No Fractional Shares; Minimum Issue. The Corporation shall at all times reserve a number of shares of Common Stock sufficient to cover the Corporation's obligations and contingent obligations to deliver shares with respect to awards then outstanding under this Plan (exclusive of any dividend equivalent obligations to the extent the Corporation has the right to settle such rights in cash). No fractional shares shall be delivered under this Plan. The Administrator may pay cash in lieu of any fractional shares in settlements of awards under this Plan. The Administrator may from time to time impose a limit (of not greater than 100 shares) on the minimum number of shares that may be purchased or exercised as to awards granted under this Plan unless (as to any particular award) the total number purchased or exercise is the total number at the time available for purchase or exercise under the award.

5. AWARDS

5.1 *Type and Form of Awards*. The Administrator shall determine the type or types of award(s) to be made to each selected Eligible Person. Awards may be granted singly, in combination or in tandem. Awards also may be made in combination or in tandem with, in replacement of, as alternatives to, or as the payment form for grants or rights under any other employee or compensation plan of the Corporation or one of its Subsidiaries. The types of awards that may be granted under this Plan are (subject, in each case, to the no repricing provisions of Section 3.2):

5.1.1 Stock Options. A stock option is the grant of a right to purchase a specified number of shares of Common Stock during a specified period as determined by the Administrator. An option may be intended as an incentive stock option within the meaning of Section 422 of the Code (an "**ISO**") or a nonqualified stock option (an option not intended to be an ISO). The award agreement for an option will indicate if the option is intended as an ISO; otherwise it will be deemed to be a nonqualified stock option. The maximum term of each option (ISO or nonqualified) shall be ten (10) years. The per share exercise price for each option shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the option. When an option is exercised, the exercise price for the shares to be purchased shall be paid in full in cash or such other method permitted by the Administrator consistent with Section 5.5.

5.1.2 Additional Rules Applicable to ISOs. To the extent that the aggregate fair market value (determined at the time of grant of the applicable option) of stock with respect to

which ISOs first become exercisable by a participant in any calendar year exceeds \$100,000, taking into account both Common Stock subject to ISOs under this Plan and stock subject to ISOs under all other plans of the Corporation or one of its Subsidiaries (or any parent or predecessor corporation to the extent required by and within the meaning of Section 422 of the Code and the regulations promulgated thereunder), such options shall be treated as nonqualified stock options. In reducing the number of options treated as ISOs to meet the \$100,000 limit, the most recently granted options shall be reduced first. To the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an ISO. ISOs may only be granted to employees of the Corporation or one of its subsidiaries (for this purpose, the term "subsidiary" is used as defined in Section 424(f) of the Code, which generally requires an unbroken chain of ownership of at least 50% of the total combined voting power of all classes of stock of each subsidiary in the chain beginning with the Corporation and ending with the subsidiary in question). There shall be imposed in any award agreement relating to ISOs such other terms and conditions as from time to time are required in order that the option be an "incentive stock option" as that term is defined in Section 422 of the Code. No ISO may be granted to any person who, at the time the option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding Common Stock possessing more than 10% of the total combined voting power of all classes of stock of the Corporation, unless the exercise price of such option is at least 110% of the fair market value of the stock subject to the option and such option by its terms is not exercisable after the expiration of five years from the date such option is granted.

5.1.3 Stock Appreciation Rights. A stock appreciation right or "SAR" is a right to receive a payment, in cash and/or Common Stock, equal to the excess of the fair market value of a specified number of shares of Common Stock on the date the SAR is exercised over the "base price" of the award, which base price shall be set forth in the applicable award agreement and shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the SAR. The maximum term of a SAR shall be ten (10) years.

5.1.4 Other Awards; Dividend Equivalent Rights. The other types of awards that may be granted under this Plan include: (a) stock bonuses, restricted stock, performance stock, stock units, phantom stock or similar rights to purchase or acquire shares, whether at a fixed or variable price or ratio related to the Common Stock, upon the passage of time, the occurrence of one or more events, or the satisfaction of performance criteria or other conditions, or any combination thereof; (b) any similar securities with a value derived from the value of or related to the Common Stock and/or returns thereon; or (c) cash awards. Dividend equivalent rights may be granted as a separate award or in connection with another award under this Plan; provided, however, that dividend equivalent rights may not be granted in connection with a stock option or SAR granted under this Plan. In addition, any dividends and/or dividend equivalents as to the unvested portion of a restricted stock award that is subject to performance-based vesting requirements or the unvested portion of a stock unit award that is subject to performance-based vesting requirements will be subject to termination and forfeiture to the same extent as the corresponding portion of the award to which they relate.

5.2 Section 162(m) Performance-Based Awards. Without limiting the generality of the foregoing, any of the types of awards listed in Section 5.1.4 above may be, and options and SARs granted to officers and employees ("Qualifying Options" and "Qualifying SARS," respectively) typically will be, granted as awards intended to satisfy the requirements for "performance-based compensation" within the meaning of Section 162(m) of the Code ("Performance-Based Awards"). The grant, vesting, exercisability or payment of Performance-Based Awards may depend (or, in the case of Qualifying Options or Qualifying SARs, may also depend) on the degree of achievement of one or more performance goals relative to a pre-established targeted level or levels using one or more of the Business Criteria set forth below (on an absolute basis or relative to the performance of other companies or upon comparisons of any of the indicators of performance relative to other companies) for the Corporation on a consolidated basis or for one or more of the Corporation's subsidiaries, segments, divisions or business units, or any combination of the foregoing. Any Qualifying Option or Qualifying SAR shall be subject only to the requirements of Section 1.62(m) of the Code. Any other Performance-Based Award shall be subject to all of the following provisions of this Section 5.2.

5.2.1 Class; Administrator. The eligible class of persons for Performance-Based Awards under this Section 5.2 shall be officers and employees of the Corporation or one of its Subsidiaries. The Administrator approving Performance-Based Awards or making any certification required pursuant to Section 5.2.4 must be constituted as provided in Section 3.1 for awards that are intended as performance-based compensation under Section 162(m) of the Code.

5.2.2 Performance Goals. The specific performance goals for Performance-Based Awards (other than Qualifying Options and Qualifying SARs) shall be, on an absolute or relative basis, established based on one or more of the following business criteria ("Business Criteria") as selected by the Administrator in its sole discretion: earnings per share, cash flow (which means cash and cash equivalents derived from either net cash flow from operations or net cash flow from operations, financing and investing activities), stock price, total stockholder return, gross revenue, revenue growth, operating income (before or after taxes), net earnings (before or after interest, taxes, depreciation and/or amortization), return on equity or on assets or on net investment, cost containment or reduction, profitability, economic value added, market share, productivity, safety, customer satisfaction, on-time performance, or any combination thereof. These terms are used as applied under generally accepted accounting principles or in the financial reporting of the Corporation or of its Subsidiaries. To qualify awards as performance-based under Section 162(m), the applicable Business Criterion (or Business Criteria, as the case may be) and specific performance goal or goals ("targets") must be established and approved by the Administrator during the first 90 days of the performance period (and, in the case of performance periods of less than one year, in no event after 25% or more of the performance period has elapsed) and while performance relating to such target(s) remains substantially uncertain within the meaning of Section 162(m) of the Code. The terms of the Performance-Based Awards may specify the manner, if any, in which performance targets shall be adjusted to mitigate the unbudgeted impact of material, unusual or nonrecurring gains and losses, accounting changes or other extraordinary events not foreseen at the time the targets were set unless the Administrator provides otherwise at the

time of establishing the targets. The applicable performance measurement period may not be less than three months nor more than 10 years.

5.2.3 Form of Payment; Maximum Performance-Based Award. Grants or awards under this Section 5.2 may be paid in cash or shares of Common Stock or any combination thereof. Grants of Qualifying Options and Qualifying SARs to any one participant in any one calendar year shall be subject to the limit set forth in Section 4.2(b). The maximum number of shares of Common Stock which may be delivered pursuant to Performance-Based Awards (other than Qualifying Options and Qualifying SARs, and other than cash awards covered by the following sentence) that are granted to any one participant in any one calendar year shall not exceed 1,200,000 shares, either individually or in the aggregate, subject to adjustment as provided in Section 7.1. In addition, the aggregate amount of compensation to be paid to any one participant in respect of all Performance-Based Awards payable only in cash and not related to shares of Common Stock and granted to that participant in any one calendar year shall not exceed \$1,000,000. Awards that are cancelled during the year shall be counted against these limits to the extent required by Section 162(m) of the Code.

5.2.4 *Certification of Payment.* Before any Performance-Based Award under this Section 5.2 (other than Qualifying Options and Qualifying SARs) is paid and to the extent required to qualify the award as performance-based compensation within the meaning of Section 162(m) of the Code, the Administrator must certify in writing that the performance target(s) and any other material terms of the Performance-Based Award were in fact timely satisfied.

5.2.5 *Reservation of Discretion.* The Administrator will have the discretion to determine the restrictions or other limitations of the individual awards granted under this Section 5.2 including the authority to reduce awards, payouts or vesting or to pay no awards, in its sole discretion, if the Administrator preserves such authority at the time of grant by language to this effect in its authorizing resolutions or otherwise.

5.2.6 *Expiration of Grant Authority.* As required pursuant to Section 162(m) of the Code and the regulations promulgated thereunder, the Administrator's authority to grant new awards that are intended to qualify as performance-based compensation within the meaning of Section 162(m) of the Code (other than Qualifying Options and Qualifying SARs) shall terminate upon the first meeting of the Corporation's stockholders that occurs in the fifth year following the year in which the Corporation's stockholders first approve this Plan, subject to any subsequent extension that may be approved by stockholders.

5.3 *Award Agreements*. Each award shall be evidenced by either (1) a written award agreement in a form approved by the Administrator and executed by the Corporation by an officer duly authorized to act on its behalf, or (2) an electronic notice of award grant in a form approved by the Administrator and recorded by the Corporation (or its designee) in an electronic recordkeeping system used for the purpose of tracking award grants under this Plan generally (in each case, an "award agreement"), as the Administrator may provide and, in each case and if required by the Administrator, executed or otherwise electronically accepted by the recipient of the award in such form and manner as the Administrator may require. The Administrator may authorize any officer of the Corporation (other than the particular award recipient) to execute any or all award agreements on behalf of the

Corporation. The award agreement shall set forth the material terms and conditions of the award as established by the Administrator consistent with the express limitations of this Plan.

5.4 Deferrals and Settlements. Payment of awards may be in the form of cash, Common Stock, other awards or combinations thereof as the Administrator shall determine, and with such restrictions as it may impose. The Administrator may also require or permit participants to elect to defer the issuance of shares or the settlement of awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares.

5.5 Consideration for Common Stock or Awards. The purchase price for any award granted under this Plan or the Common Stock to be delivered pursuant to an award, as applicable, may be paid by means of any lawful consideration as determined by the Administrator, including, without limitation, one or a combination of the following methods:

- services rendered by the recipient of such award;
- cash, check payable to the order of the Corporation, or electronic funds transfer;
- notice and third party payment in such manner as may be authorized by the Administrator;
- the delivery of previously owned shares of Common Stock;
- by a reduction in the number of shares otherwise deliverable pursuant to the award; or
- subject to such procedures as the Administrator may adopt, pursuant to a "cashless exercise" with a third party who provides financing for the purposes of (or who otherwise facilitates) the purchase or exercise of awards.

In no event shall any shares newly-issued by the Corporation be issued for less than the minimum lawful consideration for such shares or for consideration other than consideration permitted by applicable state law. Shares of Common Stock used to satisfy the exercise price of an option shall be valued at their fair market value on the date of exercise. The Corporation will not be obligated to deliver any shares unless and until it receives full payment of the exercise or purchase price therefor and any related withholding obligations under Section 8.5 and any other conditions to exercise or purchase have been satisfied. Unless otherwise expressly provided in the applicable award agreement, the Administrator may at any time eliminate or limit a participant's ability to pay the purchase or exercise price of any award or shares by any method other than cash payment to the Corporation.

5.6 Definition of Fair Market Value. For purposes of this Plan, "fair market value" shall mean, unless otherwise determined or provided by the Administrator in the circumstances, the closing price (in regular trading) for a share of Common Stock on the New York Stock Exchange (the "Exchange") for the date in question or, if no sales of

Common Stock were reported on the Exchange on that date, the closing price (in regular trading) for a share of Common Stock on the Exchange for the next preceding day on which sales of Common Stock were reported on the Exchange. The Administrator may, however, provide with respect to one or more awards that the fair market value shall equal the closing price (in regular trading) for a share of Common Stock on the Exchange on the last trading day preceding the date in question or the average of the high and low trading prices of a share of Common Stock on the Exchange for the date in question or the most recent trading day. If the Common Stock is no longer listed or is no longer actively traded on the Exchange as of the applicable date, the fair market value of the Common Stock shall be the value as reasonably determined by the Administrator for purposes of the award in the circumstances. The Administrator also may adopt a different methodology for determining fair market value with respect to one or more awards if a different methodology is necessary or advisable to secure any intended favorable tax, legal or other treatment for the particular award(s) (for example, and without limitation, the Administrator may provide that fair market value for purposes of one or more awards will be based on an average of closing prices (or the average of high and low daily trading prices) for a specified period preceding the relevant date).

5.7 Transfer Restrictions.

5.7.1 Limitations on Exercise and Transfer. Unless otherwise expressly provided in (or pursuant to) this Section 5.7 or required by applicable law: (a) all awards are non-transferable and shall not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge; (b) awards shall be exercised only by the participant; and (c) amounts payable or shares issuable pursuant to any award shall be delivered only to (or for the account of) the participant.

5.7.2 *Exceptions*. The Administrator may permit awards to be exercised by and paid to, or otherwise transferred to, other persons or entities pursuant to such conditions and procedures, including limitations on subsequent transfers, as the Administrator may, in its sole discretion, establish in writing. Any permitted transfer shall be subject to compliance with applicable federal and state securities laws and shall not be for value (other than nominal consideration, settlement of marital property rights, or for interests in an entity in which more than 50% of the voting interests are held by the Eligible Person or by the Eligible Person's family members).

5.7.3 *Further Exceptions to Limits on Transfer*. The exercise and transfer restrictions in Section 5.7.1 shall not apply to:

- (a) transfers to the Corporation (for example, in connection with the expiration or termination of the award),
- (b) the designation of a beneficiary to receive benefits in the event of the participant's death or, if the participant has died, transfers to or exercise by the participant's beneficiary, or, in the absence of a validly designated beneficiary, transfers by will or the laws of descent and distribution,

- (c) subject to any applicable limitations on ISOs, transfers to a family member (or former family member) pursuant to a domestic relations order if approved or ratified by the Administrator,
- (d) if the participant has suffered a disability, permitted transfers or exercises on behalf of the participant by his or her legal representative, or
- (e) the authorization by the Administrator of "cashless exercise" procedures with third parties who provide financing for the purpose of (or who otherwise facilitate) the exercise of awards consistent with applicable laws and the express authorization of the Administrator.

5.8 *International Awards*. One or more awards may be granted to Eligible Persons who provide services to the Corporation or one of its Subsidiaries outside of the United States. Any awards granted to such persons may be granted pursuant to the terms and conditions of any applicable sub-plans, if any, appended to this Plan and approved by the Administrator.

6. EFFECT OF TERMINATION OF EMPLOYMENT OR SERVICE ON AWARDS

6.1 *General.* The Administrator shall establish the effect of a termination of employment or service on the rights and benefits under each award under this Plan and in so doing may make distinctions based upon, inter alia, the cause of termination and type of award. If the participant is not an employee of the Corporation or one of its Subsidiaries and provides other services to the Corporation or one of its Subsidiaries, the Administrator shall be the sole judge for purposes of this Plan (unless a contract or the award otherwise provides) of whether the participant continues to render services to the Corporation or one of its Subsidiaries and the date, if any, upon which such services shall be deemed to have terminated.

6.2 *Events Not Deemed Terminations of Service.* Unless the express policy of the Corporation or one of its Subsidiaries, or the Administrator, otherwise provides, the employment relationship shall not be considered terminated in the case of (a) sick leave, (b) military leave, or (c) any other leave of absence authorized by the Corporation or one of its Subsidiaries, or the Administrator; provided that, unless reemployment upon the expiration of such leave is guaranteed by contract or law or the Administrator otherwise provides, such leave is for a period of not more than three months. In the case of any employee of the Corporation or one of its Subsidiaries on an approved leave of absence, continued vesting of the award while on leave from the employ of the Corporation or one of its Subsidiaries may be suspended until the employee returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an award be exercised after the expiration of the term set forth in the applicable award agreement.

6.3 Effect of Change of Subsidiary Status. For purposes of this Plan and any award, if an entity ceases to be a Subsidiary of the Corporation a termination of employment or service shall be deemed to have occurred with respect to each Eligible Person in respect of such Subsidiary who does not continue as an Eligible Person in respect of the Corporation or another Subsidiary that continues as such after giving effect to the transaction or other

event giving rise to the change in status unless the Subsidiary that is sold, spun-off or otherwise divested (or its successor or a direct or indirect parent of such Subsidiary or successor) assumes the Eligible Person's award(s) in connection with such transaction.

7. ADJUSTMENTS; ACCELERATION

7.1 *Adjustments.* Subject to Section 7.2, upon (or, as may be necessary to effect the adjustment, immediately prior to): any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split; any merger, combination, consolidation, or other reorganization; any spin-off, split-up, or similar extraordinary dividend distribution in respect of the Common Stock; or any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall equitably and proportionately adjust (1) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of awards (including the specific share limits, maximums and numbers of shares set forth elsewhere in this Plan), (2) the number, amount and type of shares of Common Stock (or other securities or property) subject to any outstanding awards, (3) the grant, purchase, or exercise price (which term includes the base price of any SAR or similar right) of any outstanding awards, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding awards.

Unless otherwise expressly provided in the applicable award agreement, upon (or, as may be necessary to effect the adjustment, immediately prior to) any event or transaction described in the preceding paragraph or a sale of all or substantially all of the business or assets of the Corporation as an entirety, the Administrator shall equitably and proportionately adjust the performance standards applicable to any then-outstanding performance-based awards to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding performance-based awards.

It is intended that, if possible, any adjustments contemplated by the preceding two paragraphs be made in a manner that satisfies applicable U.S. legal, tax (including, without limitation and as applicable in the circumstances, Section 424 of the Code, Section 409A of the Code and Section 162(m) of the Code) and accounting (so as to not trigger any charge to earnings with respect to such adjustment) requirements.

Without limiting the generality of Section 3.3, any good faith determination by the Administrator as to whether an adjustment is required in the circumstances pursuant to this Section 7.1, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

7.2 Corporate Transactions - Assumption and Termination of Awards. Upon the occurrence of any of the following: any merger, combination, consolidation, or other reorganization in connection with which the Corporation does not survive (or does not survive as a public company in respect of its Common Stock); any exchange of Common Stock or other securities of the Corporation in connection with which the Corporation does not survive (or does not survive as a public company in respect of its Common Stock); any exchange of common survive as a public company in respect of its Corporation does not survive (or does not survive as a public company in respect of its Common Stock); a

sale of all or substantially all the business, stock or assets of the Corporation in connection with which the Corporation does not survive (or does not survive as a public company in respect of its Common Stock); a dissolution of the Corporation; or any other event in which the Corporation does not survive (or does not survive as a public company in respect of its Common Stock); then the Administrator may make provision for a cash payment in settlement of, or for the termination, assumption, substitution or exchange of any or all outstanding share-based awards or the cash, securities or property deliverable to the holder of any or all outstanding share-based awards, based upon, to the extent relevant under the circumstances, the distribution or consideration payable to holders of the Common Stock upon or in respect of such event. Upon the occurrence of any event described in the preceding sentence, then, unless the Administrator has made a provision for the substitution, assumption, exchange or other continuation or settlement of the award or (unless the Administrator has provided for the termination of the award) the award would otherwise continue in accordance with its terms in the circumstances: (1) unless otherwise provided in the applicable award agreement, each then-outstanding option and SAR shall become fully vested, all shares of restricted stock then outstanding shall fully vest free of restrictions, and each other award granted under this Plan that is then outstanding shall become payable to the holder of such award; and (2) each award shall terminate upon the related event; provided that the holder of an option or SAR shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding vested options and SARs (after giving effect to any accelerated vesting required in the circumstances) in accordance with their terms before the termination of such awards (except that in no case shall more than ten days' notice of the impending termination be required and any acceleration of vesting and any exercise of any portion of an award that is so accelerated may be made contingent upon the actual occurrence of the event).

The Administrator may adopt such valuation methodologies for outstanding awards as it deems reasonable in the event of a cash or property settlement and, in the case of options, SARs or similar rights, but without limitation on other methodologies, may base such settlement solely upon the excess if any of the per share amount payable upon or in respect of such event over the exercise or base price of the award.

In any of the events referred to in this Section 7.2, the Administrator may take such action contemplated by this Section 7.2 prior to such event (as opposed to on the occurrence of such event) to the extent that the Administrator deems the action necessary to permit the participant to realize the benefits intended to be conveyed with respect to the underlying shares. Without limiting the generality of the foregoing, the Administrator may deem an acceleration to occur immediately prior to the applicable event and, in such circumstances, will reinstate the original terms of the award if an event giving rise to an acceleration does not occur.

Without limiting the generality of Section 3.3, any good faith determination by the Administrator pursuant to its authority under this Section 7.2 shall be conclusive and binding on all persons.

7.3 Other Acceleration Rules. The Administrator may override the provisions of Section 7.2 by express provision in the award agreement and may accord any Eligible Person a right to refuse any acceleration, whether pursuant to the award agreement or

otherwise, in such circumstances as the Administrator may approve. The portion of any ISO accelerated in connection with an event referred to in Section 7.2 (or such other circumstances as may trigger accelerated vesting of the award) shall remain exercisable as an ISO only to the extent the applicable \$100,000 limitation on ISOs is not exceeded. To the extent exceeded, the accelerated portion of the option shall be exercisable as a nonqualified stock option under the Code.

7.4 **Definition of Change in Control.** With respect to a particular award granted under this Plan, a "Change in Control" shall be deemed to have occurred as of the first day, after the date of grant of the particular award, that any one or more of the following conditions shall have been satisfied:

- (i) the consummation of:
 - (A) any consolidation or merger of the Corporation in which the Corporation is not the continuing or surviving corporation or pursuant to which shares of common stock of the Corporation would be converted into cash, securities or other property, other than a merger of the Corporation in which the holders of common stock of the Corporation immediately prior to the merger have the same proportionate ownership of common stock of the surviving corporation immediately after the merger; or
 - (B) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, the assets of the Corporation.
- (ii) at any time during a period of twenty-four (24) months, fewer than a majority of the members of the Board are Incumbent Directors. For these purposes, "Incumbent Directors" means (A) individuals who constitute the Board at the beginning of such period; and (B) individuals who were nominated or elected by all of, or a committee composed entirely of, the individuals described in (A); and (C) individuals who were nominated or elected by individuals described in (B).
- (iii) any Person (meaning any individual, entity or group within the meaning of Section 13(d)(3) or 14(d) of the Exchange Act) shall, as a result of a tender or exchange offer, open market purchases, privately-negotiated purchases or otherwise, become the beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act), directly or indirectly, of the then-outstanding securities of the Corporation ordinarily (and apart from rights accruing under special circumstances) having the right to vote in the election of members of the Board ("Voting Securities" to be calculated as provided in paragraph (d) of Rule 13d-3 in the case of rights to acquire common stock of the Corporation) representing 20% or more of the combined voting power of the then-outstanding Voting Securities.
- (iv) approval by the stockholders of the Corporation of any plan or proposal for the liquidation or dissolution of the Corporation.

Unless the Board shall determine otherwise, a Change of Control shall not be deemed to have occurred by reason of any corporate reorganization, merger, consolidation, transfer of assets, liquidating distribution or other transaction entered into solely by and between the Corporation and any affiliate thereof, provided such transaction has been approved by at

least two-thirds (2/3) of the Incumbent Directors (as defined above) then in office and voting.

8. OTHER PROVISIONS

8.1 Compliance with Laws. This Plan, the granting and vesting of awards under this Plan, the offer, issuance and delivery of shares of Common Stock, and/or the payment of money under this Plan or under awards are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities law and federal margin requirements) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation or one of its Subsidiaries, provide such assurances and representations to the Corporation or one of its Subsidiaries as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.

8.2 No Rights to Award. No person shall have any claim or rights to be granted an award (or additional awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.

8.3 No Employment/Service Contract. Nothing contained in this Plan (or in any other documents under this Plan or in any award) shall confer upon any Eligible Person or other participant any right to continue in the employ or other service of the Corporation or one of its Subsidiaries, constitute any contract or agreement of employment or other service or affect an employee's status as an employee at will, nor shall interfere in any way with the right of the Corporation or one of its Subsidiaries to change a person's compensation or other benefits, or to terminate his or her employment or other service, with or without cause. Nothing in this Section 8.3, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract other than an award agreement.

8.4 *Plan Not Funded.* Awards payable under this Plan shall be payable in shares or from the general assets of the Corporation, and no special or separate reserve, fund or deposit shall be made to assure payment of such awards. No participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly otherwise provided) of the Corporation or one of its Subsidiaries by reason of any award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or one of its Subsidiaries and any participant, beneficiary or other person. To the extent that a participant, beneficiary or other person acquires a right to receive payment pursuant to any award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Corporation.

8.5 *Tax Withholding*. Upon any exercise, vesting, or payment of any award, or upon the disposition of shares of Common Stock acquired pursuant to the exercise of an ISO prior to satisfaction of the holding period requirements of Section 422 of the Code, or upon

any other tax withholding event with respect to any award, the Corporation or one of its Subsidiaries shall have the right at its option to:

- (a) require the participant (or the participant's personal representative or beneficiary, as the case may be) to pay or provide for payment of at least the minimum amount of any taxes which the Corporation or one of its Subsidiaries may be required to withhold with respect to such award event or payment; or
- (b) deduct from any amount otherwise payable in cash (whether related to the award or otherwise) to the participant (or the participant's personal representative or beneficiary, as the case may be) the minimum amount of any taxes which the Corporation or one of its Subsidiaries may be required to withhold with respect to such award event or payment.

In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 8.1) require or grant (either at the time of the award or thereafter) to the participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, that the Corporation reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their fair market value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy the minimum applicable withholding obligation on exercise, vesting or payment. In no event shall the shares withheld exceed the minimum whole number of shares required for tax withholding under applicable law.

8.6 Effective Date, Termination and Suspension, Amendments.

8.6.1 Effective Date. This Plan is effective as of March 13, 2008, the date of its approval by the Board (the "Effective Date"). This Plan shall be submitted for and subject to stockholder approval no later than twelve months after the Effective Date. Unless earlier terminated by the Board, this Plan shall terminate at the close of business on the day before the tenth anniversary of the Effective Date. After the termination of this Plan either upon such stated expiration date or its earlier termination by the Board, no additional awards may be granted under this Plan, but previously granted awards (and the authority of the Administrator with respect thereto, including the authority to amend such awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan.

8.6.2 *Board Authorization*. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No awards may be granted during any period that the Board suspends this Plan.

8.6.3 *Stockholder Approval.* To the extent then required by applicable law or any applicable listing agency or required under Sections 162, 422 or 424 of the Code to preserve the intended tax consequences of this Plan, or deemed necessary or advisable by the Board, any amendment to this Plan shall be subject to stockholder approval.

8.6.4 *Amendments to Awards.* Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on awards to participants that the Administrator in the prior exercise of its discretion has imposed, without the consent of a participant, and (subject to the requirements of Sections 3.2 and 8.6.5) may make other changes to the terms and conditions of awards. Any amendment or other action that would constitute a repricing of an award is subject to the limitations set forth in Section 3.2.

8.6.5 *Limitations on Amendments to Plan and Awards*. No amendment, suspension or termination of this Plan or amendment of any outstanding award agreement shall, without written consent of the participant, affect in any manner materially adverse to the participant any rights or benefits of the participant or obligations of the Corporation under any award granted under this Plan prior to the effective date of such change. Changes, settlements and other actions contemplated by Section 7 shall not be deemed to constitute changes or amendments for purposes of this Section 8.6.

8.7 *Privileges of Stock Ownership*. Except as otherwise expressly authorized by the Administrator, a participant shall not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the participant. Except as expressly required by Section 7.1 or otherwise expressly provided by the Administrator, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.

8.8 Governing Law; Construction; Severability.

8.8.1 Choice of Law. This Plan, the awards, all documents evidencing awards and all other related documents shall be governed by, and construed in accordance with the laws of the State of Delaware.

8.8.2 Severability. If a court of competent jurisdiction holds any provision invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.

8.8.3 Plan Construction.

- (a) <u>Rule 16b-3</u>. It is the intent of the Corporation that the awards and transactions permitted by awards be interpreted in a manner that, in the case of participants who are or may be subject to Section 16 of the Exchange Act, qualify, to the maximum extent compatible with the express terms of the award, for exemption from matching liability under Rule 16b-3 promulgated under the Exchange Act. Notwithstanding the foregoing, the Corporation shall have no liability to any participant for Section 16 consequences of awards or events under awards if an award or event does not so qualify.
- (b) <u>Section 162(m)</u>. Awards under Section 5.1.4 to persons described in Section 5.2 that are either granted or become vested, exercisable or payable based on attainment of one or more performance goals related to the Business Criteria, as well as Qualifying Options and Qualifying SARs granted to

persons described in Section 5.2, that are approved by a committee composed solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code) shall be deemed to be intended as performance-based compensation within the meaning of Section 162(m) of the Code unless such committee provides otherwise at the time of grant of the award. It is the further intent of the Corporation that (to the extent the Corporation or one of its Subsidiaries or awards under this Plan may be or become subject to limitations on deductibility under Section 162(m) of the Code) any such awards and any other Performance-Based Awards under Section 5.2 that are granted to or held by a person subject to Section 162(m) will qualify as performance-based compensation or otherwise be exempt from deductibility limitations under Section 162(m).

8.9 *Captions*. Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.

8.10 Stock-Based Awards in Substitution for Stock Options or Awards Granted by Other Corporation. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, SARs, restricted stock or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Corporation or one of its Subsidiaries, in connection with a distribution, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Corporation or one of its Subsidiaries, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The awards so granted need not comply with other specific terms of this Plan, provided the awards reflect only adjustments giving effect to the assumption or substitution consistent with the conversion applicable to the Common Stock in the transaction and any change in the issuer of the security. Any shares that are delivered and any awards that are granted by, or become obligations of, the Corporation, as a result of the assumption by the Corporation of, or in substitution for, outstanding awards previously granted by an acquired company (or previously granted by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Corporation or one of its Subsidiaries in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan.

8.11 Non-Exclusivity of Plan. Nothing in this Plan shall limit or be deemed to limit the authority of the Board or the Administrator to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.

8.12 No Corporate Action Restriction. The existence of this Plan, the award agreements and the awards granted hereunder shall not limit, affect or restrict in any way the right or power of the Board or the stockholders of the Corporation to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the capital structure or business of the Corporation or any Subsidiary, (b) any merger, amalgamation, consolidation or change in the ownership of the Corporation or any Subsidiary, (c) any

issue of bonds, debentures, capital, preferred or prior preference stock ahead of or affecting the capital stock (or the rights thereof) of the Corporation or any Subsidiary, (d) any dissolution or liquidation of the Corporation or any Subsidiary, (e) any sale or transfer of all or any part of the assets or business of the Corporation or any Subsidiary, or (f) any other corporate act or proceeding by the Corporation or any Subsidiary. No participant, beneficiary or any other person shall have any claim under any award or award agreement against any member of the Board or the Administrator, or the Corporation or any employees, officers or agents of the Corporation or any Subsidiary, as a result of any such action.

8.13 Other Company Benefit and Compensation Programs. Payments and other benefits received by a participant under an award made pursuant to this Plan shall not be deemed a part of a participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Subsidiary, except where the Administrator expressly otherwise provides or authorizes in writing. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans or arrangements of the Corporation or its Subsidiaries.

8.14 *Clawback Policy*. The awards granted under this Plan are subject to the terms of the Corporation's recoupment, clawback or similar policy as it may be in effect from time to time, as well as any similar provisions of applicable law, any of which could in certain circumstances require repayment or forfeiture of awards or any shares of Common Stock or other cash or property received with respect to the awards (including any value received from a disposition of the shares acquired upon payment of the awards).

###

ALASKA AIR GROUP, INC. 2010 EMPLOYEE STOCK PURCHASE PLAN

Adopted by the. Board of Directors on March 11, 2010

Approved by Stockholders on May 18, 2010

1. **PURPOSE**

The purpose of this Plan is to assist Eligible Employees in acquiring a stock ownership interest in the Company, at a favorable price and upon favorable terms, pursuant to a plan which is intended to qualify as an "employee stock purchase plan" under Section 423 of the Code. This Plan is also intended to encourage Eligible Employees to remain in the employ of the Company or a Participating Subsidiary and to provide them with an additional incentive to advance the best interests of the Company.

2. **DEFINITIONS**

Capitalized terms used herein which are not otherwise defined shall have the following meanings.

- (a) **"Account**" means the bookkeeping account maintained by the Company, or by a record keeper on behalf of the Company, for a Participant pursuant to Section 7(a).
- (b) **"Board**" means the Board of Directors of the Company.
- (c) "Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time.
- (d) "Commission" means the U.S. Securities and Exchange Commission.
- (e) "Committee" means the committee appointed by the Board to administer the Plan pursuant to Section 12.
- (f) **"Common Stock**" means the common stock, par value \$1.00 per share, of the Company, and such other securities or property as may become the subject of Options pursuant to an adjustment made under Section 17.
- (g) **"Compensation**" means an Eligible Employee's base pay, inclusive of overtime and any employer paid leave. Compensation also includes any amounts contributed as salary reduction contributions to a plan qualifying under Section 401(k), 125, or 129 of the Code. Any other form of remuneration is excluded from Compensation, including (but not limited to) the following: cash bonuses, severance pay, hiring bonuses, prizes, awards, relocation or housing allowances, stock option exercises, stock appreciation right payments, the vesting or grant of restricted stock, the payment of stock units, performance awards, auto allowances, tuition reimbursement, perquisites, noncash compensation and other forms of imputed income. Notwithstanding the foregoing, Compensation shall not include any amounts deferred under or paid from any nonqualified deferred compensation plan maintained by the Company or any Subsidiary (including, without limitation, the Company's Nonqualified Deferred Compensation Plan).
- (h) **"Contributions"** means the bookkeeping amounts credited to the Account of the Participant pursuant to this Plan, equal in amount to the amount of Compensation that the Participant has elected to contribute for the purchase of Common Stock under and in accordance with this Plan.

- (i) "Company" means Alaska Air Group, Inc., a Delaware corporation, and its successors.
- (j) "Effective Date" means March 11, 2010, the date on which this Plan was initially adopted by the Board.
- (k) **"Eligible Employee**" means any employee of the Company, or of any Subsidiary which has been designated in writing by the Committee as a "Participating Subsidiary." Notwithstanding the foregoing, "Eligible Employee" shall not include any employee:
 - (1) who has not completed at least six (6) months of continuous employment with the Company or a Subsidiary; or
 - (2) whose customary employment is for less than five (5) months in a calendar year.
- (1) **"Exchange Act**" means the U.S. Securities Exchange Act of 1934, as amended from time to time.
- (m) **"Exercise Date**" means, with respect to an Offering Period, the last day of that Offering Period.
- (n) **"Fair Market Value**" on any date means:
 - (1) if the Common Stock is listed or admitted to trade on a national securities exchange, the closing price of a share of Common Stock on such date on the principal national securities exchange on which the Common Stock is so listed or admitted to trade, or, if there is no trading of the Common Stock on such date, then the closing price of a share of Common Stock on such exchange on the next preceding date on which there was trading in the shares of Common Stock;
 - (2) in the absence of exchange data required to determine Fair Market Value pursuant to the foregoing, the value as established by the Committee as of the relevant time for purposes of this Plan.
- (o) "Grant Date" means, with respect to an Offering Period, the first day of that Offering Period.
- (p) "Individual Limit" has the meaning given to such term in Section 4(b).
- (q) "Offering Period" means the six (6) month period commencing on each Grant Date; provided, however, that the Committee may declare, as it deems appropriate and in advance of the applicable Offering Period, a shorter (not to be less than three months) Offering Period or a longer (not to exceed 27 months) Offering Period. Unless otherwise expressly provided by the Committee in advance of a particular Offering Period, the Grant Date for that Offering Period may not occur on or before the Exercise Date for the immediately preceding Offering Period.
- (r) "Option" means the stock option to acquire shares of Common Stock granted to a Participant pursuant to Section 8.
- (s) "Option Price" means the per share exercise price of an Option as determined in accordance with Section 8(b).
- (t) **"Parent**" means any corporation (other than the Company) in an unbroken chain of corporations ending with the Company in which each corporation (other than the Company) owns stock possessing 50% or more of the total combined voting power of all classes of stock in one or more of the other corporations in the chain.

- (u) **"Participant**" means an Eligible Employee who has elected to participate in this Plan and who has filed a valid and effective Subscription Agreement to make Contributions pursuant to Section 6.
- (v) "Participating Subsidiary" shall have the meaning given to such term in Section 19(c).
- (w) **"Plan**" means this Alaska Air Group, Inc. 2010 Employee Stock Purchase Plan, as it may be amended or restated from time to time.
- (x) **"Subscription Agreement**" means the written agreement or applicable electronic form of agreement filed by an Eligible Employee with the Company pursuant to Section 6 to participate in this Plan.
- (y) **"Subsidiary**" means any corporation (other than the Company) in an unbroken chain of corporations (beginning with the Company) in which each corporation (other than the last corporation) owns stock possessing 50% ore more of the total combined voting power of all classes of stock in one or more of the other corporations in the chain.

3. ELIGIBILITY

Any person employed as an Eligible Employee as of a Grant Date shall be eligible to participate in this Plan during the Offering Period in which such Grant Date occurs, subject to the Eligible Employee satisfying the requirements of Section 6.

4. STOCK SUBJECT TO THIS PLAN; SHARE LIMITATIONS

- (a) Aggregate Share Limit. Subject to the provisions of Section 17, the capital stock that may be delivered under this Plan will be shares of the Company's authorized but unissued Common Stock. The maximum number of shares of Common Stock that may be delivered pursuant to Options granted under this Plan is 8,000,000 shares, subject to adjustments pursuant to Section 17.
- (b) Individual Share Limit. The maximum number of shares of Common Stock that any one individual may acquire upon exercise of his or her Option with respect to any one Offering Period is 8,000, subject to adjustments pursuant to Section 17 (the "Individual Limit"). The Committee may amend the Individual Limit, effective no earlier than the first Offering Period commencing after the adoption of such amendment, without stockholder approval.
- (c) **Shares Not Actually Delivered**. Shares that are subject to or underlie Options, which for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall again, except to the extent prohibited by law, be available for subsequent Options under this Plan.

5. OFFERING PERIODS

During the term of this Plan, the Company will grant Options to purchase shares of Common Stock in each Offering Period to all Participants in that Offering Period. Unless otherwise specified by the Committee in advance of the Offering Period, Offering Periods will be of approximately six (6) months duration and will commence on September 1 and March 1 each year and will end on the following February 28 (or 29, in the case of a leap year) and August 31, respectively. Each Option shall become effective on the Grant Date of the Offering Period with respect to which the Option is granted. The term of each Option shall be the duration of the related Offering Period and shall end on the Exercise Date of that Offering Period. The first Offering Period shall commence as of a date determined by the Board or Committee, but no earlier than the Effective Date. Offering Periods shall continue until this Plan is terminated in

accordance with Section 18 or 19, or, if earlier, until no shares of Common Stock remain available for Options pursuant to Section 4.

6. PARTICIPATION

- (a) Enrollment. An Eligible Employee may become a participant in this Plan by completing a Subscription Agreement on a form approved by and in a manner prescribed by the Committee (or its delegate). To become effective, a Subscription Agreement must be signed by the Eligible Employee and be filed with the Company at the time specified by the Committee, but in all cases prior to the start of the Offering Period with respect to which it is to become effective, and must set forth a whole percentage (or, if the Committee so provides, a stated amount) of the Eligible Employee's Compensation to be credited to the Participant's Account as Contributions each pay period.
- (b) **Contribution Limits.** Notwithstanding the foregoing, a Participant may not elect to contribute less than one percent (1%) nor more than ten percent (10%) (or such other limit as the Committee may establish prior to the start of the applicable Offering Period) of his or her Compensation during any one pay period as Plan Contributions. The Committee also may prescribed other limits, rules or procedures for Contributions.
- (c) Content and Duration of Subscription Agreements. Subscription Agreements shall contain the Eligible Employee's authorization and consent to the Company's withholding from his or her Compensation the amount of his or her Contributions. An Eligible Employee's Subscription Agreement, and his or her participation election and withholding consent thereon, shall remain valid for all Offering Periods until (1) the Eligible Employee's participation terminates pursuant to the terms hereof, (2) the Eligible Employee files a new Subscription Agreement that becomes effective, or (3) the Committee requires that a new Subscription Agreement be executed and filed with the Company.

7. METHOD OF PAYMENT OF CONTRIBUTIONS

- (a) Participation Accounts. The Company shall maintain on its books, or cause to be maintained by a record keeper, an Account in the name of each Participant. The percentage of Compensation elected to be applied as Contributions by a Participant shall be deducted from such Participant's Compensation on each payday during the period for payroll deductions set forth below and such payroll deductions shall be credited to that Participant's Account as soon as administratively practicable after such date. A Participant may not make any additional payments to his or her Account. A Participant's Account shall be reduced by any amounts used to pay the Option Price of shares acquired, or by any other amounts distributed pursuant to the terms hereof.
- (b) **Payroll Deductions**. Subject to such other rules as the Committee may adopt, payroll deductions with respect to an Offering Period shall commence as of the first day of the payroll period which coincides with or immediately follows the applicable Grant Date and shall end on the last date of the payroll period which coincides with or immediately proceeds the applicable Exercise Date, unless sooner terminated by the Participant as provided in Section 7(d) or until his or her participation terminates pursuant to Section 11.
- (c) **Changes in Contribution Elections for Next Offering Period**. A Participant may discontinue, increase, or decrease the level of his or her Contributions (within the Plan limits) by completing and filing with the Company, on such terms as the Committee (or its delegate) may prescribe, a new Subscription Agreement which indicates such election. Subject to any other timing requirements that the Committee may impose, an election

pursuant to this Section 7(c) shall be effective with the first Offering Period that commences after the Company's receipt of such election. Except as contemplated by Section 7(d) and 7(e), changes in Contribution levels may not take effect during an Offering Period. Other modifications or suspensions of Subscription Agreements are not permitted.

- (d) **Withdrawal During an Offering Period**. A Participant may terminate his or her Contributions during an Offering Period (and receive a distribution of the balance of his or her Account in accordance with Section 11) by completing and filing with the Company, in such form and on such terms as the Committee (or its delegate) may prescribe, a written withdrawal form or applicable electronic withdrawal form which shall be signed by the Participant. Such termination shall be effective as soon as administratively practicable after its receipt by the Company. A withdrawal election pursuant to this Section 7(d) with respect to an Offering Period shall only be effective, however, if it is received by the Company prior to the Exercise Date of the Offering Period (or such earlier deadline that the Committee may reasonably require to process the withdrawal prior to the applicable Exercise Date). Partial withdrawals of Accounts are not permitted.
- (e) Discontinuance of Contributions During an Offering Period. A Participant may discontinue his or her Contributions at any time during an Offering Period by completing and filing with the Company, on such terms as the Committee (or its delegate) may prescribe, a new Subscription Agreement which indicates such election. If a Participant elects to discontinue his or her Contributions pursuant to this Section 7(e), the Contributions previously credited to the Participant's Account for that Offering Period shall be used to exercise the Participant's Option as of the applicable Exercise Date in accordance with Section 9 (unless the Participant makes a timely withdrawal election in accordance with Section 7(d), in which case such Participant's Account shall be paid to him or her in cash in accordance with Section 11(a)).
- (f) **Leaves of Absence**. During leaves of absence approved by the Company or a Participating Subsidiary and meeting the requirements of Regulation 1.421-1(h)(2) under the Code, a Participant may elect to continue participation in this Plan by delivering cash payments to the Company on his or her normal paydays equal to the reduction in his or her Plan Contributions caused by his or her leave.

8. GRANT OF OPTION

- (a) Grant Date; Number of Shares. On each Grant Date, each Eligible Employee who is a Participant during that Offering Period shall be granted an Option to purchase a number of shares of Common Stock. The Option shall be exercised on the Exercise Date. The number of shares of Common Stock subject to the Option shall be determined by dividing the Participant's Account balance as of the applicable Exercise Date by the Option Price, subject to the limits of Section 8(c).
- (b) Option Price. The Option Price per share of the shares subject to an Option for an Offering Period shall be the <u>lesser</u> of: (i) 85% of the Fair Market Value of a Share on the Grant Date of that Offering Period; or (ii) 85% of the Fair Market Value of a Share on the Exercise Date of that Offering Period; provided, however, that the Committee may provide prior to the start of any Offering Period that the Option Price for that Offering Period shall be determined by applying a discount amount (not to exceed 15%) to either (1) the Fair Market Value of a share of Common Stock on that Grant Date of that Offering Period, or (2) the Fair Market Value of a share of Common Stock on the Exercise Date of that Offering Period, or (3) the lesser of the Fair Market Value of a share of Common Stock on

the Grant Date of that Offering Period or the Fair Market Value of a share of Common Stock on the Exercise Date of that Offering Period. Notwithstanding anything to the contrary in the preceding provisions of this Section 8(b), in no event shall the Option Price per share be less than the par value of a share of Common Stock.

- (c) Limits on Share Purchases. Notwithstanding anything else contained herein, the maximum number of shares subject to an Option for an Offering Period shall be subject to the Individual Limit in effect on the Grant Date of that Offering Period (subject to adjustment pursuant to Section 17) and any person who is otherwise an Eligible Employee shall not be granted any Option (or any Option granted shall be subject to compliance with the following limitations) or other right to purchase shares under this Plan to the extent:
 - (1) it would, if exercised, cause the person to own stock (within the meaning of Section 423(b)(3) of the Code) possessing 5% or more of the total combined voting power or value of all classes of stock of the Company, or of any Parent, or of any Subsidiary; or
 - (2) such Option causes such individual to have rights to purchase stock under this Plan and any other plan of the Company, any Parent, or any Subsidiary which is qualified under Section 423 of the Code which accrue at a rate which exceeds \$25,000 of the fair market value of the stock of the Company, of any Parent, or of any Subsidiary (determined at the time the right to purchase such stock is granted, before giving effect to any discounted purchase price under any such plan) for each calendar year in which such right is outstanding at any time.

For purposes of the foregoing, a right to purchase stock accrues when it first becomes exercisable during the calendar year. In determining whether the stock ownership of an Eligible Employee equals or exceeds the 5% limit set forth above, the rules of Section 424(d) of the Code (relating to attribution of stock ownership) shall apply, and stock which the Eligible Employee may purchase under outstanding options shall be treated as stock owned by the Eligible Employee.

9. EXERCISE OF OPTION

- (a) Purchase of Shares. Unless a Participant withdraws pursuant to Section 7(d) or the Participant's Plan participation is terminated as provided in Section 11, his or her Option for the purchase of shares shall be exercised automatically on the Exercise Date for that Offering Period, without any further action on the Participant's part, and the maximum number of whole shares of Common Stock subject to such Option (subject to the limits of Section 8(c)) shall be purchased at the Option Price with the balance of such Participant's Account.
- (b) Account Balance Remaining After Purchase. If any amount which is not sufficient to purchase a whole share remains in a Participant's Account after the exercise of his or her Option on the Exercise Date: (1) such amount shall be credited to such Participant's Account for the next Offering Period, if he or she is then a Participant; or (2) if such Participant is not a Participant in the next Offering Period, or if the Committee so elects, such amount shall be refunded to such Participant as soon as administratively practicable after such date. If the share limit of Section 4(a) is reached, any amount that remains in a Participant's Account after the exercise of his or her Option on the Exercise Date to purchase the number of shares that he or she is allocated shall be refunded to the Participant as soon as administratively practicable after such date. If any amount which exceeds the limits of Section 8(c) remains in a Participant's Account after the exercise of

his or her Option on the Exercise Date, such amount shall be refunded to the Participant as soon as administratively practicable after such date.

10. DELIVERY OF SHARES

As soon as administratively practicable after the Exercise Date, the Company shall, in its discretion, either deliver to each Participant a certificate representing the shares of Common Stock purchased upon exercise of his or her Option, provide for the crediting of such shares in book entry form in the name of the Participant, or provide for an alternative arrangement for the delivery of such shares to a broker or record keeping service for the benefit of the Participant. In the event the Company is required to obtain from any commission or agency authority to issue any such certificate or otherwise deliver such shares, the Company will seek to obtain such authority. If the Company is unable to obtain from any such certificate or other delivery of such shares, or if for any reason the Company cannot issue or deliver shares of Common Stock and satisfy Section 21, the Company shall be relieved from liability to any Participant except that the Company shall return to each Participant to whom such shares cannot be issued or delivered the amount of the balanced credited to his or her Account that would have otherwise been used for the purchase of such shares.

11. TERMINATION OF EMPLOYMENT; CHANGE IN ELIGIBLE STATUS

- (a) General. Except as provided in Section 11(b) below, if a Participant ceases to be an Eligible Employee for any reason (including, without limitation, due to the Participant's death, disability, resignation or retirement, or due to a layoff or other termination of employment with or without cause), or if the Participant elects to withdraw from the Plan pursuant to Section 7(d), at any time prior to the last day of an Offering Period in which he or she participates, such Participant's Account shall be paid to him or her (or, in the event of the Participant's death, to the person or persons entitled thereto under Section 13) in cash, and such Participant's Option and participation in the Plan shall automatically terminate as of the time that the Participant ceased to be an Eligible Employee.
- (b) Change in Eligible Status; Leave. If a Participant (1) ceases to be an Eligible Employee during an Offering Period but remains an employee of the Company or a Subsidiary through the Exercise Date (for example, and without limitation, due to a change in the Participant's employer from the Company or a Participating Subsidiary to a non-Participating Subsidiary, if the Participant's employer ceases to maintain the Plan as a Participating Subsidiary but otherwise continues as a Subsidiary, or if the Participant's customary level of employment no longer satisfies the requirements set forth in the definition of Eligible Employee), or (2) during an Offering Period commences a sick leave, military leave, or other leave of absence approved by the Company or a Participating Subsidiary, and the leave meets the requirements of Treasury Regulation Section 1.421-1(h)(2) and the Participant's Contributions shall cease (subject to Section 7(d)), and the Contributions previously credited to the Participant's Account for that Offering Period shall be used to exercise the Participant's Option as of the applicable Exercise Date in accordance with Section 7(d), in which case such Participant's Account shall be paid to him or her in cash in accordance with Section 11(a)).
- (c) **Re-Enrollment**. A Participant's termination from Plan participation precludes the Participant from again participating in this Plan during that Offering Period. However, such termination shall not have any effect upon his or her ability to participate in any

succeeding Offering Period, provided that the applicable eligibility and participation requirements are again then met. A Participant's termination from Plan participant shall be deemed to be a revocation of that Participant's Subscription Agreement and such Participant must file a new Subscription Agreement to resume Plan participation in any succeeding Offering Period.

(d) **Change in Subsidiary Status**. For purposes of this Plan, if a Subsidiary ceases to be a Subsidiary, each person employed by that Subsidiary will be deemed to have terminated employment for purposes of this Plan, unless the person continues as an employee of the Company or another Subsidiary.

12. ADMINISTRATION

- (a) The Committee. The Board shall appoint the Committee, which shall be composed of not less than two members of the Board. The Board may, at any time, increase or decrease the number of members of the Committee, may remove from membership on the Committee all or any portion of its members, and may appoint such person or persons as it desires to fill any vacancy existing on the Committee, whether caused by removal, resignation, or otherwise. The Board may also, at any time, assume the administration of all or a part of this Plan, in which case references (or relevant references in the event the Board assumes the administration of only certain aspects of this Plan) to the "Committee" shall be deemed to be references to the Board. Action of the Committee with respect to this Plan shall be taken pursuant to a majority vote or by the unanimous written consent of its members. No member of the Committee shall be entitled to act on or decide any matters relating solely to himself or herself or solely to any of his or her rights or benefits under this Plan.
- (b) **Powers and Duties of the Committee.** Subject to the express provisions of this Plan, the Committee shall supervise and administer this Plan and shall have the full authority and discretion: (1) to construe and interpret this Plan and any agreements defining the rights and obligations of the Company, any Subsidiary, and Participants under this Plan; (2) to further define the terms used in this Plan; (3) to prescribe, amend and rescind rules and regulations relating to the administration of this Plan (including, without limitation, deadlines for making elections or for providing any notices contemplated by this Plan, which deadlines may be more restrictive than any deadlines otherwise contemplated by this Plan); and (4) to make all other determinations and take such other action as contemplated by this Plan or as may be necessary or advisable for the administration of this Plan or the effectuation of its purposes. Notwithstanding anything else contained in this Plan to the contrary, the Committee may also adopt rules, procedures or sub-plans applicable to particular Subsidiaries or locations, which sub-plans may be designed to be outside the scope of Section 423 of the Code and need not comply with the otherwise applicable provisions of this Plan.
- (c) **Decisions of the Committee are Binding**. Any action taken by, or inaction of, the Company, any Subsidiary, the Board or the Committee relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons.
- (d) **Indemnification**. Neither the Board nor any Committee, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan, and all such persons shall be entitled to indemnification and reimbursement by the Company in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees)

arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.

- (e) **Reliance on Experts.** In making any determination or in taking or not taking any action under this Plan, the Committee or the Board, as the case may be, may obtain and may rely upon the advice of experts, including professional advisors to the Company. No director, officer or agent of the Company or any Participating Subsidiary shall be liable for any such action or determination taken or made or omitted in good faith.
- (f) **Delegation**. The Committee may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Company or a Subsidiary.

13. DESIGNATION OF BENEFICIARY

If the Committee permits beneficiary designations with respect to this Plan, then each Participant may file, on a form and in a manner prescribed by the Committee (or its delegate), a written designation of a beneficiary who is to receive any shares or cash from or with respect to such Participant's Account under this Plan in the event of such Participant's death. If a Participant is married and the designated beneficiary is not solely his or her spouse, spousal consent shall be required for such designation to be effective unless it is established (to the satisfaction of the Committee or its delegate) that there is no spouse or that the spouse cannot be located. The Committee may rely on the last designation of a beneficiary filed by a Participant in accordance with this Plan. Beneficiary designations may be changed by the Participant (and his or her spouse, if required) at any time on forms provided and in the manner prescribed by the Committee (or its delegate).

If a Participant dies with no validly designated beneficiary under this Plan who is living at the time of such Participant's death (or in the event the Committee does not permit beneficiary designations under this Plan), the Company shall deliver all shares and/or cash payable pursuant to the terms hereof to the executor or administrator of the estate of the Participant, or if no such executor or administrator has been appointed, the Company, in its discretion, may deliver such shares and/or cash to the spouse or to any one or more dependents or relatives of the Participant, or if no spouse, dependent or relative is known to the Company, then to such other person as the Company may designate.

If a Participant's death occurs before the end of an Offering Period or subsequent to the end of an Offering Period but prior to the delivery to him or her or for his or her benefit of any shares deliverable under the terms of this Plan, and the Company has notice of the Participant's death, then any shares purchased for that Offering Period and any remaining balance of such Participant's Account shall be paid to such beneficiary (or such other person entitled to such payment pursuant to this Section 13). If the Committee permits beneficiary designations with respect to this Plan, any such designation shall have no effect with respect to shares purchased and actually delivered (or credited, as the case may be) to or for the benefit of the Participant.

14. TRANSFERABILITY

Neither Contributions credited to a Participant's Account nor any Options or rights with respect to the exercise of Options or right to receive shares under this Plan may be anticipated, alienated, encumbered, assigned, transferred, pledged or otherwise disposed of in any way (other than by will, the laws of descent and distribution, or as provided in Section 13) by the Participant. Any such attempt at anticipation, alienation, encumbrance, assignment, transfer, pledge or other disposition shall be without effect and all amounts shall be paid and all shares shall be delivered in accordance with the provisions of this Plan. Amounts payable or shares deliverable pursuant to this Plan shall be paid or delivered only to (or credited in the name of, as the case may be) the

Participant or, in the event of the Participant's death, the Participant's beneficiary pursuant to Section 13.

15. USE OF FUNDS; INTEREST

All Contributions received or held by the Company under this Plan will be included in the general assets of the Company and may be used for any corporate purpose. Notwithstanding anything else contained herein to the contrary, no interest will be paid to any Participant or credited to his or her Account under this Plan (in respect of Account balances, refunds of Account balances, or otherwise). Amounts payable under this Plan shall be payable in shares of Common Stock or from the general assets of the Company and, except for any shares that may be reserved on the books of the Company for issuance with respect to this Plan, no special or separate reserve, fund or deposit shall be made to assure payment of amounts that may be due with respect to this Plan.

16. **REPORTS**

Statements shall be provided (either electronically or in written form, as the Committee may provide from time to time) to Participants as soon as administratively practicable following each Exercise Date. Each Participant's statement shall set forth, as of such Exercise Date, that Participant's Account balance immediately prior to the exercise of his or her Option, the Option Price, the number of whole shares purchased and his or her remaining Account balance, if any.

17. ADJUSTMENTS OF AND CHANGES IN THE STOCK

Upon or in contemplation of any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend), or reverse stock split; any merger, combination, consolidation, or other reorganization; split-up, spin-off, or any similar extraordinary dividend distribution in respect of the Common Stock (whether in the form of securities or property); any exchange of Common Stock or other securities of the Company, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; or a sale of substantially all of the assets of the Company as an entirety occurs; then the Committee shall equitably and proportionately adjust (1) the number and type of shares or the number and type of other securities that thereafter may be made the subject of Options (including the specific maxima and numbers of shares set forth elsewhere in this Plan), (2) the number, amount and type of shares (or other securities or property) subject to any or all outstanding Options, (3) the Option Price of any or all outstanding Options, and/or (4) the securities, cash or other property deliverable upon exercise of any outstanding Options, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding Options.

Upon the occurrence of any event described in the preceding paragraph, or any other event in which the Company does not survive (or does not survive as a public company in respect of its Common Stock); then the Committee may make provision for a cash payment or for the substitution or exchange of any or all outstanding Options for cash, securities or property to be delivered to the holders of any or all outstanding Options based upon the distribution or consideration payable to holders of the Common Stock upon or in respect of such event.

The Committee may adopt such valuation methodologies for outstanding Options as it deems reasonable in the event of a cash or property settlement and, without limitation on other methodologies, may base such settlement solely upon the excess (if any) of the amount payable upon or in respect of such event over the Option Price of the Option.

In any of such events, the Committee may take such action sufficiently prior to such event to the extent that the Committee deems the action necessary to permit the Participant to realize the benefits intended to be conveyed with respect to the underlying shares in the same manner as is or will be available to stockholders generally.

18. POSSIBLE EARLY TERMINATION OF PLAN AND OPTIONS

Upon a dissolution or liquidation of the Company, or any other event described in Section 17 that the Company does not survive or does not survive as a publicly-traded company in respect of its Common Stock, as the case may be, the Plan and, if prior to the last day of an Offering Period, any outstanding Option granted with respect to that Offering Period shall terminate, subject to any provision that has been expressly made by the Board for the survival, substitution, assumption, exchange or other settlement of the Plan and Options. In the event a Participant's Option is terminated pursuant to this Section 18 without a provision having been made by the Board for a substitution, exchange or other settlement of the Option, such Participant's Account shall be paid to him or her in cash without interest.

19. TERM OF PLAN; AMENDMENT OR TERMINATION

- (a) Effective Date; Termination. Subject to Section 19(b), this Plan shall become effective as of the Effective Date. No new Offering Periods shall commence on or after March 1, 2020 and this Plan shall terminate as of the Exercise Date on or immediately following such date unless sooner terminated pursuant to Section 18 or this Section 19. In the event that all of the shares of Common Stock made available under this Plan are subscribed prior to the expiration of this Plan, this Plan shall terminate at the end of that Offering Period and the shares available shall be allocated for purchase by Participants in that Offering Period on a pro-rata basis determined with respect to Participants' Account balances.
- (b) Board Amendment Authority. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part and without notice. Stockholder approval for any amendment or modification shall not be required, except to the extent required by law or applicable stock exchange rules, or required under Section 423 of the Code in order to preserve the intended tax consequences of this Plan. No Options may be granted during any suspension of this Plan or after the termination of this Plan, but the Committee will retain jurisdiction as to Options then outstanding in accordance with the terms of this Plan. No amendment, modification, or termination pursuant to this Section 19(b) shall, without written consent of the Participant, affect in any manner materially adverse to the Participant any rights or benefits of such Participant or obligations of the Company under any Option granted under this Plan prior to the effective date of such change. Changes contemplated by Section 17 or Section 18 shall not be deemed to constitute changes or amendments requiring Participant consent.
- (c) **Certain Additional Committee Authority**. Notwithstanding the amendment provisions of Section 19(b) and without limiting the Board's authority thereunder and without limiting the Committee's authority pursuant to any other provision of this Plan, the Committee shall have the right (1) to designate from time to time the Subsidiaries whose employees may be eligible to participate in this Plan (including, without limitation, any Subsidiary that may first become such after the date stockholders first approve this Plan) (each a "**Participating Subsidiary**"), and (2) to change the service and other qualification requirements sets forth under the definition of Eligible Employee in Section 2 (subject to the requirements of Section 423(b) of the Code and applicable rules and regulations thereunder). Any such change shall not take effect earlier than the first Offering Period that starts on or after the effective date of such change. Any such change shall not require stockholder approval.

20. NOTICES

All notices or other communications by a Participant to the Company contemplated by this Plan shall be deemed to have been duly given when received in the form and manner specified by the Committee (or its delegate) at the location, or by the person, designated by the Committee (or its delegate) for that purpose.

21. CONDITIONS UPON ISSUANCE OF SHARES

This Plan, the granting of Options under this Plan and the offer, issuance and delivery of shares of Common Stock are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities laws) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Company, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Company and as a condition precedent to the exercise of his or her Option, provide such assurances and representations to the Company as the Committee may deem necessary or desirable to assure compliance with all applicable legal requirements.

22. PLAN CONSTRUCTION

- (a) Section 16. It is the intent of the Company that transactions involving Options under this Plan (other than "Discretionary Transactions" as that term is defined in Rule 16b-3(b)(1) promulgated by the Commission under Section 16 of the Exchange Act, to the extent there are any Discretionary Transactions under the Plan), in the case of Participants who are or may be subject to the prohibitions of Section 16 of the Exchange Act, satisfy the requirements for exemption under Rule 16b-3(c) promulgated by the Commission under Section 16 of the Exchange Act to the maximum extent possible. Notwithstanding the foregoing, the Company shall have no liability to any Participant for Section 16 consequences of Options or other events with respect to this Plan.
- (b) Section 423. Except as the Committee may expressly provide in the case of one or more sub-plans adopted pursuant to Section 12(b), this Plan and Options are intended to qualify under Section 423 of the Code. Accordingly, all Participants are to have the same rights and privileges (within the meaning of Section 423(b)(5) of the Code and except as not required thereunder to qualify this Plan under Section 423) under this Plan, subject to differences in Compensation among Participants and subject to the Contribution and share limits of this Plan.
- (c) **Interpretation**. If any provision of this Plan or of any Option would otherwise frustrate or conflict with the intents expressed above, that provision to the extent possible shall be interpreted so as to avoid such conflict. If the conflict remains irreconcilable, the Committee may disregard the provision if it concludes that to do so furthers the interest of the Company and is consistent with the purposes of this Plan as to such persons in the circumstances.

23. EMPLOYEES' RIGHTS

(a) No Employment Rights. Nothing in this Plan (or in any Subscription Agreement or other document related to this Plan) will confer upon any Eligible Employee or Participant any right to continue in the employ or other service of the Company or any Subsidiary, constitute any contract or agreement of employment or other service or effect an employee's status as an employee at will, nor shall interfere in any way with the right of the Company or any Subsidiary to change such person's compensation or other benefits or to terminate his or her employment or other service, with or without cause. Nothing contained in this Section 23(a), however, is intended to adversely affect any express independent right of any such person under a separate employment or service contract other than a Subscription Agreement.

- (b) **No Rights to Assets of the Company.** No Participant or other person will have any right, title or interest in any fund or in any specific asset (including shares of Common Stock) of the Company or any Subsidiary by reason of any Option hereunder. Neither the provisions of this Plan (or of any Subscription Agreement or other document related to this Plan), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan will create, or be construed to create, a trust of any kind or a fiduciary relationship between the Company or any Subsidiary and any Participant, Beneficiary or other person. To the extent that a Participant, Beneficiary or other person acquires a right to receive payment pursuant to this Plan, such right will be no greater than the right of any unsecured general creditor of the Company.
- (c) **No Stockholder Rights**. A Participant will not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the Participant. No adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.

24. MISCELLANEOUS

- (a) **Governing Law**. This Plan, the Options, Subscription Agreements and other documents related to this Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware.
- (b) **Severability**. If any provision shall be held by a court of competent jurisdiction to be invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.
- (c) **Captions and Headings**. Captions and headings are given to the sections of this Plan solely as a convenience to facilitate reference. Such captions and headings shall not be deemed in any way material or relevant to the construction of interpretation of this Plan or any provision hereof.
- (d) No Effect on Other Plans or Corporate Authority. The adoption of this Plan shall not affect any other Company or Subsidiary compensation or incentive plans in effect. Nothing in this Plan will limit or be deemed to limit the authority of the Board or Committee (1) to establish any other forms of incentives or compensation for employees of the Company or any Subsidiary (with or without reference to the Common Stock), or (2) to grant or assume options (outside the scope of and in addition to those contemplated by this Plan) in connection with any proper corporate purpose; to the extent consistent with any other plan or authority. Benefits received by a Participant under an Option granted pursuant to this Plan shall not be deemed a part of the Participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Company or any Subsidiary, except where the Committee or the Board (or the Board of Directors of the Subsidiary that sponsors such plan or arrangement, as applicable) expressly otherwise provides or authorizes in writing.

25. TAX WITHHOLDING

Notwithstanding anything else contained in this Plan herein to the contrary, the Company may deduct from a Participant's Account balance as of an Exercise Date, before the exercise of the Participant's Option is given effect on such date, the amount of taxes (if any) which the Company

reasonably determines it or any Subsidiary may be required to withhold with respect to such exercise. In such event, the maximum number of whole shares subject to such Option (subject to the other limits set forth in this Plan) shall be purchased at the Option Price with the balance of the Participant's Account (after reduction for tax withholding amount).

Should the Company for any reason be unable, or elect not to, satisfy its or any Subsidiary's tax withholding obligations in the manner described in the preceding paragraph with respect to a Participant's exercise of an Option, or should the Company or any Subsidiary reasonably determine that it or an affiliated entity has a tax withholding obligation with respect to a disposition of shares acquired pursuant to the exercise of an Option prior to satisfaction of the holding period requirements of Section 423 of the Code, the Company or Subsidiary, as the case may be, shall have the right at its option to (1) require the Participant to pay or provide for payment of the amount of any taxes which the Company or Subsidiary reasonably determines that it or any affiliate is required to withhold with respect to such event or (2) deduct from any amount otherwise payable to or for the account of the Participant the amount of any taxes which the Company or Subsidiary reasonably determines that it or an affiliate is required to withhold with respect to such event.

26. NOTICE OF SALE

Any person who has acquired shares under this Plan shall give prompt written notice to the Company of any sale or other transfer of the shares if such sale or transfer occurs (1) within the two-year period after the Grant Date of the Offering Period with respect to which such shares were acquired, or (2) within the twelve-month period after the Exercise Date of the Offering Period with respect to which such shares were acquired

EXHIBIT A

ALASKA AIR GROUP, INC.

STOCK DEFERRAL PLAN FOR NON-EMPLOYEE DIRECTORS

ARTICLE I: INTRODUCTION

1.1 Purpose. Alaska Air Group, Inc. (the "Company") has established this Alaska Air Group, Inc. Stock Deferral Plan for Non-Employee Directors (the "Plan") to attract and retain non-employee members of the Company's Board of Directors (the "Board") by providing opportunities to defer stock compensation.

1.2 Legal status. The Plan is a deferred compensation plan for the Company's non-employee directors. The Company intends the Plan to be unfunded for purposes of the Internal Revenue Code of 1986, as amended (the "Code"), and to satisfy the requirements of Code Section 409A. Plan provisions will be interpreted consistent with this intent.

1.3 Effective date. The Plan takes effect May 2, 2012.

ARTICLE II: DEFINITIONS

2.1 Key terms and construction. When used in the Plan, the following terms will have the meanings set forth below. Several other terms are not defined here, but, for convenience, are defined as they are introduced into the text. References to specific Code Sections, such as Section 409A, include any final regulations, Revenue Rulings, and guidance of general applicability thereunder.

a. <u>Beneficiary</u> means the person or persons (including, without limitation, a trust) designated by a Participant to receive his or her benefits in the event of the Participant's death. A Participant may change or revoke a designation without the consent of any Beneficiary. A designation, change, or revocation must be in writing. If a Participant fails to properly designate a Beneficiary, or survives his or her Beneficiaries, benefits will be paid to the Participant's estate.

b. <u>Board Year</u> means the period beginning the day after the Company's annual shareholders' meeting and ending on the date of the following year's annual shareholders' meeting.

c. Common Stock means Alaska Air Group, Inc. Common Stock.

d. <u>Enrollment Period</u> means the election period that precedes the beginning of each Plan Year. Provided, however, that a new director may make an election within 30 days after the date he or she first becomes eligible to participate in the Plan to defer Common Stock awarded for the Board Year next following such election. For example, an individual who is first appointed a director on March 1, 2013 may elect by March 31, 2013 to defer Common Stock awarded for the Board Year beginning in May 2013.

e. <u>Participant</u> means a non-employee director who elects to participate in the Plan. A Participant will remain a Participant as long as he or she has an account balance in the Plan.

f. Plan Year means the calendar year.

g. <u>Separation from Service</u> occurs (or a Participant Separates from Service) when the Participant experiences a "separation from service" within the meaning of Code Section 409A. A Participant who becomes an employee of the Company will be treated as separating from service for Plan purposes if the Participant would be deemed to have separated from service as a director disregarding his or her service as an employee.

ARTICLE III: ADMINISTRATION

3.1 Plan administrator. The Compensation and Leadership Development Committee of the Company's Board of Directors administers the Plan (the "Administrator"). The Administrator may appoint one or more delegates to discharge any or all of the Administrator's responsibilities under the Plan. The Administrator and delegates have all of the discretionary authority, rights, and duties that are necessary or appropriate for the proper administrator of the Plan and may rely upon financial, legal, and accounting advisors, as they see fit. The decisions of the Administrator and delegates, including interpretations, determinations of amounts due, and resolutions of claims under the Plan, are final and binding on all parties.

ARTICLE IV: ELIGIBILITY

4.1 Plan eligibility. All Board members who are not employees of the Company are eligible to participate in the Plan. Non-employee directors serving on the Board as of the Plan's date of adoption will be first eligible to participate during the Enrollment Period that takes place in 2012 with respect to Common Stock awarded for the Board Year beginning in May 2013.

ARTICLE V: ACCOUNTS

5.1 Establishment of accounts. The Administrator will establish a bookkeeping account for each Participant to which the deferrals described in Article VI are credited. The Administrator may establish sub-accounts for Participants, as the Administrator deems appropriate. The accounts and sub-accounts will be adjusted to reflect the distribution of benefits and dividends on and changes in the value of Common Stock. Dividends will be deemed to be reinvested in shares of Common Stock, based on the closing price of the Common Stock on the New York Stock Exchange (the "NYSE") on the date the dividends are paid. Accounts and subaccounts will be valued as of the last day of each calendar month on which the NYSE is open for business (the "Valuation Date").

ARTICLE VI: ELECTIONS

6.1 Elective stock deferrals. During an Enrollment Period (or, in the case of a new director, within 30 days after he or she first becomes eligible), a director may elect to defer all or a portion (in whole shares) of his or her Common Stock award and, in lieu thereof, have non-voting notional shares in the form of deferred stock units (or such other form as the Administrator deems appropriate) credited to his or her Plan account. Such election will become irrevocable on the last day of the applicable election period. A director's deferral election will apply only to Common Stock awarded for the Board Year next following the election.

1. **Election procedures.** To be valid, elections must be made on forms provided by the Administrator and received by the Administrator on or before the specified deadline. An election must state the number or percentage of shares of Common Stock to be awarded that the director wishes to defer.

ARTICLE VII: VESTING

7.1 Vesting. A Participant is at all times 100% vested in his or her account balance.

ARTICLE VIII: DISTRIBUTIONS

8.1 Form and time of distribution. A Participant (or the Participant's Beneficiary, in the case of death) will receive his or her entire account balance in a single lump sum as soon as practicable, but not later than 90 days, after the earliest of the Participant's Separation from Service or death. Account balances will be paid in full shares of Common Stock, with fractional shares, if any, paid in cash based on the closing price of a share of Common Stock on the NYSE on the Valuation Date immediately preceding the distribution.

8.2 Tax consequences. No person, including, without limitation, the Company and its directors, officers, agents, and employees makes any representation or guarantee about the tax consequences of participating in the Plan.

ARTICLE IX: SOURCE OF PAYMENTS

9.1 Benefits and expenses. Benefits and expenses of administering the Plan will be paid from the Company's general assets. Benefits are unfunded and the Company is not required to set aside, earmark, or escrow any funds or other assets to satisfy its obligations under the Plan. Participants and Beneficiaries have no interest in any specific assets of the Company, other than the unsecured right to receive shares of Common Stock (and cash for fractional shares) pursuant to the terms of the Plan. In this regard, Participants acknowledge that the accounts referred to in Article V are merely bookkeeping entries used to track their benefits.

ARTICLE X: MISCELLANEOUS PROVISIONS

10.1 Plan amendment and termination. The Company reserves the right to amend, discontinue, or terminate the Plan at any time by action of its Board. However, no amendment, discontinuance, or termination will reduce the benefits of a Participant that are earned and vested as of the date the Board approves the amendment, discontinuance, or termination. The Company may not accelerate the timing of distributions, except to the extent Code Section 409A permits.

10.2 Non-alienation of benefits. The interest of a Participant or Beneficiary in his or her Plan benefits are not subject to the claims of the Participant's or Beneficiary's creditors and may not be voluntarily or involuntarily sold, transferred, pledged, alienated, assigned, anticipated, or encumbered. Any attempt by a Participant or Beneficiary to do so will be null and void.

10.3 No retention rights. Participation in the Plan does not give a Participant the right to be retained as a director of the Company or to receive any benefits not specifically provided under the Plan.

10.4 Compliance with securities laws. Subject to Code Section 409A, the Board may impose additional, or modify or eliminate existing, Plan terms as it deems necessary or appropriate in order to comply with federal or state securities laws.

10.5 Indemnification. To the extent the law allows, the Company will indemnify and hold harmless the Administrator, any of the Administrator's delegates, and any employee who may act on behalf of the Company in the administration of the Plan from and against any liability, loss, cost, or expense (including reasonable attorneys' fees) incurred at any time as a result of or in connection with any claims, demands, actions, or causes of action of any Participant, any person claiming through any Participant, or any other person, party, or authority claiming to have an interest in the Plan or standing to act for any person or groups having an interest in the Plan, for or on account of any of the acts or omissions (or alleged acts or omissions) of the Administrator, any delegate, or any employee, except to the extent resulting from that person's willful misconduct.

10.7 Successors. The Plan is binding on the Company and its successors and assigns and on each Participant and Beneficiary.

10.7 Severability. In case any provision of the Plan is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining provisions of the Plan and the Plan will be construed and enforced as if the illegal or invalid provision had never been set forth.

10.8 Controlling law. To the extent not superseded by the laws of the United States, the laws of the state of Delaware are controlling in all matters relating to the Plan, without regard to that state's choice of laws provisions.

IN WITNESS WHEREOF, the Company has caused this Plan to be signed by the Chair of its Compensation and Leadership Development Committee this _____ day of _____, 2012.

Alaska Air Group, Inc.

By:				
-	 	 	 	

Title:

Change of Control Agreement

THIS CHANGE OF CONTROL AGREEMENT (this "Agreement") by and between Alaska Air Group, Inc., a Delaware corporation ("Air Group"), and **NAME** (the "Executive") is hereby entered into effective as of the **DATE** day of **MONTH**, **YEAR** (the "Effective Date").

The Board of Directors (the "Board") of Air Group has determined that it is in the best interests of Air Group and its subsidiaries will have the continued dedication of the Executive, notwithstanding the possibility, threat or occurrence of a Change of Control (as defined in Section 3). The Board believes that it is imperative to diminish the inevitable distraction of the Executive by virtue of the personal uncertainties and risks created by a pending or threatened Change of Control and to encourage the Executive's full attention and dedication to Air Group currently and in the event of any threatened or pending Change of Control, and to provide the Executive with compensation and benefits arrangements upon a Change of Control that ensure that the compensation and benefits expectations of the Executive's interests with those of Air Group's stockholders. Therefore, in order to accomplish these objectives, the Board has caused Air Group to enter into this Agreement.

NOW, THEREFORE, IT IS HEREBY AGREED AS FOLLOWS:

1. Term

This Agreement shall be effective as of the Effective Date. This Agreement will continue in effect through the third anniversary of the Effective Date. However, upon the first anniversary of the Effective Date and upon each subsequent anniversary of the Effective Date, the term of this Agreement shall be extended automatically for one (1) additional year (such that upon the first anniversary of the Effective Date the term of this Agreement shall be extended through the fourth anniversary of the Effective Date and so on), unless Air Group delivers written notice prior to such anniversary of the Effective Date to the Executive that this Agreement will not be extended or further extended, as the case may be, and if such notice is given this Agreement will terminate at the end of the term then in progress.

Notwithstanding the foregoing, in the event a Change of Control occurs during the original or any extended term of this Agreement, this Agreement will remain in effect for the longer of: (i) twenty-four (24) months beyond the month in which such Change of Control occurred; or (ii) until all obligations of Air Group hereunder have been fulfilled and all benefits required hereunder have been paid to the Executive. For purposes of clarity, subject to Section 4.1, benefits shall be payable to the Executive under this Agreement only with respect to a single Change of Control of Air Group. Accordingly, no Change of Control after the first Change of Control shall be considered for purposes of this Agreement.

2. Certain Definitions

- (a) "Accrued Obligations" is defined in Section 7(a)(i).
- (b) "affiliated company" means any company controlled by, controlling or under common control with Air Group.
- (c) "Annual Base Salary" is defined in Section 5(b)(i).
- (d) "Annual Bonus" is defined in Section 5(b)(ii).

(e) "Business Combination" means (i) a reorganization, exchange of securities, merger or consolidation involving Air Group or (ii) the sale or other disposition of all or substantially all the assets of Air Group.

(f) "Cause" means basis for termination for reason of admission by the Executive or substantiation by the Employer of:

- (i) embezzlement, dishonesty or other fraud, conviction of a felony or conspiracy against the Employer; or
- (ii) if prior to a Change of Control, any willful or intentional injury to either the Employer, its property, or its employees in connection with the business affairs of the Employer.

(g) "Change of Control Date" means the first date (if any) during the term of this Agreement (determined in accordance with Section 1) on which a Change of Control occurs. Anything in this Agreement to the contrary notwithstanding, if a Change of Control occurs and if the Executive's employment with the Employer is terminated within six (6) months prior to the Change of Control Date, and if it is reasonably demonstrated by the Executive that such termination of employment (i) was at the request of a third party who has taken steps reasonably calculated to effect the Change of Control or (ii) otherwise arose in connection with or anticipation of the Change of Control, then for all purposes of this Agreement the "Change of Control Date" shall mean the date immediately prior to the date of such termination of employment.

- (h) "Code" means the Internal Revenue Code of 1986, as amended.
- (i) "Employer" means, collectively, Air Group and any of its subsidiaries that employs the Executive.
- (j) "Employment Period" is defined in Section 4.
- (k) "Exchange Act" means the Securities Exchange Act of 1934, as amended.

(l) "Good Reason Separation" means the Executive's voluntary Separation from Service within two years after the occurrence without the Executive's consent of one or more of the following events:

- (i) the material reduction in the Executive's annual base salary;
- (ii) the material diminution or reduction of the Executive's authority, duties, or responsibilities;
- (iii) a material change in the geographic location at which the Executive must perform services; or
- (iv) any material breach by the Employer of any other provision of this Agreement;

provided, however, that an Executive shall not be entitled to a Good Reason Separation unless the Executive shall have furnished written notice to the Employer of the condition claimed to constitute the basis for the Good Reason Separation within 90 days of the initial existence of such condition, and the Employer shall have not remedied such condition within a period of 30 days after its receipt of such notice from the Executive.

(m) "Incentive Plan" means Air Group's Management Incentive Plan.

(n) "Incumbent Director" means a member of the Board who has been either (i) nominated by a majority of the directors of Air Group then in office or (ii) appointed by directors so nominated, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board.

- (o) "Notice of Termination" is defined in Section 6(a).
- (p) "Person" means any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d) of the Exchange Act).
- (q) "Recent Average Bonus" is defined in Section 5(b)(ii).

(r) "Retirement Plan" means the Employer's funded pension plan or any successor plan thereto.

(s) "Separation from Service" (and its derivatives, such as "Separates from Service") means a termination of services provided by the Executive to the Employer, whether such termination of services is voluntary or involuntary, as determined by the Board in accordance with Section 409A of the Code and Treasury Regulation Section 1.409A-1(h).

(t) "Welfare Benefit Continuation" is defined in Section 7(b).

3. Change of Control

For the purpose of this Agreement, a "Change of Control" means the occurrence of any of the following:

- (a) the consummation of:
 - (i) any consolidation or merger of Air Group in which Air Group is not the continuing or surviving corporation or pursuant to which shares of common stock of Air Group would be converted into cash, securities or other property, other than a merger of Air Group in which the holders of common stock of Air Group immediately prior to the merger have the same proportionate ownership of common stock of the surviving corporation immediately after the merger; or
 - (ii) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, the assets of Air Group.

(b) at any time during a period of twenty-four (24) months, fewer than a majority of the members of the Board are Incumbent Directors. "Incumbent Directors" means:

- (i) individuals who constitute the Board at the beginning of such period; and
- (ii) individuals who were nominated or elected by all of, or a committee composed entirely of, the individuals described in (i); and
- (iii) individuals who were nominated or elected by individuals described in (ii).
- (c) any Person shall, as a result of a tender or exchange offer, open market purchases, privately-negotiated purchases or otherwise, become the beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act), directly or indirectly, of the then-outstanding securities of Air Group ordinarily (and apart from rights accruing under special circumstances) having the right to vote in the election of members of the Board ("Voting Securities" to be calculated as provided in paragraph (d) of Rule 13d-3 in the case of rights to acquire common stock of Air Group) representing 20% or more of the combined voting power of the then-outstanding Voting Securities.
- (d) approval by the stockholders of Air Group of any plan or proposal for the liquidation or dissolution of Air Group.

Unless the Board shall determine otherwise, a Change of Control shall not be deemed to have occurred by reason of any corporate reorganization, merger, consolidation, transfer of assets, liquidating distribution or other transaction entered into solely by and between Air Group and any Affiliate thereof, provided such transaction has been approved by at least two-thirds (2/3) of the Incumbent Directors (as defined above) then in office and voting.

4. Employment Period

Air Group hereby agrees to continue the Executive in its employ, and the Executive hereby agrees to remain in the employ of Air Group, in accordance with the terms and provisions of this Agreement, for the period commencing on the Change of Control Date and ending on the [third (new CEO/EVP), second (new AS VP and QX President),

first (new QX VPs)] anniversary of such date (the "Employment Period"), in an executive capacity, responsible for, among other things, duties associated with such capacity, and, subject to the general supervision of the Board as required by the Delaware General Corporation Law, such other duties and responsibilities as are not inconsistent with the express terms of this Agreement. Such employment may be with Air Group or any of its principal operating subsidiaries, as appropriate to the management structure developed by Air Group. Air Group agrees that it will not take any action, or make any demands on the Executive, that may be deemed to arbitrarily, unreasonably or unnecessarily interfere with the performance of the services to be rendered by the Executive hereunder.

Prior to the Change of Control Date, the Executive's employment with the Employer is at will.

5. Terms of Employment

(a) Position and Duties.

(i) During the Employment Period, (A) the Executive's position (including status, offices, titles and reporting requirements), authority, duties and responsibilities shall be in accordance with Section 4 and (B) the Executive's services shall be performed within the metropolitan area in which the Executive was situated immediately prior to the Change of Control Date, except for required travel in the Employer business to the extent consistent with the Executive's duties in Section 4.

(ii) During the Employment Period, and excluding any periods of vacation and sick leave to which the Executive is entitled, the Executive agrees to devote reasonable attention and time during normal business hours to the business and affairs of the Employer and, to the extent necessary to discharge the responsibilities assigned to the Executive hereunder, to use the Executive's reasonable best efforts to perform faithfully and efficiently such responsibilities. During the Employment Period it shall not be a violation of this Agreement for the Executive to (A) serve on corporate, civic or charitable boards or committees, (B) deliver lectures, fulfill speaking engagements or teach at educational institutions, or (C) manage personal investments, so long as such activities do not significantly interfere with the performance of the Executive's responsibilities as an employee of the Employer in accordance with this Agreement. It is expressly understood and agreed that to the extent that any such activities have been conducted by the Executive prior to the Change of Control Date, the continued conduct of such activities (or the conduct of activities similar in nature and scope thereto) subsequent to the Change of Control Date shall not thereafter be deemed to interfere with the performance of the Executive's responsibilities to the Employer.

(b) Compensation.

(i) <u>Base Salary</u>. During the Employment Period, the Executive shall receive an annual base salary ("Annual Base Salary"), which shall be paid in equal installments in accordance with the regular payroll schedule applicable to similarly-situated executives, at least equal to 12 times the highest monthly base salary paid or payable to the Executive by the Employer in respect of the 12-month period immediately preceding the month in which the Change of Control Date occurs. For purposes of this Agreement, Annual Base Salary shall not include any payments by the Employer on the Executive's behalf pursuant to any incentive, savings or retirement plans, any welfare benefit plans or any fringe benefit plans, in each case, of the Employer or any affiliated company, of the type identified in paragraphs (iii) through (vi) of this Section 5(b), or any reimbursement of expenses by the Employer or any affiliated company in accordance with paragraph (v) of this Section 5(b), but shall include vacation pay in accordance with paragraph (viii) of this Section 5(b). During the Employment Period, the Annual Base Salary shall be reviewed at least annually and shall be increased at any time and from time to time as shall be substantially consistent with increases in base salary generally awarded in the ordinary course of business to other peer executives of the Employer and any affiliated companies. Any increase in Annual Base Salary shall not serve to limit or reduce any other obligation to the Executive under this Agreement. Annual Base Salary shall not be reduced after any such increase, and the term Annual Base Salary as utilized in this Agreement shall refer to Annual Base Salary as so increased.

(ii) <u>Annual Bonus</u>. In addition to Annual Base Salary, the Executive shall be awarded, for each fiscal year ending during the Employment Period, an annual bonus (the "Annual Bonus") in cash at least equal to the greater of (A) the Executive's target annual bonus (annualized if such target bonus is based on a period of less than

12 full months) in effect on the Change of Control Date and (B) the average annualized (for any fiscal year consisting of less than 12 full months or with respect to which the Executive has been employed by the Employer for less than 12 full months) bonus paid or payable, including by reason of any deferral, to the Executive by the Employer in respect of the three fiscal years immediately preceding the fiscal year in which the Change of Control Date occurs (the "Recent Average Bonus"). Each such Annual Bonus shall be paid between January 1 and March 15 of the year next following the fiscal year for which the Annual Bonus is awarded, unless the Executive shall elect, pursuant to the terms of the AAGI Nonqualified Deferred Compensation Plan (or any successor to that plan), to defer the receipt of such Annual Bonus.

(iii) Incentive, Savings and Retirement Plans. During the Employment Period, the Executive shall be entitled to participate in all incentive, savings and retirement plans, practices, policies and programs applicable generally to other peer executives of the Employer, but in no event shall such plans, practices, policies and programs provide the Executive with incentive opportunities (measured with respect to both regular and special incentive opportunities, to the extent, if any, that such distinction is applicable), savings opportunities and retirement benefit opportunities, in each case, that are less favorable, in the aggregate, than the most favorable of those provided by the Employer for the Executive under such plans, practices, policies and programs as in effect at any time during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, those provided generally at any time after the Change of Control Date to other peer executives of the Employer.

(iv) <u>Welfare Benefit Plans</u>. During the Employment Period, the Executive and/or the Executive's family, as the case may be, shall be eligible for participation in and shall receive all benefits under welfare benefit plans, practices, policies and programs provided by the Employer (including, without limitation, medical, prescription, dental, disability, salary continuance, employee life, group life, accidental death and travel accident insurance plans and programs) to the extent applicable generally to other peer executives of the Employer, but in no event shall such plans, practices, policies and programs provide the Executive with benefits that are less favorable, in the aggregate, than the most favorable of such plans, practices, policies and programs in effect for the Executive at any time during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, those provided generally at any time after the Change of Control Date to other peer executives of the Employer.

(v) <u>Expenses</u>. During the Employment Period, the Executive shall be entitled to reimbursement promptly, but in no event later than the end of the calendar year following the year in which the expense is incurred, for all reasonable employment expenses incurred by the Executive in accordance with the most favorable policies, practices and procedures of the Employer in effect for the Executive at any time during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, as in effect generally at any time thereafter with respect to other peer executives of the Employer.

(vi) <u>Fringe Benefits</u>. During the Employment Period, the Executive shall be entitled to fringe benefits in accordance with the most favorable plans, practices, programs and policies of the Employer in effect for the Executive at any time during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, as in effect generally at any time thereafter with respect to other peer executives of the Employer. To the extent that a plan, practice, program, or policy provides for the reimbursement of the Executive's expenses, such reimbursements shall be made promptly, but in no event later than the end of the calendar year following the year in which the expense is incurred.

(vii) <u>Vacation</u>. During the Employment Period, the Executive shall be entitled to paid vacation in accordance with the most favorable plans, policies, programs and practices of the Employer as in effect for the Executive at any time during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, as in effect generally at any time thereafter with respect to other peer executives of the Employer.

6. Termination of Employment

(a) <u>Termination</u>. The Executive's employment shall terminate automatically upon the Executive's death during the Employment Period. The Executive's employment may be terminated at any time during the Employment

Period for any reason by either the Executive or by the Employer, communicated by a notice of termination to the other party hereto given in accordance with Section 13(b) (a "Notice of Termination").

(b) <u>Date of Termination</u>. "Date of Termination" means (i) if the Executive's employment is terminated by the Employer or by the Executive, the date of receipt of the Notice of Termination or any later date specified therein, as the case may be, and (ii) if the Executive's employment is terminated by reason of death, the date of death of the Executive.

7. Obligations of the Employer Upon Certain Terminations; Release

If the Executive's employment is terminated during the Employment Period by the Executive in a Good Reason Separation or by the Employer without Cause, and such termination constitutes a Separation from Service:

(a) the Employer shall pay to the Executive in a lump sum in cash the aggregate of the following amounts:

(i) A lump sum amount equal to all payments to which the Executive would have been entitled during the Employment Period, but for the Separation from Service, including, without limitation, the aggregate amounts of the Executive's Annual Base Salary (calculated in accordance with Section 5(b)(i) hereof) and the aggregate amounts of the Executive's Annual Bonus (calculated in accordance with Section 5(b)(ii) hereof), payable in each case during the Employment Period, less any amounts comprising any portion of Annual Base Salary or Annual Bonus actually received by the Executive during the period commencing on the Change of Control Date and ending on the date of such Separation from Service.

(ii) A separate lump sum supplemental retirement benefit equal to the difference between (1) the actuarial equivalent (utilizing for this purpose the actuarial assumptions utilized with respect to the Employer defined benefit retirement plan during the 90-day period immediately preceding the Change of Control Date) of the benefits payable under the Employer defined benefit retirement plans, the 1995 Elected Officers' Supplementary Retirement Plan (or if applicable to the Executive the Defined Contribution OSRP Plan feature of the AAGI Nonqualified Deferred Compensation Plan) and any similar plans (other than the deferred bonus or deferred retention incentive features of the AAGI Nonqualified Deferred Compensation Plan) providing benefits for the Executive that the Executive would receive if the Executive's employment continued at the compensation level provided for in Section 5(b) and for the remainder of the Employment Period (assuming for this purpose that all accrued benefits are fully vested and that benefit accrual formulas are no less advantageous to the Executive than those in effect during the 90-day period immediately preceding the Change of Control Date), and (2) the actuarial equivalent (utilizing for this purpose the same assumptions as outlined above) of the Executive's actual benefit paid (or payable), if any, under the foregoing plans; and

(b) for the remainder of the Employment Period, or such longer period as any plan, program, practice or policy may provide, the Employer shall continue benefits to the Executive and/or the Executive's family at least equal to those that would have been provided to them in accordance with the plans, programs, practices and policies described in Section 5(b)(iv) if the Executive had not incurred a Separation from Service in accordance with the most favorable plans, practices, programs or policies of the Employer as in effect and applicable generally to other executives and their families during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, as in effect generally at any time thereafter with respect to other peer executives of the Employer and their families; provided, however, that if the Executive becomes reemployed with another employer and is eligible to receive medical or other welfare benefits under another employer-provided plan, the medical and other welfare benefits described herein shall be secondary to those provided under such other plan during such applicable period of eligibility (such continuation of such benefits for the applicable period herein set forth shall be hereinafter referred to as "Welfare Benefit Continuation"). For purposes of determining eligibility of the Executive for retiree benefits pursuant to such plans, practices, programs and policies, the Executive shall be considered to have remained employed until the end of the Employment Period and to have retired on the last day of such period; provided, however, that the Executive shall be entitled to the more favorable of the retiree benefits in effect on the date of the Executive's Separation from Service or the retiree benefits in effect on the date that would have been the last date of the Employment Period if the Executive had remained employed. Notwithstanding anything in this Section 7(b) to the contrary, in no event shall any health care benefit (whether for medical, dental, or vision care) that is subject to Code Section 409A be continued for a period longer than 18 months after the date of the Executive's Separation from Service;

(c) to the extent not theretofore paid or provided, the Employer shall timely pay or provide to the Executive and/or the Executive's family any other amounts or benefits required to be paid or provided or which the Executive and/or the Executive's family is eligible to receive pursuant to Section 5(b)(v) and (vi) of this Agreement under any plan, program, policy or practice or contract or agreement of the Employer as in effect and applicable generally to other peer executives and their families during the 90-day period immediately preceding the Change of Control Date or, if more favorable to the Executive, as in effect generally thereafter with respect to other peer executives of the Employer and their families (such other amounts and benefits shall be hereinafter referred to as the "Other Benefits"). Notwithstanding anything in this Section 7(c) to the contrary, in no event shall any Other Benefit be paid to the extent that such payment would trigger any additional tax, penalty or interest imposed by Code Section 409A.

(d) Release. Notwithstanding anything else contained in this Agreement to the contrary, as a condition precedent to any Employer obligation to the Executive pursuant to this Section 7, the Executive shall, within 21 days following his or her Separation from Service with the Employer (and within the period of time provided for in the release), provide the Employer with a valid, executed general release agreement in substantially the form attached hereto as Exhibit A, and such release agreement shall have not been revoked by the Executive pursuant to any revocation rights afforded by applicable law. The Employer shall have no obligation to make any payment to the Executive in accordance with all applicable laws, rules and regulations. Notwithstanding the foregoing, the Employer shall provide the benefits described in Sections 7(b) and 7(c) above following the Executive's Separation from Service (or Change of Control if Section 7(e) applies) but, if the Executive does not timely provide the release agreement contemplated by this Section 7(d) or revokes such release agreement, the Employer shall have no further obligation to provide the benefits set forth in Section 7(b) or, except to the extent required under the applicable plan, program or policy, the benefits set forth in Section 7(c).

(e) <u>Timing of Payment</u>. The lump sum amount specified in Section 7(a) above shall be paid in the second calendar month following the month in which the Separation from Service occurs; provided, however, that if such payment is to be made pursuant to a termination of employment that occurs prior to a Change of Control as contemplated by Section 2(g), and provided that the Executive provides a release agreement as contemplated by Section 7(d) within 21 days following the Change of Control (and does not revoke such release within any revocation period provided by applicable law), such payment shall be made in the month following the month in which the Change of Control occurs and any benefits to which the Executive may be entitled pursuant to this Section 7 shall commence with the month in which the Change of Control occurs.

8. Nonexclusivity of Rights

Nothing in this Agreement shall prevent or limit the Executive's continuing or future participation in any plan, program, policy or practice provided by the Employer and for which the Executive may qualify, nor shall anything herein limit or otherwise affect such rights as the Executive may have under any contract or agreement with the Employer. Amounts that are vested benefits or that the Executive is otherwise entitled to receive under any plan, policy, practice or program of or any contract or agreement with the Employer or any of its affiliated companies at or subsequent to the Date of Termination shall be payable in accordance with such plan, policy, practice or program or contract or agreement except as explicitly modified by this Agreement.

9. Full Settlement; Resolution of Disputes

(a) The Employer obligation to make the payments provided for in this Agreement and otherwise to perform its obligations hereunder shall not be affected by any set-off, counterclaim, recoupment, defense or other claim, right or action that the Employer may have against the Executive or others. In no event shall the Executive be obligated to seek other employment or take any other action by way of mitigation of the amounts payable to the Executive under any of the provisions of this Agreement, and, except as provided in Section 7(b), such amounts shall

not be reduced whether or not the Executive obtains other employment. The Employer agrees to pay promptly upon invoice, to the full extent permitted by law, all legal fees and expenses that the Executive may incur as a result of any contest (regardless of the outcome thereof) by the Employer, the Executive or others of the validity or enforceability of, or liability under, any provision of this Agreement or any guarantee of performance thereof (including as a result of any contest by the Executive about the amount of any payment pursuant to this Agreement).

(b) If there shall be any dispute between the Employer and the Executive (i) in the event of any termination of the Executive's employment by the Employer, whether such termination was in connection with or in anticipation of a Change of Control so as to trigger the Change of Control Date under the definition of that term in Section 2, then, unless and until there is a final, nonappealable judgment by a court of competent jurisdiction declaring that such termination was in connection with or in anticipation of a Change of Control, the Employer shall pay all amounts, and provide all benefits, to the Executive and/or the Executive's family or other beneficiaries, as the case may be, that the Employer would be required to pay or provide pursuant to Section 7 as though such termination were in connection with or in anticipation of a Change of Control; provided, however, that the Employer shall not be required to pay any disputed amounts pursuant to this Section 9(b) except upon receipt of an undertaking by or on behalf of the Executive to repay all such amounts to which the Executive is ultimately adjudged by such court not to be entitled.

10. Certain Adjustments

Notwithstanding anything else contained in this Agreement, in the event that the Executive becomes entitled to the (a) payments or other benefits described in Section 7 hereof and the Executive becomes or would be subject to the tax imposed by Section 4999 of the Code or any successor provision (the "Excise Tax") as a result of such payments and benefits and any other payments or benefits from the Employer required to be taken into account under Code Section 280G(b)(2) (collectively, "Parachute Payments"), the Parachute Payments shall be reduced (but not below zero) so that the maximum amount of Parachute Payments (after reduction) is one dollar (\$1.00) less than the amount that would cause the Parachute Payments to be subject to the Excise Tax; provided that such a reduction shall only be made if and to the extent that a reduction in the Parachute Payments would result in the Executive retaining a larger amount, on an after-tax basis (taking into account federal, state and local income taxes and the Excise Tax), than if the Executive received all of the Parachute Payments (such reduced amount is referred to hereinafter as the "Limited Benefit Amount"). Unless the Executive shall have given prior written notice specifying a different order to Air Group to effectuate the Limited Benefit Amount, any such notice to be consistent with the requirements of Section 409A of the Code to avoid the imputation of any tax, penalty or interest thereunder, Air Group shall reduce or eliminate the Parachute Payments by first reducing or eliminating any cash severance benefits, then by reducing or eliminating any accelerated vesting of stock options, then by reducing or eliminating any accelerated vesting of other equity-based awards, then by reducing or eliminating any other remaining Parachute Payments. The preceding provisions of this Section 10(a) shall take precedence over the provisions of any other plan, arrangement or agreement governing the Executive's rights and entitlements to any benefits or compensation.

(b) A determination as to whether the Parachute Payments shall be reduced to the Limited Benefit Amount pursuant to this Agreement and the amount of such Limited Benefit Amount shall be made by a certified public accounting or compensation consulting firm of national reputation (the "Accounting Firm") which shall provide detailed supporting calculations to both the Employer and the Executive within 15 business days of the receipt of notice from the Executive that the Executive has received a payment under Section 7, or such earlier time as is requested by the Employer. In the event that the Accounting Firm is serving as accountant or auditor for the individual, entity or group effecting the Change of Control, the Executive shall appoint another nationally recognized accounting firm to make the determinations required hereunder (which accounting firm shall then be referred to as the Accounting Firm hereunder). All fees and expenses of the Accounting Firm shall be borne solely by the Employer. If the Accounting Firm determines that no Excise Tax is payable by the Executive, it shall furnish the Executive with a written opinion that failure to report the Excise Tax on the Executive's applicable federal income tax return would not result in the imposition of a negligence or similar penalty. Any determination by the Accounting Firm shall be binding upon the Employer and Executive.

11. Confidential Information

The Executive shall hold in a fiduciary capacity for the benefit of the Employer all secret or confidential information, knowledge or data relating to the Employer or any of its affiliated companies, and their respective businesses, that shall have been obtained by the Executive during the Executive's employment by the Employer or any of its affiliated companies and that shall not be or become public knowledge (other than by acts by the Executive or representatives of the Executive in violation of this Agreement). After termination of the Executive's employment with the Employer, the Executive shall not, without the prior written consent of the Employer or as may otherwise be required by law or legal process, communicate or divulge any such information, knowledge or data to anyone other than the Employer and those designated by it. In no event shall an asserted violation of the provisions of this Section 11 constitute a basis for deferring or withholding any amounts otherwise payable to the Executive under this Agreement.

12. Successors

(a) This Agreement is personal to the Executive and without the prior written consent of the Employer shall not be assignable by the Executive otherwise than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by the Executive's legal representatives.

(b) This Agreement shall inure to the benefit of and be binding on the Employer and its successors and assigns.

(c) The Employer will require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all the business and/or assets of the Employer to assume expressly and agree to perform this Agreement in the same manner and to the same extent that the Employer would be required to perform it if no such succession had taken place. As used in this Agreement, Employer shall mean the Employer as hereinbefore defined and any successor to its business and/or assets as aforesaid that assumes and agrees to perform this Agreement by operation of law, or otherwise.

13. Miscellaneous

(a) This Agreement shall be governed by and construed in accordance with the laws of the state of Washington, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and shall have no force or effect. This Agreement may not be amended or modified otherwise than by a written agreement executed by the parties hereto or their respective successors and legal representatives.

(b) All notices and other communications hereunder shall be in writing and shall be given by hand delivery to the other party or by registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to the Executive:

If to Air Group: Alaska Air Group, Inc. P.O. Box 68947 Seattle, WA 98168 Attention: Corporate Secretary

or to such other address as either party shall have furnished to the other in writing in accordance herewith. Notice and communications shall be effective when actually received by the addressee.

(c) The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.

(d) The Employer may withhold from any amounts payable under this Agreement such federal, state or local taxes as shall be required to be withheld pursuant to any applicable law or regulation.

(e) The Executive's or the Employer's failure to insist on strict compliance with any provision hereof or any other provision of this Agreement or the failure to assert any right the Executive or the Employer may have hereunder, shall not be deemed to be a waiver of such provision or right or any other provision or right of this Agreement.

(f) The Executive and the Employer acknowledge that, except as may otherwise be provided under any other written agreement between the Executive and the Employer, the employment of the Executive by the Employer is "at will" and, prior to the Change of Control Date, may be terminated by either the Executive or the Employer at any time. Moreover, if prior to the Change of Control Date, the Executive's employment with the Employer terminates, then the Executive shall have no further rights under this Agreement.

(g) This Agreement may be executed in counterparts, each of which counterparts shall be deemed an original, but all of which together shall constitute one and the same instrument.

(h) Section 409A.

(i) It is intended that any amounts payable under this Agreement and the Employer's and the Executive's exercise of authority or discretion hereunder shall either be exempt from or comply with Section 409A of the Code (including the Treasury regulations and other published guidance relating thereto) ("Section 409A") so as not to subject the Executive to payment of any interest or additional tax imposed under Section 409A. To the extent that any amount payable under this Agreement would trigger the additional tax, penalty or interest imposed by Section 409A, this Agreement shall be modified to avoid such additional tax, penalty or interest extent reasonably possible) the intended benefit payable to the Executive.

(ii) Notwithstanding any provision of this Agreement to the contrary, if the Executive is a "specified employee" (within the meaning of Treasury Regulation Section 1.409A-1(i)), the Executive shall not be entitled to any payments upon a termination of the Executive's employment until the earlier of (i) the date which is six (6) months after the Executive's Separation from Service with the Employer for any reason other than death, or (ii) the date of the Executive's death. Furthermore, with regard to any benefit to be provided upon a termination of employment, to the extent required by Section 409A, the Executive shall pay the premium for such benefit during the aforesaid period and be reimbursed by the Employer therefor promptly after the end of such period. Any amounts otherwise payable to the Executive following a termination of his employment that are not so paid by reason of this Section 13(h)(ii) shall be paid as soon as practicable after the date that is six (6) months after the Executive's Separation from Service (or, if earlier, the date of the Executive's death). The provisions of this Section 13(h)(ii) shall only apply if, and to the extent, required to comply with Section 409A.

(iii) To the extent that any benefits or reimbursements pursuant to this Agreement are taxable to the Executive, any such benefit or reimbursement payment due to the Executive pursuant to any such provision shall be paid to the Executive on or before the last day of the Executive's taxable year following the taxable year in which the related expense was incurred. The benefits and reimbursements pursuant to such provisions are not subject to liquidation or exchange for another benefit and the amount of such benefits and reimbursements that the Executive receives in one taxable year shall not affect the amount of such benefits or reimbursements that the Executive receives in any other taxable year.

IN WITNESS WHEREOF, the Executive has hereunto set the Executive's hand and, pursuant to authorization from the Board, Air Group has caused this Agreement to be executed in its name and on its behalf, all as of the day and year first above written.

ALASKA AIR GROUP, INC.

By

Its Chairman (and Chief Executive Officer)

EXECUTIVE

NAME TITLE COMPANY

EXHIBIT A

GENERAL RELEASE AGREEMENT

(the "Executive"), on behalf of himself or herself, his or her descendants, dependents, heirs, 1. <u>Release</u>. executors, administrators, assigns, and successors, and each of them, hereby covenants not to sue and fully releases and discharges Alaska Air Group, Inc. ("Air Group") and each of its parents, subsidiaries and affiliates, past and present, as well as its and their trustees, directors, officers, members, managers, partners, agents, attorneys, insurers, employees, stockholders, representatives, assigns, and successors, past and present, and each of them (hereinafter together and collectively referred to as the "Releasees") with respect to and from any and all claims, wages, demands, rights, liens, agreements or contracts (written or oral), covenants, actions, suits, causes of action, obligations, debts, costs, expenses, attorneys' fees, damages, judgments, orders and liabilities of whatever kind or nature in law, equity or otherwise, whether now known or unknown, suspected or unsuspected, and whether or not concealed or hidden (each, a "Claim"), which the Executive now owns or holds or has at any time heretofore owned or held or may in the future own or hold as against any of said Releasees (including, without limitation, any Claim arising out of or in any way connected with the Executive's service as an officer, director, employee, member or manager of any Releasee, the Executive's separation from his or her position as an officer, director, employee, manager and/or member, as applicable, of any Releasee, or any other transactions, occurrences, acts or omissions or any loss, damage or injury whatever), resulting from any act or omission by or on the part of said Releasees, or any of them, committed or omitted prior to the date of this General Release Agreement (this "Agreement"), including, without limiting the generality of the foregoing, any Claim under Title VII of the Civil Rights Act of 1964, the Americans with Disabilities Act, or any other federal, state or local law, regulation or ordinance; provided, however, that the foregoing release does not apply to any obligation of the Employer to Executive pursuant to any of the following: (1) Section 7 of the Change of Control Agreement between the Executive and] (the "Change of Control Agreement"); (2) any equity-based awards previously granted by Air Air Group dated as of [Group to the Executive, to the extent that such awards continue after the termination of the Executive's employment with Air Group and its subsidiaries in accordance with the applicable terms of such awards (and subject to any limited period in which to exercise such awards following such termination of employment); (3) any right to indemnification that Executive may have pursuant to the Bylaws or Certificate of Incorporation of Air Group or under any written indemnification agreement with Air Group (or any of its subsidiaries or affiliates) or under applicable state law with respect to any loss, damages or expenses (including but not limited to attorneys' fees to the extent otherwise provided) that the Executive may in the future incur with respect to his or her service as an employee, officer or director of Air Group or any of its subsidiaries or affiliates; (4) with respect to any rights that the Executive may have to insurance coverage for such losses, damages or expenses under any directors and officers liability insurance policy of Air Group (or any of its subsidiaries or affiliates); (5) any rights to continued medical or dental coverage that the Executive may have under the Consolidated Omnibus Budget Reconciliation Act; or (6) any rights to payment of the Executive's accrued and vested benefits (if any) that Executive may have under a retirement plan sponsored or maintained by Air Group or any of its subsidiaries or affiliates that is intended to qualify under Section 401(a) of the Internal Revenue Code of 1986, as amended. In addition, this Release does not cover any Claim that cannot be so released as a matter of applicable law. The Executive acknowledges and agrees that he or she has received any and all leave and other benefits that he or she has been and is entitled to pursuant to the Family and Medical Leave Act of 1993.

2. <u>Acknowledgement of Payment of Wages</u>. The Executive acknowledges that he or she has received all amounts owed for his or her regular and usual salary (including, but not limited to, any bonus or other wages), and usual benefits through the date of this Agreement.

3. <u>Unknown Claims</u>. It is the intention of the Executive in executing this Agreement that the same shall be effective as a bar to each and every Claim hereinabove specified. The Executive acknowledges that he or she may hereafter discover Claims or facts in addition to or different from those which the Executive now knows or believes to exist with respect to the subject matter of this Agreement and which, if known or suspected at the time

of executing this Agreement, may have materially affected this settlement. Nevertheless, the Executive hereby waives any right, Claim or cause of action that might arise as a result of such different or additional Claims or facts.

4. <u>ADEA Waiver</u>. The Executive expressly acknowledges and agrees that by entering into this Agreement, he or she is waiving any and all rights or claims that he may have arising under the Age Discrimination in Employment Act of 1967, as amended ("<u>ADEA</u>"), which have arisen on or before the date of execution of this Agreement. The Executive further expressly acknowledges and agrees that:

(a) In return for this Agreement, the Executive will receive consideration beyond that which he or she was already entitled to receive before entering into this Agreement;

Agreement;

to sign it;

(c) The Executive has voluntarily chosen to enter into this Agreement and has not been forced or pressured in any way

(b) The Executive is hereby advised in writing by this Agreement to consult with an attorney before signing this

(d) The Executive was given a copy of this Agreement on [_____] and informed that he or she had twentyone (21) days within which to consider the Agreement and that if he or she wished to execute this Agreement prior to expiration of such 21-day period, he or she should execute the Acknowledgement and Waiver attached hereto as Exhibit A-1;

(e) Nothing in this Agreement prevents or precludes the Executive from challenging or seeking a determination in good faith of the validity of this waiver under the ADEA, nor does it impose any condition precedent, penalties or costs from doing so, unless specifically authorized by federal law; and

(f) The Executive was informed that he or she has seven (7) days following the date of execution of this Agreement in which to revoke this Agreement, and this Agreement will become null and void if the Executive elects revocation during that time. Any revocation must be in writing and must be received by Air Group during the seven-day revocation period. In the event that the Executive exercises his or her right of revocation, neither Air Group nor the Executive will have any obligations under this Agreement.

5. <u>No Transferred Claims</u>. The Executive warrants and represents that the Executive has not heretofore assigned or transferred to any person not a party to this Agreement any released matter or any part or portion thereof and he or she shall defend, indemnify and hold Air Group and each of its subsidiaries and affiliates harmless from and against any claim (including the payment of attorneys' fees and costs actually incurred whether or not litigation is commenced) based on or in connection with or arising out of any such assignment or transfer made, purported or claimed.

6. <u>Miscellaneous</u>. The following provisions shall apply for purposes of this Agreement:

(a) This Agreement shall be governed by and construed in accordance with the laws of the state of Washington, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and shall have no force or effect. This Agreement may not be amended or modified otherwise than by a written agreement executed by the parties hereto or their respective successors and legal representatives.

(b) The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.

(c) The Executive's or Air Group's failure to insist on strict compliance with any provision hereof or any other provision of this Agreement or the failure to assert any right the Executive or Air Group may have hereunder, shall not be deemed to be a waiver of such provision or right or any other provision or right of this Agreement.

(d) This Agreement may be executed in counterparts, each of which counterparts shall be deemed an original, but all of which together shall constitute one and the same instrument.

[Remainder of page intentionally left blank]

The undersigned have read and understand the consequences of this Agreement and voluntarily sign it. The undersigned declare under penalty of perjury under the laws of the State of Washington that the foregoing is true and correct.

EXECUTED this	day of	20, at	County,
	EX	KECUTIVE	
	[N	ame]	
EXECUTED this	day of	20. at	County,
	uay 01	20, at	County,
	Al	LASKA AIR GROUP, IN	С.
	[N	r: _ ame]	
	[1]	itle]	

EXHIBIT A-1

ACKNOWLEDGMENT AND WAIVER

I, _____, hereby acknowledge that I was given 21 days to consider the foregoing General Release Agreement and voluntarily chose to sign the General Release Agreement prior to the expiration of the 21-day period.

I declare under penalty of perjury under the laws of the State of Washington that the foregoing is true and correct.

EXECUTED this ____ day of ______ 20__, at _____ County, _____.

[Name]